

University of South Florida Scholar Commons

Graduate Theses and Dissertations

Graduate School

November 2017

Understanding the Linkages between Urban Transportation Design and Population Exposure to Traffic-Related Air Pollution: Application of an Integrated Transportation and Air Pollution Modeling Framework to Tampa, FL

Sashikanth Gurram University of South Florida, sgurram@mail.usf.edu

Follow this and additional works at: http://scholarcommons.usf.edu/etd Part of the <u>Environmental Engineering Commons</u>, <u>Environmental Law Commons</u>, and the <u>Urban Studies and Planning Commons</u>

Scholar Commons Citation

Gurram, Sashikanth, "Understanding the Linkages between Urban Transportation Design and Population Exposure to Traffic-Related Air Pollution: Application of an Integrated Transportation and Air Pollution Modeling Framework to Tampa, FL" (2017). *Graduate Theses and Dissertations.*

http://scholarcommons.usf.edu/etd/7030

This Dissertation is brought to you for free and open access by the Graduate School at Scholar Commons. It has been accepted for inclusion in Graduate Theses and Dissertations by an authorized administrator of Scholar Commons. For more information, please contact scholarcommons@usf.edu.

Understanding the Linkages between Urban Transportation Design and Population Exposure to Traffic-

Related Air Pollution: Application of an Integrated Transportation and Air Pollution Modeling

Framework to Tampa, FL

by

Sashikanth Gurram

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Civil Engineering Department of Civil and Environmental Engineering College of Engineering University of South Florida

Co-Major Professor: Amy L. Stuart, Ph.D. Co-Major Professor: Abdul R. Pinjari, Ph.D. James Mihelcic, Ph.D. Elizabeth Strom, Ph.D. Delcie Durham, Ph.D.

> Date of Approval: November 14, 2017

Keywords: urban sustainability, smart growth, environmental justice, population health, activity and travel patterns

Copyright © 2017, Sashikanth Gurram

DEDICATION

This dissertation is dedicated to my mother Seshu Kumari Gurram, my father Appa Rao Gurram, and my wife Swetha Gurram for their support, sacrifices, and unconditional love.

ACKNOWLEDGMENTS

I sincerely thank my "gurus" (co-advisers) Dr. Amy Stuart and Dr. Abdul Pinjari for their constant support and encouragement throughout this journey. Over the past seven years, I have grown both professionally and personally due to their efforts and inquiry. I would not have succeeded if not for their infectious passion towards science and their push towards quality work. I am a better person as a result of each and every meeting with them and am truly fortunate to have worked with two brilliant and passionate minds. I would also like to acknowledge all the financial support they have provided to me over the years.

I thank the members of my dissertation committee, Dr. James Mihelcic, Dr. Delcie Durham, and Dr. Elizabeth Strom, for the valuable feedback they provided—their critiques greatly improved this work. Special thanks to Dr. Fenda Akiwumi who agreed to be the external chair for my defense committee on very short notice.

This work would not have come to fruition if not for the kind help from several organizations and individuals. I would like to thank Elaine Martino with the Florida Department of Transportation (FDOT), who provided me access to the Tampa Bay Activity-Based Travel Demand Model; Dr. John Gliebe (RSG Inc.), who patiently went over the details of the activity-based model (ABM); Chunyu Lu (Gannett Fleming), Yifan Zhang and Han (BCC Engineering), who helped me with several operational details of the ABM; the MATSim developer and user group, with thanks to Dr. Kai Nagel and Dr. Amit Agarwal, who patiently addressed my queries on the MATSim model runs; Dr. Marcel Rieser, who provided me with a free license for SENOZON Via; Michael Kay of Saxonica, who provided me with a free enterprise license for Saxon XSLT, given my unfunded project; my very good friend Chetan Pinumalla for helping me with Java programming in the course of working with MATSim; and, finally, the user group of Stackoverflow, which helped with several of my queries in R programming.

I have been funded through a variety of sources over the course of my dissertation. I would like to acknowledge the financial support provided by the National Science Foundation (NSF); this material is based upon work supported by NSF under Grant No. 0846342, and any opinions, findings, and conclusions or recommendations expressed in this material are those of the author(s) and do not necessarily reflect the views of the NSF. I sincerely thank the USF Department of Civil and Environmental Engineering, which retained me as a teaching assistant for several semesters and provided me with financial stability; the Writing Studio at USF for retaining me as a writing consultant, a position I truly enjoyed, since August 2013; Writing Studio coordinators Karen Langbehn, Megan McIntyre, and Danielle Farrar for retaining my services on a continuous basis; my co-advisers Dr. Stuart and Dr. Pinjari, who believed in my ability to perform the writing consultant duties, and Dr. Sarina Ergas for backing up my application; the USF Graduate School financially supporting me with a University Graduate Fellowship in the first year of my Ph.D.; a grant from the Center for Transportation, Environment, and Community Health (CTECH), which provided me with the necessary funding; and Dr. Stuart for putting me on this grant; Patricia Ball, who provide editorial help for this dissertation; and Barbara Johnson and Shivali Vyas of the USF Department of Civil and Environmental Engineering and Nolan Kimball and Cathy Silva of the USF Department of Environmental and Occupational Health for their help with the tons of paperwork I had to deal with.

I was fortunate enough to make several lasting friendships over the course of my graduate career. I thank Dr. Ryan Michael, a truly great friend, with whom I enjoyed several productive sessions of R programming over food and fun talk—without his encouragement and guidance, I would still be stuck with SPSS; my good friend Haofei Yu, who, despite being busy, was always available to answer my questions on air quality modeling; Dr. Sujan Sikder for his friendship and guidance during the initial years of my Ph.D.; and Dr. Vijay Sivaraman and Aayush Thakur for their help at various stages of my career, especially with the ArcGIS package. I want to acknowledge the influence of some of my best friends from Virginia Tech, Dr. Aly Tawfik, Dr. Hao Chen, and Dr. Charan Tej Tanneru, on my decision to pursue a Ph.D., along with all of the fun times we spent together that made my graduate journey more enjoyable. I would like to specially acknowledge my friend Jennifer Iceton for all her support and advice that turned me into a better person; I sometimes feel that she believes more in me than I believe in myself. Special thanks to Rema Ramakrishnan for inspiring me through her work and for the feedback she provided to improve my defense presentation. Finally, I want to thank my graduate friends Surya Balusu, Divyakant Tahlyan, Talha Kocak, Mokaddesul Hoque (MD), Lukai Guo, Farshad Ebrahimi, Kaleb Marroquin, Steven Slinn, Trang Luong, Manvitha Rajalingola, Akbar Bakshi, Mohammadreza (Maz) Kamali, Bertho Augustin, Jill Sears, and Sarah Burns—know that you are a very special group of people from whom I learned a lot, and I earnestly thank you all for your friendship.

A person may be nothing without his/her family. None of this would have been possible without the great care, affection, and love that my mother Seshu Kumari Gurram has displayed over the past 32 years. I feel that this tribute pales in comparison to the towering selfless love that she has for me. My father Appa Rao Gurram has been my bedrock of support throughout my life, but especially in the last seven years when I needed it the most. He genuinely inspires me through his dedication and honesty towards work. I will succeed if I can put in half the effort he puts into anything he does. My sister Sirisha Gurram is one of the best persons I know and I thank her for the constant support, trust, and love for me. Over the last couple of years, my son Arihant Gurram has brought a lot of joy into our lives and made this journey very memorable. I thank him for all the cuteness and silliness he brought into my life. Finally, I cannot thank my wife Swetha Gurram enough for all her support and love over past seven years. The word "support" does not properly characterize all she has done for me. I truly appreciate her runs to grocery stores so I could focus on my work, all the teas and coffees at midnight, the wonderful and delicious food, and the living within means despite several financial tribulations and instabilities. I would not have been able to cope with such a lifestyle if I were in her place; she is a much better person than I can ever be. For this, I am indebted to her forever, which I can only hope to repay with everlasting love.

TABLE OF CONTENTS

LIST OF TABLES	iv
LIST OF FIGURES	v
ABSTRACT	viii
CHAPTER 1: INTRODUCTION	1
1.1 Background	1
1.2 Research Goal, Specific Aims, and Scientific Questions	4
1.3 Organization of this Dissertation	6
CHAPTER 2: LITERATURE REVIEW	8
2.1 Introduction	8
2.2 Health Impacts of Exposure to Traffic-Related Air Pollution	10
2.3 Exposure Inequalities	11
2.4 Interactions of Urban Form and Air Pollution Exposures	14
2.4.1 Interactions between Urban Form and Transport	15
2.4.2 Interactions between Urban Form and Air Quality	16
2.4.3 Interactions between Urban Form, Exposures, and Human Health	19
2.5 Characterization of Individual Exposure to Traffic-Related Pollution	21
2.5.1 Characterization of Spatiotemporal Distributions of Pollutant	
Concentrations	21
2.5.1.1 Ambient Air Sampling	21
2.5.1.2 Ambient Air Modeling	22
2.5.1.3 Gaussian Formulation	23
2.5.2 Characterization of Spatiotemporal Distributions of Individual Activity and	
Travel Patterns	24
2.5.2.1 Sample-Based Activity and Travel Surveys	24
2.5.2.2 Modeling the Activity and Travel Patterns of Individuals	26
2.5.2.3 Utility Maximization-Based Travel Demand Models	30
2.5.3 Characterization of Individual Exposures to Traffic-Related Pollution	32
2.6 Conclusion	35
CHAPTER 3: IMPACTS OF TRAVEL ACTIVITY AND URBANICITY ON EXPOSURES TO	
AMBIENT OXIDES OF NITROGEN AND ON EXPOSURE DISPARITIES	37
3.1 Note to Reader	37
3.2 Introduction	37
3.3 Methods	39
3.3.1 Study Area and Pollutant Focus	39
3.3.2 Estimation of Spatiotemporal Human Activity and Travel Patterns	41
3.3.3 Estimation of Diurnal Pollutant Concentrations at Activity Locations	42
3.3.4 Estimation of Daily Exposure Concentration and Exposure Error	42
3.3.5 Analysis of Exposure Distributions and Inequality	44
3.4 Results and Discussion	45

3.4.1 Distributions of Human Activity in the Tampa Area	45
3.4.2 Diurnally Varying Spatial Distributions of NO _x Concentration	48
3.4.3 Daily Time-weighted Activity-based Exposure Concentrations and their	
Social Distribution	49
3.4.4 Urban Form, Activity, and Exposure Relationships	55
3.4.5 Exposure Error	60
3.4.6 Social Distribution of Exposure Error	61
3.5 Limitations	64
3.6 Conclusion	66
CHAPTER 4: AN INTEGRATED MODELING FRAMEWORK TO ESTIMATE EMISSIONS.	
CONCENTRATIONS, AND EXPOSURES UNDER ALTERNATE TRANSPORTATION	
DESIGN SCENARIOS IN THE TAMPA AREA	69
4.1 Introduction	69
4.2 Methods	75
4.2.1 Study Area Demographics and Pollutant Focus	75
4.2.2 Transportation and Air Quality Modeling Framework	77
4.2.2 1 Estimation of Travel Demand using DaySim	77
4.2.2.7 Estimation of Dynamic Traffic Patterns using MATSim	70
4.2.2.2 Estimation of Mobile-Source Emissions using MOVES	81
4.2.2.5 Estimation of Pollutant Concentrations using P. I. INE	81 81
4.2.2.5 Analysis of Exposure	01
4.2.2.5 Analysis of Exposure	0.
4.5 Kesuits	84
4.3.2 Spatiotemporal Distributions of Emissions and Concentrations	01
4.3.2 Spanotemporal Distributions of Emissions and Concentrations	91
4.5.5 Distribution of Exposures by Demographics, Ordanicity, and Traver	05
Activity	95
4.3.4 Spatiotemporal Distributions of Exposure and Exposure Density	90
4.5.5 Impact of Spatiolemporal Resolution of Modeled Data on Human	100
Exposures	104
4.4 Discussion	104
4.5 Conclusion	108
CUADTED 5. IMPACT OF TRANSIT ODIENTED COMPACT CROWTH ON AIR OUAL ITY	
CHAPTER 5: IMPACT OF TRANSIT-ORIENTED COMPACT GROWTH ON AIR QUALITY	100
AND EXPOSURES TO TRAFFIC-RELATED AIR POLLUTION IN TAMPA AREA	105
5.1 Introduction	105
5.2 Methods	112
5.2.1 Study Area and Pollutant Focus	112
5.2.2 Modeling Framework	113
5.2.2.1 Specifications for the Transportation Models	114
5.2.2.2 Specifications for the Air Pollution Models	110
5.2.3 Alternate Urban Design Scenarios	117
5.2.3.1 Spatial Distribution of Population	11/
5.2.3.2 Bus-Transit Infrastructure	119
5.3 Results	121
5.3.1 Mode Shares and Travel Characteristics for Alternative Urban Design	
Scenarios	121
5.3.2 Distributions of Emissions and Concentrations of NO_x	123
5.3.3 Population Exposure	129
5.4 Discussion	130
5.5 Limitations	133

5.6 Conclusion	134
CHAPTER 6: SYNTHESIS	135
6.1 Introduction	135
6.2 Summary of Results	137
6.2.1 Impact of Activity and Travel Patterns, and Urbanicity on Exposures to	
NO _x	137
6.2.2 Integration of Models for Travel Demand and Air Pollution to Predict	
Population Exposures	138
6.2.3 Impact of Transit-Oriented Compact-Growth Policies on Population	
Exposure	139
6.3 Implications	141
6.3.1 Impacts of Activity and Travel Patterns on Exposure Estimates	141
6.3.2 Exposure Disparities	141
6.3.3 Robustness of Exposure Estimates using a Sample-Based and Population-	
Level Analysis	142
6.3.4 Impacts of using Low-Resolution Data Versus High-Resolution Data on	
Exposure Estimates	142
6.3.5 Impact of Transit-Oriented Compact-Growth on Air Quality and Population	
Exposure	143
6.4 Limitations and Future Research	144
6.4.1 Travel-Survey Samples	144
6.4.2 Single Pollutant Analysis	145
6.4.3 Microenvironmental Exposure Factors	145
6.4.4 Data Requirements for the Integrated Modeling Framework	145
6.4.5 Health Impact Assessments	146
6.4.6 Model Uncertainty	146
6.5 Conclusion	147
REFERENCES	148
APPENDIX A: COPYRIGHT PERMISSIONS	170

APPENDIX	A٠	COPYRIGHT PERMISSIONS	

LIST OF TABLES

Table 3.1 Average time spent per day by activity location type	46
Table 3.2 Group distribution statistics for daily activity-based exposure concentration and exposure error	52
Table 3.3 Linear regression model for activity-based exposure concentration	57
Table 4.1 Comparison of activity-based exposure modeling frameworks	72
Table 5.1 Summary of urban land use and transportation infrastructure characteristics for three alternate urban design scenarios	121

LIST OF FIGURES

Figure 2.1 Pollutant emissions by source category	9
Figure 2.2 Nested mode choice model structure	32
Figure 3.1 Study area of Hillsborough County, Florida	40
Figure 3.2 Spatial distribution of sample population activity-time (% of time spent) and urbanicity in study area	47
Figure 3.3 Estimated diurnal cycle of hourly average ambient NO _x concentrations ($\mu g/m^3$) in study area, from dispersion modeling results	49
Figure 3.4 Cumulative distributions for daily exposure concentrations	50
Figure 3.5 Cumulative distributions of daily activity-based exposure concentration for population subgroups	53
Figure 3.6 Cumulative distributions of time-weighted NO _x concentration ($\mu g/m^3$) and NO _x exposure (μg -hr/m ³) by activity type location for all sampled daily records, including some activity away from residence	58
Figure 3.7 Cumulative distributions of exposure error for population subgroups	62
Figure 4.1 Study area of Hillsborough County, Florida	76
Figure 4.2 Integrated transportation and air quality modeling framework for population exposure estimation	77
Figure 4.3 Diurnal cycle of daily cumulative duration spent by all county residents	85
Figure 4.4 Total activity duration (person-hours) by block group and hour of day for all individuals in Hillsborough County	87
Figure 4.5 Activity-duration densities (person-hours/km ²) for Hillsborough resident sample in 2010	88
Figure 4.6 Bi-hourly average passenger car volumes (left) and travel speeds (right) for Hillsborough County on typical weekday	90
Figure 4.7 NO _x emissions by hour of day from cars for representative winter day in Hillsborough County	92
Figure 4.8 Diurnal cycle of NO_x concentrations from on-road passenger cars	93

Figure 4.9 NO _x concentration distribution by hour of day due to on-road passenger cars in Hillsborough County for winter months 2010	94
Figure 4.10 Cumulative distributions of personal exposure concentration for NO _x resulting from passenger cars	95
Figure 4.11 Subgroup inequality index by demographics versus percentile of cumulative distribution of individual daily exposure concentrations	98
Figure 4.12 Total cumulative exposure ($\mu g/m^3$ person-hr) to NO _x by block group and hour for sample of Hillsborough County residents	99
Figure 4.13 Cumulative exposure density (µg/m ³ * person hr/km ²) for Hillsborough County resident sample 2010	101
Figure 4.14 Relative percent difference in exposures obtained using highly spatiotemporally- resolved activity and concentration data and low-resolution activity and travel, and concentration data	102
Figure 4.15 Diurnal cycle of hourly observed and estimated concentrations at near-roadway monitor	106
Figure 5.1 Spatial distribution of block group-level residential density in 2040 base and compact- growth scenarios	118
Figure 5.2 Highway and transit infrastructure in 2040 for low-transit and enhanced-transit scenarios	120
Figure 5.3 Mode shares for low-transit, enhanced-transit, and compact-growth scenarios	121
Figure 5.4 Percent change in cumulative travel distance, travel time, and number of trips for enhanced-transit and compact-growth scenarios compared with low-transit scenario	123
Figure 5.5 Diurnal NO _x emissions for low-transit, enhanced-transit, and compact-growth scenarios	124
Figure 5.6 Diurnal cycle of domain-average NO _x concentrations for low-transit, enhanced-transit, and compact-growth scenarios	125
Figure 5.7 Distribution of hourly NO _x concentration for low-transit, enhanced-transit, and compact-growth scenarios	125
Figure 5.8 Spatial locations of maximum NO _x concentrations for low-transit, enhanced-transit, and compact-growth scenarios	126
Figure 5.9 Spatial distribution of the difference in NO _x concentrations between enhanced-transit and low-transit scenarios (enhanced transit-low transit) for morning and evening peaks hours	127
Figure 5.10 Spatial distribution of difference in NO _x concentrations between compact-growth and low-transit scenarios (compact growth-low transit) for morning and evening peaks hours	128

Figure 5.11 Spatial distribution of difference in NO _x concentrations between compact-growth and enhanced-transit scenarios (compact growth-enhanced transit) for morning and evening peaks hours	128
Figure 5.12 Distribution of population exposure for low-transit, enhanced-transit, and compact- growth scenarios	129
Figure 5.13 Differences in block group-level aggregated exposure densities between different scenarios	130

ABSTRACT

Rapid and unplanned urbanization has ushered in a variety of public health challenges, including exposure to traffic pollution and greater dependence on automobiles. Moreover, vulnerable population groups often bear the brunt of negative outcomes and are subject to disproportionate exposure and health effects. This makes it imperative for urban transportation engineers, land use planners, and public health professionals to work synergistically to understand both the relationship between urban design and population exposure to traffic pollution, and its social distribution. Researchers have started to pay close attention to this connection, mainly by conducting observational studies on the relationship between transportation, urban form, and air quality. However, research on this topic is still nascent. Further, most studies do not predict exposures under alternative urban design scenarios. Hence, to understand the relationship between urban design and population exposures, there is a need to build and apply integrated modeling tools that can predict exposures under alternative urban design scenarios.

Within this context, the overarching goal of this dissertation is to understand how the transportation infrastructure of cities can be designed for improved urban air quality and mitigation of population exposure to traffic pollution. The study area is Hillsborough County, Florida, a sprawling region with limited transit availability and a diverse population along with a mix of urban, suburban, and rural areas. The rank of the county for sprawl and congestion metrics (i.e., yearly delay and travel time index) fall in the mid-range in comparison with other US urban regions. Thus, the study area may be representative of other US urban regions with medium sprawl and above-average congestion levels. Oxides of nitrogen (NO_x), a surrogate for traffic pollution, is the focus pollutant. The Health Effects Institute's report on traffic-related air pollution identifies NO_x as a potential surrogate due to its relative ease of measurement and the abundance of epidemiologic studies that characterize exposures to NO_x .

viii

Because exposures are dependent on the spatial and temporal distributions of both people and pollution, this study first sought to understand the importance of activity and travel patterns of individuals for exposure estimation. To estimate exposures, the 2009 National Household Travel Survey (NHTS) data containing daily individual activity records, ArcGIS-estimated shortest-time travel route profiles, and the annual-average diurnal cycle of NO_x derived from hourly CALPUFF dispersion model results from 2002, were combined. Two exposure measures were estimated: activity-based exposure that considers the daily activity and travel patterns of individuals, and residence-based exposure that considers only the pollutant concentrations at the residences. Exposure estimation without inclusion of activity and travel patterns was found to slightly underestimate activity-based exposures on average. Additionally, disproportionately-high exposures were found for blacks, Hispanics, below poverty groups, urban residents, and people whose daily travel time is greater than one hour. Finally, urbanicity and travel time variables were found to be the strongest predictors of daily exposure.

Following this, a modeling framework was developed to predict population exposure by integrating activity-based travel demand modeling (DaySim), dynamic traffic assignment simulation (MATSim), mobile-source emission estimation (EPA MOVES), and pollutant dispersion modeling (R-LINE). This modeling framework was used to predict daily population and subgroup exposures by estimating the high-resolution spatial and temporal distributions of both pollution and individual activities for the year 2010. Persistent exposure inequalities were found at the population-level; blacks, Hispanics, active age groups (19-65 years), below-poverty and middle-income groups, urban residents, and individuals with daily travel times above one hour had higher estimated exposures than the population mean. These inequalities for blacks, Hispanics, and below-poverty non-white groups worsened at higher exposure levels. Use of low-resolution activity and pollution data as opposed to high-resolution data led to underestimation of exposures (by 10% on average).

Finally, the integrated modeling framework was employed to understand the relationship between urban transportation and land use design, air quality, and population exposure. Three scenarios that are based on a combination of diesel-bus transit services and residential distribution were simulated.

ix

Specifically, the low-transit scenario used the 2040 base residential distribution and the 2010 bus services. The enhanced-transit scenario applied the 2040 bus services proposed for the county instead. The compact-growth scenario added an increase of residential density to this latter scenario. Specifically, about 37% of total households were redistributed from locations with low accessibility to jobs and transit to locations near employment and bus stops. Results indicate slight higher non-car travel mode shares in the enhanced-transit and compact-growth scenarios compared to the low-transit scenario (with a 7.1% increase for walking, 0.2% for bicycle, and 1.8% for transit for the compact-growth scenario versus the low-transit scenario). The enhanced-transit scenario resulted in slightly lower daily total travel distances and times compared with the low-transit scenario, but daily total emissions and winter mean concentration of NO_x were higher, i.e., the increase in bus transit services did not induce sufficient shifts in travel mode to overcome the concomitant increase in diesel-bus emissions. The compact-growth scenario resulted in lower daily total travel distance (9%) and travel time (2.1%) and daily total emissions of NO_x (11%) and its winter mean concentration (9%), compared with both the low-transit and enhancedtransit scenarios. Although the compact-growth scenario improved the air quality of the region on average, daily population mean exposure was higher compared with both the low-transit (29%) and enhanced-transit scenarios (25%). This is largely due to the redistribution of population to urban core locations that had higher pollutant levels. Overall, neither the bus-transit improvements nor residential compaction strategies alone were sufficient to mitigate population exposures. Combining them with transit that services both origins and destinations, uses clean fuel technologies, and separates major roadways from dense residential pockets may be needed for greater exposure reductions.

Overall, this dissertation has implications for population exposure to traffic pollution and public health through transportation and land use interventions. Results presented here may be applicable to other study regions that have similar composite sprawl scores as the Tampa Bay area. Future studies should exploit spatially-and temporally-resolved data on human activities and travel, vehicular activities, and air quality for better characterization of population exposure. Engineers and planners should pay greater attention to integrated land use and transport planning; lone, disjointed, and ill-planned design

Х

interventions may exacerbate population exposure to air pollution. The integrated modeling framework presented here may be applied in a wide variety of urban contexts to further explore the nexus between travel demand, air quality, and exposures. However, before such an exercise is undertaken, a preliminary analysis should be conducted to assess the transferability of the framework. Policies that could be studied include mixed land use design, urban compaction with controlled sociodemographic distributions (to assess exposure inequality), and inclusion of additional types of transit and fuel technologies.

CHAPTER 1: INTRODUCTION

1.1 Background

Urbanization refers to the increase in size, density, and heterogeneity of cities and is frequently associated with factors including population mobility, segregation, and industrialization (Vlahov & Galea, 2002). Rapid urbanization is a common feature across the globe today (Alig et al., 2004; Soubbotina, 2004). The global urban population has been on the rise since 1950; it exceeded the global rural population for the first time in 2007, and the world population has predominantly remained urban since then. Additionally, it is projected that two-thirds (approximately 66%) of the world population will be urban by 2050 (United Nations et al., 2015). This level of rapid urbanization poses a serious challenge for sustainable development (Cohen, 2006).

Although urbanization is associated with several positive outcomes, including economic growth, poverty reduction, and improved access to infrastructure and services, it also has ushered in an array of concerns. Evidence from around the world suggests that changes in land use and urbanization adversely impact human lifestyle and health (Galea & Vlahov, 2005; Moore et al., 2003; Popkin, 1999), environment (Burak et al., 2004; Seto et al., 2010), and climate (Kalnay & Cai, 2003). It could be argued that the impact of urbanization on human health assumes special significance due to the complex linkages between them. Rapid and unplanned urbanization is associated with several health concerns, including cardiovascular diseases (Yusuf et al., 2001), diabetes (Hu, 2011), and cancer (World Cancer Research Fund & American Institute for Cancer Research, 2007). It should not be surprising that these health outcomes culminate from multiple pathways associated with urbanization, including air pollution, physical inactivity, access to transport and healthy food, and non-communicable diseases (Giles-Corti et al., 2016; World Health Organization, 2010).

One of the major problems of rapid and inadequately planned urbanization is ambient urban air pollution. The World Health Organization (WHO) has estimated that ambient air pollution (in both urban and rural areas) is linked to 3.7 million deaths globally, making it the largest environmental health risk (World Health Organization, 2014). Within the context of the United States, combustion-related emissions were linked to approximately 200,000 premature deaths (Caiazzo et al., 2013). Urban air pollution also has been associated with a variety of environmental and health concerns, including but not limited to acid deposition, asthma, and cardiovascular and cardiopulmonary diseases (Gauderman et al., 2000; HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). In addition to its impact on health, urban air pollution also has significant economic costs, at 2–5% of a nation's gross domestic product (United Nations Environment Programme, 2003). In the US, urban air pollution is responsible for nearly three-quarters of the gross annual damages resulting from air pollution (Muller & Mendelsohn, 2007). Finally, on a positive note, even modest decreases in air pollution levels were associated with significant cost savings (Deschenes et al., 2012; Grabow et al., 2012). Recognizing these complex linkages between air pollution and urbanization, a few studies argue that urban development policies that seek to abate urban air pollution levels hold the key for mitigating the associated human health, environmental, and economic costs.

Whereas urban air pollution is an amalgamation from various sources, pollution from the transportation sector contributes significantly towards it (US Environmental Protection Agency, 1994). The US Environmental Protection Agency (US EPA) estimated that the transportation sector accounted for about 26% of total US greenhouse gas (GHG) emissions for 2014 (US Environmental Protection Agency, 2016). Several studies linked exposure to traffic-related pollution with exacerbation of asthma, onset of childhood asthma, non-asthma respiratory symptoms, impaired lung function, total and cardiovascular mortality, and cardiovascular morbidity (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Further, inequalities in exposure to traffic-related pollution also were documented (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Further, inequalities in exposure to traffic-related pollution, 2010; Yu & Stuart, 2013).

Considering the broader health- and equity-related impacts, it is necessary to investigate the factors that are linked with transport emissions.

Urban form is suspected to be a factor that determines the air quality of a region. The lowdensity urban form, pursued during the early and mid-20th century, resulted in increased spatial segregation between communities, increased automobile dependence, sedentary lifestyle, incompetent public transit, and negative health impacts (Camagni et al., 2002; Frank & Engelke, 2001; Frank & Engelke, 2005). Additionally, the prevalence of health and environmental inequalities associated with air pollution has been well documented across the world (Deguen & Zmirou-Navier, 2010; Samet & White, 2004; Stuart & Zeager, 2011). Although past research efforts identified correlations between factors including race, socioeconomic status, and urbanicity and health-related inequalities, the pathways through which these inequalities manifest are not very clear. To this end, researchers proposed a few strategies, including restriction of the outward expansion of cities, increase of residential density, development of mixed-use neighborhoods, and investment in transit services to curb the use of personal vehicles, as a way to design cities that improve public health and reduce environmental concerns; these strategies are collectively referred to as "smart growth" (Burchell et al., 2000; Downs, 2005).

The impact of the aforementioned smart growth approaches on curbing air pollution, population exposure, and inequalities are unclear. A few researchers studied the effects of compact urban forms on traffic-related emissions and found that emissions were generally lower in compact urban forms compared to sprawling forms (Hankey & Marshall, 2010; Makido et al., 2012; Stone et al., 2007). Other studies found a lower number of ozone exceedances (Stone, 2008), lower PM_{2.5} levels across most of the urban region except for urban centers (Hixson et al., 2009) and lower particulate and ozone levels (De Ridder et al., 2008b) in compact urban forms. Despite these improvements in the air quality, several studies observed an increase in population-weighted exposures to traffic-related pollution in compact and high-density urban forms (Clark et al., 2011; Hixson et al., 2009). Thus, it is entirely reasonable to ask, what makes growth strategies "smart" and how can we design cities to simultaneously improve the air quality while reducing the population exposure and the inequalities?

No investigation of the impact of smart growth policies (as currently defined) on air quality would be complete without considering the transportation and land use piece of the smart growth puzzle. Transportation infrastructure plays a crucial role in shaping the urban form of a region and constitutes one of the vital design concepts for sustainable urban forms (Jabareen, 2006). Previous literature clearly identifies the connection between land use and transportation (Cervero & Gorham, 1995), albeit there is some disagreement with regard to the strength and direction of this connection (Crane, 2000; Giuliano, 1995; Handy, 2005). Preliminary investigations show that use of alternate fuels, advanced vehicle technologies, and promotion of public transit systems could lead to the realization of sustainable urban forms (Chen & Whalley, 2012; Harford, 2006). However, a comprehensive understanding of the attributes of transportation infrastructure that advance urban sustainability has not yet been attained. Thus, it is extremely important to investigate the linkages between urban form, transportation infrastructure and land use, air pollution, and the social distribution of exposures. This study attempts to add to the body of literature on sustainable urban design policies that may help achieve health and environmental equity through exploring alternate urban transportation infrastructure and land use design scenarios.

1.2 Research Goal, Specific Aims, and Scientific Questions

The overarching goal of this research was to understand and predict the impact of urban transportation infrastructure and land use design on human exposure to traffic-related pollutants with a focus on impacts on social inequality. To this end, the specific aims addressed here are outlined and discussed below.

Aim 1 of this dissertation was to understand the impacts of spatiotemporally-resolved activity and travel patterns on estimated exposures to traffic-related air pollution in the Tampa area. The specific science questions that were addressed include:

- How are population activities distributed spatiotemporally in the study domain?
- How are exposures distributed among population groups in the study domain?
- What is the strength and direction of disparities between groups?

- Does urban form influence the strength of exposures and their social distribution?
- How much does the representation of spatiotemporal activity locations impact exposure estimates?
- Are the errors associated with exposure estimation different for different population subgroups?

I hypothesize that consideration of activity and travel patterns leads to significantly different exposure estimates, as opposed to residence-based exposure estimates.

Aim 2 of this dissertation was to develop a modeling system that integrates activity-based traveldemand simulation, mobile source emissions estimation, and pollutant dispersion simulation for the study of impacts of urban transportation infrastructure on human exposures to air pollution. A few science questions that were addressed during this model development include:

- How does the spatiotemporal distribution of activities, emissions, and exposures change when the sample is scaled to the full population using the modeling framework?
- Are the results from the modeling framework and the earlier sample-based analysis consistent with one another?
- How much do the exposures and their social distribution vary between sample-based and full population studies?
- Does the use of a highly spatiotemporally-resolved framework as opposed to the low resolution frameworks used in earlier studies warrant improved population and group-wise exposure estimates?

I hypothesize that a population-level analysis will reveal distinct exposure patterns that are not apparent in a sample-based analysis.

Aim 3 of this study was to understand and predict impacts of transit-oriented compact-growth design scenarios on patterns of exposure to select traffic-related air pollutants in the Tampa area. The modeling framework developed under Aim 2 was used to investigate the following science questions:

- How may population activities, traffic-related pollution, and population exposure be distributed spatiotemporally in the study domain under different urban transportation and land use design scenarios?
- What may be the individual contribution of urban transportation and land use design on urban air quality and population exposure?

I hypothesize that transit-oriented compact urban forms will lower population exposure compared to a sprawling urban form with limited transit services.

Thus, this work seeks to improve the current understanding on the nexus between transportation infrastructure and land use design, emissions, air quality, and exposures to urban air pollution.

1.3 Organization of this Dissertation

This dissertation is organized as follows. Chapter 2 presents the current state of science on the impact of smart growth policies on air quality and exposures to traffic-pollution, specifically focusing on studies that investigate the impact of transit-oriented compact-growth scenarios on air quality and exposures. Additionally, air pollution exposure methodologies are reviewed, as are tools that can forecast air quality under alternate urban growth scenarios.

Chapter 3 focuses on the impact of activities and travel on exposures to traffic-related pollution and their social distribution. Data from the 2009 National Household Travel Survey (NHTS) were used to generate activity and travel patterns of individuals, which were used in conjunction with spatiotemporal estimates of 2002 NO_x concentrations to characterize the population and group-wise exposures and their social distribution in the Tampa region.

Chapter 4 focuses on the development of the modeling framework that combines activity-based travel demand simulation, dynamic traffic assignment, mobile source emission estimation, and dispersion simulation to estimate the spatiotemporal distributions of the human activities, regional air quality, and exposures to traffic-pollution. This system was applied to Tampa to estimate the full population and group-wise exposures. Additionally, the system also was used to understand the need for high-resolution population activity and air pollution data for exposure estimation.

Chapter 5 presents the application of the modeling framework developed under Chapter 4 to estimate the spatiotemporal distributions of the regional air quality and exposures to traffic-pollution under alternate urban design scenarios. Three scenarios—low-transit, enhanced-transit, and compact-growth—were simulated for 2040 to predict the impact of compaction strategies and additional bus transit on activity and travel patterns of individuals, urban air quality, and population exposure.

Finally, Chapter 6 provides a synthesis of this research and recommendations for future work.

CHAPTER 2: LITERATURE REVIEW

2.1 Introduction

Anthropogenic air pollution is predominantly driven by the combustion process, which generates a variety of pollutants (World Health Organization, 2004) and significantly impacts human health and the environment. The 2014 World Health Organization factsheet on household and ambient air pollution estimates that seven million worldwide premature deaths are attributable to air pollution, making it the largest environmental health risk (World Health Organization, 2014). Thus, in the short term, combustion emissions may pose a great risk to the human health and well-being; even in the longer term, combustion-related emissions are forecast to greatly affect sea-level rise, thus significantly altering current world geography and displacing wide sections of civilization (Winkelmann et al., 2015). In view of these current estimates and long-term forecasts, mitigation of combustion-related emissions assumes great importance.

Energy use in the commercial, industrial, residential, and transportation sectors accounts for a majority of anthropogenic emissions. Within these sectors, transportation often accounts for a substantive portion of the emissions, as shown in Figure 2.1. The primary emissions from motor vehicles are collectively termed "traffic-related air pollution" (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Exposure to traffic-related air pollution has been associated with a wide spectrum of negative health outcomes and health inequalities (Apelberg et al., 2005; HEI Panel on the Health Effects of Traffic-related Air Pollution, 2010). As such, accurate characterizations of exposure to traffic-related pollution are needed to understand its impact on human health and the corresponding social inequalities.



Figure 2.1 Pollutant emissions by source category. (Source: US Environmental Protection Agency, 2016) Whereas accurately estimating exposures to traffic-related pollution is important, it is also equally important to identify and understand the factors that are linked with this pollution. An emerging area of research suggests that urban form including the transportation infrastructure within it, may perhaps be one of the important factors that influence the air quality, personal exposure levels, and population health (Giles-Corti et al., 2016; Hankey & Marshall, 2010; Stevenson et al., 2016; Stone et al., 2007). However, there is a considerable debate on the effectiveness of urban forms that encapsulate "smart growth" features in realizing sustainability goals (Dieleman et al., 1999; Echenique et al., 2012; Gordon & Richardson, 1997; Neuman, 2005). Thus, there is a need to gather further evidence in this area using tools that can robustly forecast the impact of alternate urban design scenarios on human activity and travel patterns, pollutant emissions, concentrations, and personal exposure to them.

The remainder of this chapter reviews the current state of literature on linkages between urban transportation and land use design, air quality, and exposure to traffic-related pollution and its social distribution. The health impacts of exposure to traffic-related pollution are reviewed first. The literature pertaining to the social distribution of the exposures and the resulting inequalities is reviewed next, followed by a review of the literature that seeks to understand the relationship between urban design, air quality, and population exposure. Finally, the methodological aspects pertaining to the estimation of population exposure to traffic-related pollution is presented with a focus on choosing the appropriate tools for this research.

2.2 Health Impacts of Exposure to Traffic-Related Air Pollution

Exposure to combustion-related air pollution is associated with mortality and a host of negative health outcomes. Specifically, the Harvard Six Cities study showed that high levels of fine and sulfate particles were strongly associated with mortality (Dockery et al., 1993). Following this, Pope III et al. (2002) found associations between particulate and sulfate pollution and all-cause, lung cancer, and cardiopulmonary mortality. More recently, combustion emissions of PM_{2.5} and ozone were estimated to be associated with about 210,000 premature deaths per year in the US (Caiazzo et al., 2013); of these, transportation-related PM_{2.5} and ozone emissions were estimated to account for approximately 53,000 and 5,000 premature deaths, respectively.

Recognizing the significant contribution of transport emissions toward air pollution, the Health Effects Institute (HEI) commissioned a special panel to review the literature and understand the associations between traffic-related air pollution and negative health outcomes. Traffic-related air pollution comprises a complex mix of primary pollutants emitted from motor vehicles. It should be noted that secondary pollutants such as ozone that originate from the primary pollutants are not considered traffic-related pollutants. Some important traffic-related pollutants include carbon dioxide (CO₂), carbon monoxide (CO), hydrocarbons (HC), nitrogen oxides (NO_x), particulate matter (PM), and mobile source air toxics (MSATs) including benzene, formaldehyde, acetaldehyde, 1,3-butadiene, and lead (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Since it is extremely difficult to measure all the components of this traffic-pollutant mix, surrogate pollutants are usually selected to approximate the impact of exposure to traffic pollutants (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010; Oglesby et al., 2000). The HEI report identifies pollutants including CO, nitrogen dioxide (NO₂), and benzene as potential surrogates for exposure to traffic-related air pollution, although

they caution that none of these pollutants meet the criteria for an ideal surrogate. In this study, NO_x was chosen as the surrogate for traffic-related exposure for a few reasons. First, NO_x consists of both directly emitted NO and secondary nitrogen dioxide, a criteria pollutant regulated by the US EPA. NO₂ is also measured by the US EPA near-road monitoring network due to the influence of traffic-related air pollution. Additionally, ambient NO₂ concentrations, in conjunction with surrogates of traffic (i.e., traffic volumes or distance to roadways), provide a good characterization of traffic-related pollution at the local scale. Finally, several previous studies focusing on exposure to traffic-related pollution used NO_x as a surrogate pollutant (Beevers et al., 2013; Gurram et al., 2015; Kim et al., 2004; Raaschou-Nielsen et al., 1997; Yu & Stuart, 2013), providing for good comparisons for this study.

The HEI panel concluded that individuals living within a range of 300–500 meters from a major roadway are the most affected due to traffic emissions. Given the substantial proportion of population that live near roadways in the United States, exposure to traffic-related pollution is a likely public health concern and deserves attention (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). The panel also concluded that sufficient evidence exists to infer a causal association between exposure to traffic-related pollution and exacerbation of asthma. Additionally, they found suggestive evidence to infer causal association between exposure to traffic-related pollution and onset of asthma, non-asthma respiratory symptoms, reduced lung function, total and cardiovascular mortality, and cardiovascular morbidity (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Thus, exposure to traffic-related air pollution can be categorized as a public health concern, and further studies are needed to characterize the disaggregate, group-wise, and population-level exposures to traffic-pollution.

2.3 Exposure Inequalities

Several studies suggested that exposure levels for different types of pollutants may vary depending on the socio-economic makeup of the population subgroups (Chakraborty, 2009; Marshall, 2008; Samet & White, 2004; Stuart & Zeager, 2011; Yu & Stuart, 2013). For example, Marshall (2008) suggested that certain sections of population residing near an industry might get exposed to higher levels of primary pollutants emitted from the industry when compared to the subgroups residing farther away.

Alternatively, the groups living farther away from the industry might get exposed to higher levels of secondary pollutants, formed due to the chemical reactions between primary pollutants from the industrial source and pollutants from other sources, when compared to the groups residing near the industrial source. These differences in exposure to air pollutants within different population demographics is called environmental inequality (Anderton et al., 1994) and lack of environmental justice (US EPA Office of Minority Health, 2003). Environmental justice has been defined by the US EPA as the fair treatment and meaningful involvement of all people regardless of race, color, national origin, culture, or income with respect to the development, implementation, and enforcement of environmental laws, regulations, and policies (US EPA Office of Minority Health, 2003).

Past studies tried to identify and understand the underlying patterns corresponding to the social distribution of exposures and inequalities (Deguen & Zmirou-Navier, 2010; Marshall, 2008). Deguen and Zmirou-Navier (2010) conducted a literature review on studies pertaining to social inequalities in health risks related to ambient air quality in Europe and concluded that subgroups that belong to lower socioeconomic status were subject to greater harmful health effects despite not always being exposed to higher pollutant levels. This could mainly be attributed to inferior medical treatment, limited access to good food, micronutrient deficiencies, and concurrent illnesses (Deguen & Zmirou-Navier, 2010; Romieu et al., 2004). Marshall (2008) used spatially- and temporally-resolved and microenvironment-adjusted ambient air pollutant concentrations to study the pollutant exposure patterns among different population subgroups in California's south coast air basin. The study concluded that non-white and low-income groups were subject to high primary pollutant exposures when compared to white and high-income groups. Conversely, for ozone, white and high-income groups were exposed to higher concentrations when compared to non-white and low-income groups. Similarly, people living in high-density regions were subject to high primary pollutant and low secondary pollutant exposures, whereas people living in low-density regions were exposed to low primary pollutant and high secondary pollutant concentrations. Hajat et al. (2015) conducted a global review of literature on socioeconomic disparities in exposure to air pollution that specifically synthesized 22 North American studies and found a broadly consistent pattern

of higher concentrations of criteria pollutants in lower-socioeconomic status (SES) individuals and communities; a few exceptions in which higher-SES communities were exposed to higher concentrations were observed in New York City, Toronto, and Montreal (Hajat et al., 2015).

A few exposure-related socio-demographic inequality studies were conducted in the Tampa Bay region (Chakraborty, 2009; Stuart et al., 2009; Stuart & Zeager, 2011; Yu & Stuart, 2013). Chakraborty (2009) investigated the environmental inequalities in the region for socio-economically and transportation-disadvantaged groups and concluded that minority groups such as African Americans, Hispanics, and those living below poverty levels were subject to disproportionate cancer risks and respiratory hazards. Further, the study identified that the individuals from households with no vehicles were subject to disproportionately higher health risks. Stuart et al. (2009) studied the social distribution of neighborhood-scale air pollution and identified that blacks, Hispanics, and people living in poverty were disproportionately closer to sources of air pollution yet farther from air quality monitoring sites than whites and non-poverty groups, respectively. For the same study region, Stuart and Zeager (2011) studied ambient NO₂ levels and traffic levels near elementary schools and found higher NO₂ and traffic levels near schools with predominant enrollments of black, Hispanic, and economically-disadvantaged children compared to schools with predominant enrollments of white, Asian, or Pacific Islander children. Yu and Stuart (2013) conducted a modeling study in the Tampa Bay area to characterize the spatiotemporal distributions of NO_x and the exposure inequalities and found that blacks, Hispanics, and lower-income subgroups were subject to greater exposures than the county average. Additionally, they found that this disproportionality of exposures increased with increasing levels of annual average exposures, i.e., the proportion of minorities and low-income groups at greater annual average exposures were higher compared to their proportion at a lower annual average exposure. Finally, Yu and Stuart (2016) expanded their earlier study by considering additional primary and secondary pollutants and found that exposure inequalities and inequality indices (for blacks, Hispanics, and low-income subgroups) were persistent when considering primary pollutants. However, the inequality indices for exposure to secondary

pollutants displayed complex trends and exhibited reverse disproportionality (i.e., minorities and lowerincome subgroups) at the highest concentration levels.

A limitation for most of the studies in the Tampa region and the US in general was the noninclusion of the activity and travel patterns of individuals for exposure estimation. Setton et al. (2011) suggested that non-inclusion of individual activity and travel patterns for exposure estimation results in biased exposure estimates. Thus, studies that explicitly consider individual spatiotemporal locations are warranted to understand the true strength and direction of exposure inequalities. Further, many studies focused explicitly on the associations between inequalities and sociodemographic variables including socioeconomic status, race, and income. However, it is also important to understand how inequalities are related with social identity variables including age, urban/rural status (Miao et al., 2015), and time spent in travel and out-of-home activities. Finally, it would be of interest to understand the social distribution of exposures to traffic-related pollution, in addition to including pollution from point and area sources, given the public health concerns of traffic-related pollution (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Within this context, the current study explores the social distribution of exposures to traffic-related pollution by focusing on demographic, urbanicity, and travel-related variables including race/ethnicity, income, age, residential status (urban/rural), and daily travel time. Thus, this study can add to the body of literature on environmental inequalities by focusing specifically on exposure to traffic-related pollution.

2.4 Interactions of Urban Form and Air Pollution Exposures

"Urban form is defined as a composite of characteristics related to land use patterns, transportation system, and urban design" (Handy, 1996; Jabareen, 2006). Over the past few decades, much focus has been placed on identifying and defining sustainable urban forms due to the potential impact of urban design on human physical and mental well-being (Jackson, 2003). Jabareen (2006) identified seven significant and recurring themes for sustainable urban form—compactness, sustainable transport, density, mixed land uses, diversity, passive solar design, and greening. Transport is inextricably linked with the urban morphology of a region and plays a crucial role in shaping the urban

form; at the same time, transport is also presented as one of the most important issues that connect environmental debates and urban form (Burton et al., 2003). The following sections provide an overview of the interactions between urban form, transport, air quality, exposures, and human health.

2.4.1 Interactions between Urban Form and Transport

Urban form and transport characteristics are invariably linked with one another. However, these linkages appear to propagate through various features of the urban form and built environment. Cervero and Kockelman (1997) found that built environment influences the travel demand along three dimensions i.e., "three Ds"—density, diversity, and design. Specifically, using a variety of data sources including travel surveys, the US Census, land use records, and field surveys, they found that urban forms that are dense, diverse, and pedestrian-oriented would reduce trip rates and encourage non-auto travel. However, they also concluded that the effect of built environment dimensions on transport is marginal, although not inconsequential (Cervero & Kockelman, 1997). Similarly, a review of the interactions between built environment and travel by Ewing and Cervero (2001) found that vehicle miles traveled (VMT) or vehicle hours traveled were a function of regional accessibility more than local accessibility; thus, dense and mixed-use developments that are not well connected to other dense and mixed-use neighborhoods may offer only modest regional travel benefits (Ewing & Cervero, 2001). They also found that land use patterns may have a greater impact on mode choice, and, hence, transit usage is dependent on local densities and mixed-land use. However, a more recent and updated systematic meta-analysis by the same authors showed that the relationships between built environment and travel variables are inelastic; however, interestingly, they proposed that the effect of a combination of built environment variables on travel could be significant (Ewing & Cervero, 2010). Further, they expanded the "three Ds" originally coined by Cervero and Kockelman (1997) to "six Ds" by proposing that destination accessibility, distance to transit, and demand management may be used as measures of built environment.

In a study that seeks to understand the impact of urban expansion patterns on social and environmental costs, Camagni et al. (2002) used the disaggregated travel time and mode information for commuters from 1991 census data for Milan, Italy, and socio-economic variables including population

density, ratio of jobs to resident population, and four types of urban morphology variables to categorize urban form. They used regression modeling to find associations between measures of transit efficiency including competitiveness of transit (ratio of average travel time by private transport and average travel time by public transport) and transit mode shares, and urban form and built environment variables. Using this analysis, they found that urban settlements of compact structure lead to greater competitiveness and use of public transport and lower demand for mobility. In a similar type of study, that tried to understand the impact of built environment on non-work travel, Rajamani et al. (2003) combined travel survey data for the Portland (Oregon) metro area with mode-specific travel cost and travel time data to create an accessibility index. They combined all these data sources with the urban land use data for the study region to create urban form measures. Using a multinomial logit model, they found that mixed-use urban forms promote walking behavior for non-work activities.

Stone et al. (2007) used a population reassignment technique, that resulted in higher share of urban and suburban population as opposed to rural population, to simulate the impact of compact growth on travel and transport emissions. They found that an average increase in population density by 14% results in 6% lower median VMT. More recently, Hankey and Marshall (2010) used the Monte Carlo approach to simulate the impact of six different low- and high-sprawl scenarios along with three different vehicle and fuel technology scenarios, ranging from business-as-usual to a "green fleet" comprising both fuel-efficient and electric vehicles, on the vehicle-kilometers traveled for the urban regions in the US. They reported that greenhouse gas (GHG) mitigation strategies that do not consider urban form will not be effective, as they lead to increase in VMT; on a related note, they also state that focusing purely on compact urban forms without consideration to fuel technology improvements may lead to higher emissions. Thus, overall, a few elements of built environment and urban form impact the transport choices and travel characteristics of urban residents.

2.4.2 Interactions between Urban Form and Air Quality

Since urban form impacts travel characteristics, it also could potentially influence transportrelated emissions and concentrations. Specifically, Lyons et al. (2003) used the vehicle kilometers

traveled (VKT) from a few urban centers across the world as a surrogate for vehicular emissions and found that cities that restricted outward growth/sprawl have lower transport emissions, including CO and NO_x. In a study investigating the interactions between land use and health, Frank et al. (2006b) reported that an increase in walkability index, which incorporated land use mix, street connectivity, net residential density, and retail floor area ratios, reduced the VMT and emissions of NO_x and volatile organic compounds (VOCs). Following this, using a population reassignment technique, Stone et al. (2007) simulated emissions in compact urban forms and found higher elasticities between population density and vehicular emissions. Specifically, they found that median drop in pollutant emissions for the compact scenario as opposed to the business-as-usual scenario is 6%, 5.6%, 5.6%, 5.2% for PM_{2.5}, NO_x, CO, and VOCs, respectively.

Following this, Stone (2008) attempted to quantify the associations between urban form measures and air quality by studying the urban form and air quality characteristics of 45 large US metropolitan regions; urban form measures that were considered include centeredness, connectivity, population density, land use, and sprawl index. The study found that, of all the urban form measures of interest, population density is the only measure strongly correlated with the emissions of ozone precursors i.e., NO_x and VOCs; however, the study also reported a much stronger correlation between most urban form measures and annual ozone exceedances. Specifically, centeredness, connectivity, population density, and sprawl index were found to be significantly associated with annual ozone exceedances (Stone, 2008). Marshall et al. (2009) investigated the association between walkability, which provides a measure of mixed-use neighborhood and the connectivity between destinations, and estimated concentrations of ozone and nitric oxide for 49,702 postal codes in Vancouver, British Columbia, and found that urban centers were generally representative of high walkability and high nitric oxide concentrations but low ozone concentrations. Additionally, they found that locations near the urban centers (but not at the urban centers) had reasonably high walkability but lower pollutant levels. In a similar multi-location study, Schweitzer and Zhou (2010) used air quality monitoring data from 80 metropolitan regions to understand the relation between ozone concentrations and urban form. Using the Smart Growth America (SGA)

index scores developed by Ewing et al. (2002) to characterize the urban form, they found evidence for significantly lower ozone concentrations in compact regions but could not find associations between regional compactness and fine particulate concentrations.

Bereitschaft and Debbage (2013) used the urban sprawl indices developed by El-Nasser and Overberg (2001), Ewing et al. (2002), Lopez and Hynes (2003), and Sutton (2003), along with urban form variables and meteorological variables to measure the degree of association between urban form/sprawl and ambient non-point source emissions or/and concentrations of O₃, VOCs, NO_x, PM_{2.5}, and CO₂. The two most important composite measures of urban form are urban continuity and shape complexity. Urban continuity refers to the physical continuity and connectedness of land patches across the urban form or landscape (Bereitschaft & Debbage, 2013; McGarigal et al., 2002), and shape complexity describes the shape and provides a measure of the irregularity or "raggedness" of the land patches (Bereitschaft & Debbage, 2017). They found that a 1 standard deviation increase in urban continuity led to a 9% reduction in annual VOC emissions; conversely, a 1 standard deviation increase in shape complexity led to 8.7% and 12.4% increases in NO_x and PM_{2.5} emissions, respectively. The study also found that higher urban sprawl levels were significantly associated with high concentrations of O₃ and PM_{2.5}; additionally, residential density was found to be a good predictor of O₃ and PM_{2.5} concentrations (Bereitschaft & Debbage, 2013).

Recently, Yu and Stuart (2017) investigated the impact of compact growth and electric vehicles on future air quality and exposure levels for a seven-county region including the Tampa Bay area. They found that the regional emissions under a compact growth scenario were lower compared to a sprawl growth scenario, although this effect was more pronounced for NO_x as opposed to benzene and butadiene. Additionally, they found that the spatial distribution of the difference in concentrations between the compact and sprawl growth scenarios varied by pollutant type. NO_x concentrations (both annual average and maximum one-hour concentration) in the compact scenario were lower than that of the sprawl scenario. However, benzene and butadiene concentrations under the compact growth scenario were

higher than those of the sprawl growth scenario for a substantial portion of Hillsborough County; this is due to the increase in the population and land development density under the compact scenario.

2.4.3 Interactions between Urban Form, Exposures, and Human Health

Patterns of urban forms that increase dependence on motorized travel may have a direct negative implication on human health through increased pollutant exposures (Frank & Engelke, 2001; Frumkin, 2002). Within the context of Europe, De Ridder et al. (2008b) simulated the impacts of sprawl on air quality and found that sprawl leads to higher concentrations of traffic pollutants and exposures. They allocated 12% of their urban population to the green periphery and found that this population reallocation resulted in a higher traffic volume (17% increase), higher concentrations of ozone and particulate matter (approximately 4% increase), and a 0.5% increase in domain-average exposure (De Ridder et al., 2008b).

In the US, Clark et al. (2011) used a linear regression approach to identify the relationship between urban form and population-weighted pollutant concentrations for ozone, PM_{2.5}, and other criteria pollutants. The urban form variables in the study included population centrality, road density, jobshousing imbalance, and city shape (a measure of circularity). They found that urban form was associated with air quality to the same degree as climatic factors and that population centrality was associated with lower population-weighted ozone, PM_{2.5}, and aggregate pollutant levels; transit supply was associated with lower population-weighted PM_{2.5} concentrations. This establishes the importance of urban form for air quality considerations. In contrast, they found that population density is associated with higher population-weighted PM_{2.5} concentrations and aggregate pollutant levels (Clark et al., 2011). Similar to this study, Hixson et al. (2009) simulated population-weighted PM_{2.5} concentrations for the San Joaquin Valley under sprawl and compact urban forms and found that exposures to primary PM components including elemental carbon and organic carbon was increased by 10-15% and reduced by 11-19% for the high-density and low-density scenarios, respectively. Additionally, they reported a reverse in trend (low exposures in high-density development and high exposures in low-density development) for secondary PM components including nitrate and ammonium ion. In Tampa Bay, Yu and Stuart (2017) found that compact growth results in lower population-weighted NO_x exposure concentration than the sprawl
scenario; however, exposure concentrations for benzene and butadiene under a compact scenario were found to be higher compared to a sprawl scenario.

Previously, Woodcock et al. (2009) compared the projections of health benefits for the businessas-usual scenario for 2030 with four alternate urban transport scenarios based on both the individual and combined effects of lower-carbon-emission vehicles and increased active travel for London, UK, and Delhi, India. They found that the scenario that used a combination of low-emission vehicles and increased active travel led to the highest health benefits despite an increase in the disease burden from road traffic injuries as a result of more active travel. It should be noted that the benefits from the increased active-travel-only scenario significantly outweigh those of the low-carbon-emission-vehiclesonly scenario. Continuing a similar line of inquiry, Stevenson et al. (2016) used a health impact assessment framework to investigate the impact of mode change due to alternate land use policies on population health, using six different cities that fall on the spectrum of upper-income to lower-middleincome and highly-motorized to rapidly-motorizing as their testbeds. A compact cities model was applied to each of the six cities by increasing the residential density, reducing the distance to transit, and increasing the land use diversity. They found that the compact scenario resulted in overall health gains (420–826 disability-adjusted life-years) by shifting individuals to active modes of travel and reducing transport-related particulate matter emissions for all the cities despite a small increase in road trauma incidents for pedestrians and bicyclists.

In summary, most of the literature focusing on the relationship between urban form and transport suggested that compact urban forms led to a reduction in VMT and an increase in active and non-auto mode of travel. Additionally, studies found evidence for lower traffic-related emissions and concentrations in compact urban forms. Despite these improvements in air quality, several studies found that population exposure to traffic-related air pollution increased in compact urban forms. However, a recent macro-level study by Stevenson et al. (2016) found overall health gains due to compact and transit-oriented urban forms. Although these studies provide valuable information toward understanding the relationship between urban form, travel, air quality, and exposures, further studies on this topic are

necessary. First, the research trying to understand the relationship between urban form and population exposure is still budding; as a result, further research efforts are needed on this topic. Second, many studies on this topic used observed data from air quality monitors and existing urban morphologies to understand the linkages between urban design and population exposure. These studies generally may be incapable of answering what-if urban-policy questions; hence, there is a need for studies that use modeling techniques to simulate alternate urban morphologies and estimate their impact on population exposure. Finally, although a few studies used modeling techniques to understand the relationship between urban form and population exposure, they are either macroscopic in nature or used nonbehavioral and coarser modeling techniques. Given this, my disertation focused on using a highlyspatiotemporally-resolved modeling framework to estimate population exposure under alternate urban forms.

2.5 Characterization of Individual Exposure to Traffic-Related Pollution

The characterization of individual exposure to traffic-related pollution has two essential components—the characterization of the spatiotemporal distributions of pollutant concentrations and activity and travel patterns of individuals. In this section, the methodological aspects of characterizing pollutant concentrations is presented, followed by discussion on the current state of knowledge on the characterization of activity and travel patterns. Finally, combining the information provided in the first two sections, methodological details of exposure characterization are addressed.

2.5.1 Characterization of Spatiotemporal Distributions of Pollutant Concentrations

2.5.1.1 Ambient Air Sampling

Direct sampling or measurement of pollutants is perhaps the "gold standard" in accurately characterizing the spatiotemporal distributions of pollutant concentrations. The samplers that are used for direct measurement can be broadly categorized as active and passive samplers (International Agency for Research on Cancer, 2016), and the choice of measurement or sampling device would normally depend on the type of the pollutant and the parameters of the research study.

Previously, studies conducted a wide variety of sampling campaigns to obtain the spatiotemporal distributions pollutant concentrations. For example, Stuart and Zeager (2011) used Ogawa passive samplers to measure ambient NO_2 concentrations near 75 randomly-selected elementary schools in Tampa. Similarly, to estimate human exposures, other studies measured the micro-environmental concentrations, i.e., concentrations within a small area such as a bedroom, a kitchen, or an office using a variety of active and Palmes passive sampler tubes (Kornartit et al., 2010; Lai et al., 2004; Rotko et al., 2001).

Although direct measurement results in highly-accurate characterization of spatiotemporal distributions of concentration, the measurement campaigns entail higher costs. Additionally, notwithstanding the high temporal resolution, the spatial resolution of the measured concentrations can be limited owing to the discrete placement of the samplers. Finally, although these approaches help in understanding the existing air quality trends within an urban area, they are rather limited in addressing the implications of new regulatory policies or pollution sources on air quality (Vallero, 2008). Thus, these methods are ill-equipped to answer the implications of "what-if" scenarios on air quality. As a result, instead of direct measurement, several studies have taken the alternate route of modeling pollutant concentrations.

2.5.1.2 Ambient Air Modeling

A wide variety of models were employed in the past to estimate the spatiotemporal distributions of pollutant concentrations (Brugge et al., 2007; Burke et al., 2001; Jerrett et al., 2005). Air pollution models can be broadly categorized into two types—models that try to characterize atmospheric concentrations by performing statistical analysis of data or by simulating the fundamental physical and chemical interactions within a system (Seinfeld & Pandis, 1998). Since this study aimed to understand and predict the impact of urban form on transportation emissions and exposures, it focused on models that simulate the dispersion and transport of pollutants.

Atmospheric chemical transport models estimate pollutant concentrations at particular times and locations (receptors) using pollutant emission inventories and meteorological data. As mentioned, these

models simulate pollutant concentrations by explicitly accounting for the physical and chemical interactions within the system. Although different solution approaches are used, these models are generally based on the fundamental principle of conservation of the mass of species in a control volume of solution and are generally represented by equation 2.1.

$$\begin{bmatrix} rate \ of \ accumulation \\ of \ mass \ of \ A \end{bmatrix} + \begin{bmatrix} net \ rate \ of \ outflow \ of \\ mass \ of \ A \ from \ volume \end{bmatrix} = \begin{bmatrix} rate \ of \ generationg \ of \\ mass \ of \ A \ by \ reaction \end{bmatrix}$$
2.1

A common expression of this balance at an infinitesimal point is given by the advectivedispersion expression shown in equation 2.2 (Ramaswami et al., 2005).

$$\frac{\partial C}{\partial t}(\vec{x},t) + \underbrace{\vec{u} \cdot \nabla C}_{advection} = \underbrace{\nabla \cdot D\nabla C}_{dispersion} + \underbrace{R}_{reaction} + \underbrace{S}_{source}$$
 2.2

Many models, using a variety of approaches, were developed to operationalize the solution of this mass balance. Some commonly-used models in the US include CMAQ, CAMx, and HySplit.

2.5.1.3 Gaussian Formulation

One category of solution approach with a long history of being used for pollutant concentration estimation is the Gaussian approach. Gaussian models assume that the pollutant mass spreads in the horizontal and vertical directions following a Gaussian or normal distribution. Additionally, many of these models, called Gaussian plume models, are based on analytical solution of the steady-state advective-dispersion equation. The plume equation estimates the mean concentration from a continuous point source of pollution (assuming reflection at surface) as shown in equation 2.3.

$$C(x, y, z) = \frac{q}{2\pi u \sigma_y \sigma_z} exp\left[-\frac{y^2}{2\sigma_y^2}\right] \left[exp\left[-\frac{(z-h)^2}{2\sigma_z^2}\right] + exp\left[-\frac{(z+h)^2}{2\sigma_z^2}\right]\right]$$
2.3

where,

C is the pollutant concentration (μ/L^3)

x, y, and z represent the downwind, crosswind, and vertical position, respectively, of the receptor away from the source base location

q is the emission rate of the pollutant (M/t)

u is the horizontal wind speed in the downwind direction (L/t)

 σ_v describes the cross wind dispersion in length (L)

 σ_z describes the vertical dispersion in length (L)

h is the effective height of release (L)

Changes in concentration in time (e.g., each hour) are estimated from this equation due to changes in the parameters (q, u, σ_y , σ_z , h) with time. Examples of Gaussian plume models include AERMOD, CALINE, and R-LINE. R-LINE is used for air pollution modeling in most of this dissertation. A related approach that relaxes the steady-state assumption and accounts better for rapidly-varying changes in parameters uses a Gaussian puff solution; CALPUFF (California Puff) is an example. Results from air pollution modeling using CALPUFF are used for the analysis in Chapter 3.

2.5.2 Characterization of Spatiotemporal Distributions of Individual Activity and Travel Patterns

Human activity patterns (or activity and travel patterns) refer to the activity distributions of individuals within a certain period of time. Activity and travel patterns describe the movement of individuals in a geographic area by predominantly collecting information related to the activity locations, time and duration of activities, travel mode, travel times and distances, and travel routes between the fixed-activity locations. Characterization of activity and travel patterns is important from an exposure estimation perspective because the physical location of the activity, the time and duration of the activity, and the travel path between activity locations influence the personal exposure to pollutants. A variety of sample-based approaches use computer-assisted telephone interview, travel surveys or diaries, Global Positioning System (GPS) loggers, and home sensors to collect the spatiotemporal distributions of activity and travel patterns of individuals.

2.5.2.1 Sample-Based Activity and Travel Surveys

Several studies used a travel survey or diary format to obtain the spatiotemporal distributions of individual activity and travel patterns. Axhausen et al. (2002) conducted a six-week travel diary-based survey to observe the daily activity and travel routines of individuals in two German cities. The survey collected trip-level information including day of trip, start and arrival times, trip purpose, travel modes, trip destination, and activity and travel costs for 319 individuals from 139 households. To characterize

the daily personal travel patterns of individuals across the nation, the 2009 national household travel survey (NHTS) was conducted using a computer-assisted telephone interview instrument (Santos et al., 2011). The 2009 NHTS collected household and person-level demographic information and trip-level information including origin and destination location and time, length, mode of travel, and purpose for approximately 125,000 households.

Recent studies used advances in sensor-based technologies to record the activity and travel patterns of individuals. Specifically, in a study aimed at testing the feasibility of GPS technology to track individuals, Wiehe et al. (2008) used GPS-enabled cell phones to collect the activity and travel patterns of 15 adolescent women for one week. Although they found that user error and technical issues could affect the reliability of the activity and travel data, the authors argue that GPS technology provides a feasible way for collecting activity and travel data that can be used to identify health-risk behaviors (Wiehe et al., 2008). Similarly, Abdulazim et al. (2013) developed an Android application to collect the location data for individuals and land use data to estimate the travel mode based on the motion pattern as indicated by the cell phone's sensors. Such advanced applications can assist in collecting activity and travel information for a large section of the population in an economically feasible way.

Some of the advantages of using travel diary information are availability of larger sample sizes, detailed representation of spatial and temporal locations of individuals, and wide availability of such data resources including the NHTS, Consolidated Human Activity Database, and California Activity Pattern Surveys. Although the sample sizes from the travel diaries are considered to be fairly large, they typically are not more than 1% of the population. However, larger samples may be required to capture the intra-urban spatial variations in the activity patterns necessary for the estimation of population exposures. As mentioned, the sensor-based technologies could potentially solve the issue of low sample sizes owing to their ability to collect large samples at relatively low expense; however, these technologies are still budding and the gathered sensor data is primitive and heterogeneous, leading to difficulty in assimilating activity data from various sources (Chen et al., 2012). In addition, survey or sensor-based data cannot be used directly to forecast the activity and travel patterns under alternative policy scenarios. For example,

the raw survey data collected in a specific year may not give an indication of the activity and travel patterns of individuals in a future year where a transit-oriented compact growth policy may be implemented. In such alternate policy scenarios, predictive models can be used to simulate the activity and travel patterns of the population. It should be noted that survey data are still invaluable because the predictive models will be built based on the survey data.

2.5.2.2 Modeling the Activity and Travel Patterns of Individuals

In the transportation field, travel demand models were used for several decades by the research community, metropolitan planning agencies, and consulting firms to forecast the aggregate travel patterns of metropolitan residents under alternative policy and investment scenarios. Traditionally, the travel demand of an urban area has been modeled using a four-step or trip-based travel demand model. The fundamental unit of analysis in the trip-based travel models is an individual person trip (Castiglione et al., 2015). The modeling paradigm of a trip-based model consists of four steps—trip generation, trip distribution, mode choice, and network assignment. Trip generation estimates the number of trips produced from and attracted to each zone in an urban context. Trip productions and attractions are generally modeled using a linear regression approach and can be modeled at a disaggregate level for every household and usually are a function of the characteristics of the households, land use, and transportation system. Trip attractions are modeled at the zonal-level and are normally a function of the zonal characteristics. The general formulations for the trip production for a household h in zone i (P_{bi}) and all the households in zone i (P_i) are presented in equations 2.4 and 2.5, respectively.

$$P_{hi} = \beta_0 + \sum_{i=1}^k \beta_i x_i \tag{2.4}$$

Here, $x_1, x_2, ..., x_k$ are the factors affecting trip generation, and $\beta_0, \beta_1, ..., \beta_k$ are the estimated parameters that capture the effect of an independent variable on the trip production rate. To obtain the trip productions at the zonal level, the disaggregate trip productions for all the households in the zone need to be summed up.

$$P_i = \sum_{h=1}^{N} P_{hi} \tag{2.5}$$

Here, N is the total number of households in the zone i. The trip attractions for the zone i (A_i) are estimated using an equation similar to 2.4, the only difference being the use of zonal-level characteristics including retail employment, commercial employment, gross floor area of retail activity.

Following the estimation of number of trip productions and attractions for each zone, the trips are allocated between the origin and destination zones using a gravity model shown in equation 2.6.

$$T_{ij} = P_i \left[\frac{A_j F_{ij}}{\sum_k A_k F_{ik}} \right]$$
 2.6

Here, T_{ij} is the total number of trips between production zone i and attraction zone j. P_i is total number of trips produced in zone i. A_j is the total number of trips attracted to zone j. F_{ij} is called the friction factor and captures the travel impedance, generally in the form of travel time or cost, between zones i and j.

Following the trip distribution, trips between different zones are allocated to one of the available travel modes in the mode choice step. Mode choice typically relies on the use of utility functions to estimate the travel mode shares; the utility of a mode alternative is dependent on the characteristics of the travel mode and the individual making the choice. A more detailed discussion of the utility-based models is provided under the activity-based travel demand model section below. Once the utility (U) for a travel mode (k) is calculated, the travel share for that mode (i.e., the proportion of travelers using mode k) is obtained using a multinomial model structure as shown in equation 2.7 where x represents all the available modes.

$$Pr(k) = \frac{e^{U_k}}{\sum_x e^{U_x}}$$
 2.7

Finally, the network assignment step allocates the trips to the traffic network and determines their travel path. The traffic assignment step is generally performed in multiple iterations by assigning trips to roadway links that form the shortest time-path and shifting them to non-congested links when the original roadway links to which they are assigned becomes congested. The process is repeated over multiple iterations until equilibrium is achieved between the travel demand and the supply within a tolerance.

Although trip generation models provide valuable information in terms of the current and future travel demand needs of a region, they are rather limited in their sensitivity towards proposed policy changes. This is because trip-based models consider each trip to be independent of the other (Castiglione et al., 2015; Pinjari & Bhat, 2011). However, in reality, many trips are interrelated. For example, in a household with two working adults who work at different locations and only one car, the individuals might have to coordinate their commute; it is easily conceivable that the travel patterns of this household will be different to that of another similar household with two available cars. Thus, clearly, factors including work location choice, mode choice, and mode availability influence the interrelations between trips, but the trip-based models ignore them. Trip-based models also suffer from aggregation bias (Castiglione et al., 2015; Pinjari & Bhat, 2011). These models characterize the travel behavior at an aggregate level by using broad socio-demographic categories and assuming that households belonging to similar categories behave similarly. For example, the trip generation rate for all households with two working adults and one child living in the same zone is considered to be the same. Thus, there is no way to account for the impacts of socio-demographic characteristics of individuals on the travel demand. Finally, trip-based models do not explicitly account for the temporal dimension of activities. Time is simply treated as the "cost" of making the trip without actual consideration for how participation in activities and travel takes away from the daily allocated time budget (Pinjari & Bhat, 2011). Thus, since these models are insensitive to time-of-day and scheduling choices, they are incapable of simulating the impacts of policies with time-of-day attributes.

Considering these limitations in the trip-based models, the travel demand field has moved towards the "behaviorally-oriented activity-based approaches for modeling passenger travel demand" (Pinjari & Bhat, 2011). These new streams of activity-based travel demand models try to characterize the travel using behaviorally-realistic paradigms. Whereas the trip-based models focus on trips without recognizing the need for those trips and travel, the activity-based approach views travel as a derived demand resulting from individual needs to participate in activities (Bowman & Ben-Akiva, 2001; Pinjari & Bhat, 2011). More specifically, instead of estimating independent trips, the activity-based models

focus on individual decisions to participate in activities and the scheduling and location of those activities along with the modeling of travel mode choices (Kitamura et al., 1997). As a result, activity-based models estimate chains of trips, also known as tours, as part of characterizing individual daily activity patterns (Castiglione et al., 2015). Thus, activity-based travel demand models are more sensitive to policy changes that affect the choice behavior of individuals which in turn affects their travel behavior.

Activity-based travel demand models allow for the estimation of activity and travel patterns at a disaggregate level. Specifically, they can be used to simulate the daily activity and travel patterns for each and every representative individual in the population of interest. The estimated individual-level activity and travel patterns include information on the types of daily undertaken activities, the spatial locations and timing of these activities, and the timing, mode, and routes of travel to these activities (Bradley & Bowman, 2006; Pinjari et al., 2006; Pinjari et al., 2008).

Activity-based travel demand approaches can be broadly categorized into two types—utility maximization-based and rule-based computational process systems. Utility maximization-based models operate under the philosophy that individuals choose only those activity and travel choices that maximize their utility. Here, utility can be thought of as a satisfaction derived from participation in activities. These systems use a suite of utility maximization-based discrete choice structures including multinomial logit and nested logit models, as well as other econometric structures including hazard-based duration models and ordered response models, to simulate individual-level activity and travel choices. For example, the travel mode choice module in an activity-based modeling framework may first estimate the utility of the choice alternative and then employ the multinomial logit structure (as shown in equations 2.7 and 2.8) to estimate the mode choice for an individual. Prime examples of utility maximization-based models include the Comprehensive Econometric Microsimulator for Activity-Travel Patterns (CEMDAP) (Pinjari et al., 2008) and DaySim (Bradley et al., 2010). In contrast, the rule-based computational process models use context-dependent and adaptive choice heuristics to estimate the activity and travel choices of individuals (Arentze & Timmermans, 2004; Pinjari & Bhat, 2011). In other words, instead of assuming that individuals always seek to maximize their utility, these models use an exhaustive set of if-then

condition-action rules to build daily activity and travel schedules for individuals. These condition-action rules generally are based on the observed activity and travel behavioral choices from travel surveys. Examples of rule-based computation process models include A Learning-BAsed TRansportation Oriented Simulation System (ALBATROSS) (Arentze & Timmermans, 2004) and Travel Activity Scheduler for Household Agents (TASHA) (Miller & Roorda, 2003). Since this study uses a utility maximizationbased travel demand model, further details on these model systems are presented in the next section.

2.5.2.3 Utility Maximization-Based Travel Demand Models

Utility maximization-based travel model systems are based on the economic theories of consumer choice, which suggests that individuals choose the activity and travel choices that maximize their utility. The utility function for each choice alternative is dependent on the characteristics of the individual, choice alternative, and the interaction between them as shown in equation 2.8 (Koppelman & Bhat, 2006).

$$V_{t,i} = V(S_t) + V(X_i) + V(S_t, X_i)$$
 2.8

where,

 $V_{t,I}$ is the utility of alternative i for individual t

 $V(S_t)$ is the portion of utility associated with characteristics of individual t

- (X_i) is the portion of utility of alternative i associated with the attributes of alternative i
- $V(S_t,X_i)$ is the portion of the utility which results from interactions between the attributes of alternative i and the characteristics of individual t.

The general form of utility is linear and is represented using equation 2.9, where $S_{t,1}$, $S_{t,2}$... are the characteristics of the trip maker and $X_{1, ..., r}$, Xr are the attributes of the choice alternatives. $a_{1, r} a_{2, ...} a_{r}$ are the parameters (or weights) defining the utility function.

$$Vt, i = a_1 S_{t,1} + a_2 S_{t,2} + \dots + a_r X_r$$
2.9

Thus, according to the theory of utility-maximization an individual t chooses the alternative i over a set of all other alternatives j if $V_{t,i} \ge V_{t,j} \forall j$. This gives rise to a deterministic system of equations that one may solve simultaneously to determine the activity and travel choices of individuals. However, errors pertaining to incomplete or unavailable information may result in individuals choosing an alternative with lower utility value. These unobserved factors are incorporated into the utility function as an error (or unobserved) term and the utility definition is updated as shown in equation 2.10 (Koppelman & Bhat, 2006). This gives rise to a probabilistic model.

$$U_{it} = V_{it} + \varepsilon_{it}$$
 2.10

where,

 U_{it} is the true utility of the alternative i to the decision maker t

V_{it} is the deterministic or observable portion of the utility estimated by the analyst

 ε_{it} is the error or the portion of the utility unknown to the analyst.

Assumptions regarding the distribution of the error term results in different types of model structure. Generally, the errors are assumed to be identically and independently distributed, and follow an extreme-value (gumbel) distribution across the choice alternatives and individuals. This results in a multionomial logit model structure where the probability of choosing an alternative i is given as shown in equation 2.11.

$$Pr(i) = \frac{e^{V_i}}{\sum_j e^{V_j}}$$
2.11

The multionomial logit structure fails when there is a significant correlation among the choice alternatives (since the model assumes that alternatives are independent). In such a case, the alternatives are arranged into nests as shown in Figure 2.2 with similar alternatives placed into one nest. For example, the transit options including bus and light rail may be placed under a single nest since they share the attributes of public transit.



Figure 2.2 Nested mode choice model structure

In the nested logit structure, the probability of choosing an alternative i (under nest n) is conditional upon first choosing the nth nest from a total of m nests and then choosing the ith alternative under the nth nest as given by equations 2.12 and 2.13 (Bowman & Ben-Akiva, 2001; Castiglione et al., 2015; Koppelman & Bhat, 2006). Here θ_n is a dispersion term that accounts for the correlation within a nest.

$$pr(i) = pr(i|n) * pr(n)$$
2.12

$$pr(i) = \frac{\exp(V_{i|n}/\theta_n)}{\sum_{j \in n} \exp(V_{j|n}/\theta_n)} * \frac{\exp\left[V_n + \theta_n \ln\left(\sum_{j \in n} \exp(V_{j|n}/\theta_n)\right)\right]}{\sum_{\forall m} \left[\exp\left[V_m + \theta_m \ln\left(\sum_{j \in m} \exp(V_{j|m}/\theta_m)\right)\right]\right]}$$
2.13

Although there are certainly other types of econometric modeling structures in use in activitybased travel demand modeling systems, the multinomial logit and nested logit models are perhaps the most widely used. Using these econometric model structures, the activity-based travel demand models estimate the daily activity and travel behavior of individuals given their demographics, urban land use characteristics, and highway and transit characteristics.

2.5.3 Characterization of Individual Exposures to Traffic-Related Pollution

Individual daily exposures to pollutants are dependent on the magnitude of pollutant concentrations, duration of exposure, and the frequency of exposure (Klepeis et al., 2001; Setton et al., 2011); thus, spatiotemporal characterizations of pollutant concentrations and activity and travel patterns of individuals are important. Epidemiologic studies focusing on exposures to traffic-related pollution used two broad categories of surrogates—measured or modeled concentrations of surrogate pollutant and direct measures of traffic including proximity or distance to the roadway and roadway traffic volumes (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010; Jerrett et al., 2005).

Exposure estimation based on proximity measures, or volumes, are by far the easiest to implement. However, they do not robustly account for factors including fuel type and vehicle mix (Gertler, 2005), and meteorology and terrain (Baklanov et al., 2007; Rijnders et al., 2001) that impact pollutant concentrations and, hence, exposures. As such, studies tried to estimate personal exposures to traffic pollution using a variety of measurement and modeling approaches.

Traditionally, air pollution exposure analysis has been simplistic with regard to the consideration of human activity patterns for exposure estimation. An individual's exposure often is estimated by considering the pollutant concentration at the individual's residential location, also called residence-based exposure, or at a fixed monitoring station (Cortese & Spengler, 1976; Huang & Batterman, 2000). These methods could lead to biased exposure estimates owing to the disregard of spatiotemporal concentrations near the individuals' non-residential activity locations. Realizing this limitation, studies were undertaken to characterize individual daily time use and diurnal activity patterns (Health Canada, 2010; Klepeis et al., 2001; National Exposure Research Laboratory et al., 2000). Among these, the National Human Activity Pattern Survey (NHAPS) (see Klepeis et al. (2001) is a widely-used source of data for pollutant exposure analysis in the US (Burke et al., 2001; Dong et al., 2004). NHAPS used a 24-hour activity diary to collect information pertaining to individual activity participation (type, timing and the duration of activity) and the microenvironment location type (kitchen, bedroom, office, grocery store, etc.) for the activities over one day. Whereas the detailed information collected by NHAPS and similar surveys is of immense value, the small sample size and the lack of detailed geographic coordinates for the activity locations pose an obstacle for exposure characterization. Some additional concerns include issues of representativeness of the activity data for intra-urban exposure applications and temporal validity of activity patterns especially for exposure forecasting purposes.

Recognizing these limitations, a few studies tried to conduct personal monitoring campaigns to accurately estimate exposures to traffic pollution. For example, to characterize individual exposure to NO₂, Kousa et al. (2001) and Kornartit et al. (2010) used Palmes tubes. Similarly, Good et al. (2016) used backpack-based measuring instruments to obtain spatiotemporal distributions of activity and travel patterns of individuals and personal exposures to a variety of pollutants. These personal monitor-based exposure measurement campaigns are, by far, the most accurate methods to estimate the personal exposures; however, large-scale deployment of personal monitors may not be feasible due to the high expenses involved (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010; Jerrett et al., 2005). A few studies also measured the micro-environmental concentrations (for example, bedroom, kitchen, or office) in an effort to characterize personal exposures (Kornartit et al., 2010; Lai et al., 2004; Rotko et al., 2001). In these studies, the microenvironments where individuals typically spend a large portion of their day were chosen for measurement. However, this method offers limited spatial variability and may only be pertinent to the microenvironments under study. Further, this method also can be cost prohibitive similar to the personal-monitoring approach (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010).

To overcome these limitations, a few studies pioneered the use of activity and travel patterns obtained from travel surveys in conjunction with modeled pollutant concentrations to estimate personal exposures (Kornartit et al., 2010; Marshall et al., 2006). Marshall et al. (2006) used an activity and travel survey data of about 25,000 individuals in southern California to compute the pollutant exposures. Although these travel survey-based studies have a fairly large sample size, the spatial variability of the activity and travel patterns of this sample may not be representative of the population. Additionally, this sample-based approach cannot be realistically used to estimate exposures under a hypothetical policy scenario.

Thus, to overcome all of the earlier-mentioned issues, a few recent studies chose the alternate route of modeling both pollutant concentrations and activity and travel patterns of individuals to estimate personal exposures for the entire population. Hatzopoulou and Miller (2010) used an activity-based travel

demand model (TASHA) in conjunction with a mobile source emissions estimation model (MOBILE 6.2C) and dispersion model (CALPUFF) to estimate personal exposures to NO_x for the Greater Toronto area's population. Similarly, Beckx et al. (2009a)), (Beckx et al., 2009c), and Beckx et al. (2009d) used the activity-based model ALBATROSS in combination with the emissions model MIMOSA and pollutant dispersion model AURORA to estimate personal exposures to NO₂ in the Netherlands. Recently, Vallamsundar et al. (2016) used a similar paradigm within the context of the US to estimate the population exposure of Maricopa County, Arizona, to PM_{2.5}. Although these studies certainly laid the groundwork and provided advances for population exposure estimation methodologies using activity-based modeling, their spatial resolution is generally coarse. This is of specific interest when dealing with pollutants such as NO_x that display a high spatial variability at the urban scale. However, more importantly, previous research efforts did not fully exploit the features of activity-based models to simulate the activity and travel patterns and the resulting population exposure under alternate urban design scenarios. Given this, my dissertation focused on creating a modeling framework that combines an activity-based travel demand model with emissions estimation and dispersion models to forecast high resolution activity and travel data and population exposure under alternative urban design scenarios.

2.6 Conclusion

In view of the previous discussion, a clear gap in knowledge exists with regard to the overarching question "how can we design cities so that population exposures and exposure inequalities can be mitigated?" First, although several studies showed the prevalence of exposure inequalities by ethnicity and socioeconomic status, very few studies look at the associations between other social and urbanicity variables of interest. Second, it would be of specific interest to understand if the exposure inequalities investigated the relationships between urban form and air quality, few to none have considered the interactions between them using micro-simulation models that specifically model activity and travel patterns of individuals. This is of particular interest because whereas some studies showed larger influence of urban form on exposures to traffic-pollution, many showed marginal effects. A primary reason for this could be

the non-availability of tools that can realistically predict changes in human activity and travel behavior as a function of changes to the built environment. Fourth, past research efforts that used activity-based travel demand modeling have not fully exploited them to simulate the activity and travel patterns of individuals under different transportation and land use design scenarios. Considering these limitations, this study aimed to understand the relationship between urban design features, human activity and travel patterns, air quality, and population exposure using highly spatiotemporally-resolved activity-based travel demand modeling and air pollution emission and dispersion modeling.

CHAPTER 3: IMPACTS OF TRAVEL ACTIVITY AND URBANICITY ON EXPOSURES TO AMBIENT OXIDES OF NITROGEN AND ON EXPOSURE DISPARITIES

3.1 Note to Reader

This chapter was published in *Air Quality, Atmosphere & Health*. (Gurram, S., Stuart, A.L. & Pinjari, A.R. Air Qual Atmos Health (2015) 8: 97. https://doi.org/10.1007/s11869-014-0275-6). The article is available online at http://link.springer.com/article/10.1007/s11869-014-0275-6. Permission is included in Appendix A.

3.2 Introduction

Estimation of human exposures to air pollution is important to researchers and practitioners in the fields of air quality management, environmental epidemiology, and urban design. Exposure estimation requires characterization of pollutant concentrations when and where a person or group spends time (Ott, 1982). Although personal monitoring has long been used to determine exposures in the field of air pollution epidemiology (Dockery & Spengler, 1981), it is time- and cost-intensive, resulting in small sample sizes that may be limited for representing a general population (Jerrett et al., 2005; Pekkanen & Pearce, 2001). Hence, methods of estimating exposures for a large group of people are needed for population-level risk assessment and policy decisions.

For large-sample studies, exposures to air pollutants often have been estimated using residence address to represent the location of exposure. Concentrations measured at fixed monitoring sites or concentration surrogates (such as nearby traffic counts) are used to derive exposures at the residence locations (Huang & Batterman, 2000; Meng et al., 2007; von Klot et al., 2009). Although this is a relatively simple and generalizable approach that can be applied in the context of available data, it is recognized that human activity patterns may be particularly important for explaining exposure variation (Klepeis et al., 2001; National Center for Environmental Assessment et al., 2011; Ott et al., 1986).

Hence, exposure error and misclassification are concerns, with potential outcomes of inaccurate health and environmental impact assessments and policy interventions (Huang & Batterman, 2000; Krzyzanowski, 1997; Özkaynak, 1986; Sheppard et al., 2012; Thomas et al., 1993; Zeger et al., 2000).

As a result, studies have investigated the use of more refined estimates of population location and concentration to represent personal exposures through methods that characterize or apply patterns of human activity (e.g., from time activity diaries) and microenvironment concentrations (Dons et al., 2011b; Kornartit et al., 2010; Lai et al., 2004). These methods often improve the estimate of group-level and personal exposures, but remain substantially limited in the characterization of spatiotemporal variations in concentrations and activities. A few recent case studies have used detailed activity and travel patterns derived from travel surveys or activity-based models coupled with air pollution modeling to estimate air quality exposures or health impacts (Dons et al., 2014; Gariazzo et al., 2011; Hankey et al., 2012; Hatzopoulou & Miller, 2010), including analysis of exposure error (Dhondt et al., 2012; Setton et al., 2009b; de Nazelle et al., 2013; Zhang & Batterman, 2013), urban form (Stone et al., 2007), and transportation policies (Dhondt et al., 2013). Nonetheless, the literature remains sparse, and additional case studies applying and improving these methods are needed. Additionally, limited literature exists on the social distribution of exposure error.

This study is part of an ongoing project that aims to enhance the current understanding on exposures to traffic-related air pollution, specifically on the social distribution of exposure and impacts of urban design (Evans & Stuart, 2011; Fridh & Stuart, 2014; Stuart et al., 2009; Stuart & Zeager, 2011; Yu & Stuart, 2013). This study investigates impacts of activities and urban design factors on exposures and exposure disparities and estimates the error introduced by use of residence-location-only versus detailed spatiotemporal activity on exposure estimates. Our methods combine information from an available travel survey, estimated travel routes, and concentration data from air pollution modeling results. The following questions were addressed through this work: How are population activities distributed spatiotemporally in the study domain? How are exposures distributed among population groups in the

study domain? What is the strength and direction of disparities between groups? Does urban form influence the strength of exposures and their social distribution? Which factors are most influential? Are findings robust to uncertainties in exposure estimation associated with the representation of exposure location? How much does the representation of spatiotemporal activity location impact exposure estimates? Are the errors associated with exposure estimation different for different population subgroups? Methods and findings on these questions are detailed below.

3.3 Methods

3.3.1 Study Area and Pollutant Focus

Hillsborough County, Florida, shown in Figure 3.1, is the area of study. It contains a diverse mix of air pollutant emission sources, including an extensive highway network. Further, it has undergone considerable urban sprawl during the past few decades; in 2000, Smart Growth America ranked it as the 22nd most sprawled metropolitan area of 83 with populations over a half million (Ewing et al., 2002). In 2012, the Texas Transportation Institute (TTI) ranked Tampa-St. Petersburg as 30th for congestion (yearly delay per commuter) (Schrank et al., 2012), with automobiles as the primary mode of personal transportation. Regarding measured air quality, ozone levels in the area exceed the National Ambient Air Quality Standard (NAAQS) a few times most years, with particle levels close to the 24-hour standard; the American Lung Association grades Hillsborough County's air quality as "F" for ozone and "C" for particulate matter (American Lung Association, 2011). Further, the county is interesting for social equality reasons, as its population is relatively diverse and somewhat residentially-segregated (Stuart et al., 2009).



Figure 3.1 Study area of Hillsborough County, Florida. Inset shows location of study area within state of Florida.

The pollutant focus herein is oxides of nitrogen (NO_x), which is the sum of nitrogen monoxide (NO) and nitrogen dioxide (NO_2 ,, a US criteria air pollutant with an established standard level). Although, levels of NO_2 measured by regulatory networks rarely exceed the national standard, NO_x is a precursor to both ozone and fine particles. Further, it is a common urban pollutant that has been associated with respiratory responses for susceptible individuals, particularly children, even at levels below the National Ambient Air Quality Standard (US Environmental Protection Agency, 2008). Studies have linked the exposure to oxides of nitrogen with cardiovascular and respiratory mortality (Faustini et al., 2014), gestational diabetes and preeclampsia (Malmqvist et al., 2013), diabetes mellitus and hypertension (Coogan et al., 2012), and incidence of asthma (Anderson et al., 2013). NO_x is also a recognized surrogate in health outcomes analyses for the complex mix of traffic pollution (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010).

3.3.2 Estimation of Spatiotemporal Human Activity and Travel Patterns

Human activity and travel patterns representing the study area were estimated using data from the National Household Travel Survey (NHTS), a periodic survey that characterizes the daily travel behavior of Americans (Federal Highway Administration, 2009). Data are collected on all out-of-home trips taken over approximately a 24-hour period for individuals sampled by the survey. The data collected include the purpose of each trip (work, shopping, recreation, etc.), trip start and end times, travel times, travel distances for each trip, and the geo-coded locations of activities. Socio-demographic characteristics (including age, race/ethnicity, household income, household size, and neighborhood urbanicity) of those surveyed are also collected. Here, we used data from the 2009 survey to characterize spatiotemporal locations of daily activity and travel in Hillsborough County.

The national household travel survey sample for Hillsborough County includes daily activity records for 1582 persons from 804 households. Prior to use, we filtered the sample to exclude daily activity records that were inconsistent or had missing information. We also excluded records that contained travel outside of the county boundaries (beyond which detailed NO_x concentrations were not available). For a few records, it was necessary to pare the data to exactly 24-hours (beginning at 12:00 am). The resulting sample consisted of 1224 daily activity records, including 239 with no travel away from the residence location on the survey day.

To estimate the locations of daily activities in time and space for the county sample, we first extracted data from each individual 24-hour activity record (a person-day). Specifically, we extracted the geocoded residence location (latitude, longitude), origin and destination locations for each trip, trip start times, and dwell times (time spent at the activity location) using SPSS (version 20.0, IBM Corp. Armonk, NY). Since the national household travel survey does not record information on travel path, we estimated the route of travel for each trip. Specifically, we used the Network Analyst tool in ArcGIS (version 10.0, ESRI, Redlands, CA) to select the shortest time path between each trip's origin and destination, based on roadway link times and a network shape file (NAVTEQ, 2010). Travel times for each link were calculated using link lengths and link free flow speeds provided with the network data. Spatial location

coordinates (latitude, longitude) along each trip path were extracted at a discrete interval of 100 meters of path length using the ET GeoWizards tool (version10.2, ET Spatial Techniques, Faerie Glen, South Africa). The temporal location coordinate (time of day) for each discrete spatial location was estimated by adjusting the time on each link by the ratio of the total trip time from the survey data to that from the link time estimate. We then combined the trip path location data to create a highly-resolved sequential spatiotemporal record of estimated activity location for each person-day in the filtered county sample.

3.3.3 Estimation of Diurnal Pollutant Concentrations at Activity Locations

To estimate pollutant exposures for the study sample, we used ambient NO_x results from our previous dynamic CALPUFF air pollution dispersion modeling for the study area. Details of the modeling methods, results, and evaluation are provided in Yu and Stuart (2013). In essence, concentrations were estimated using detailed emissions, including link-level roadway emissions, and meteorological data for 8760 hours (all hours of 2002) for the study area. The results provide estimated concentrations on a receptor grid with 1 km spatial resolution for Hillsborough County. For matching with the daily activity and travel records here, we estimated the diurnal cycle of the spatial distribution of NO_x concentration from the model results by averaging the hourly modeled concentration results at each receptor over each hour of the day.

3.3.4 Estimation of Daily Exposure Concentration and Exposure Error

One goal of this work was to investigate the impact of activity and travel patterns on exposure estimates. To do this, we calculated and compared daily exposure concentrations for each person-day using two methods. Both methods estimate the time-weighted exposure concentration, $C = (1/T) \int c \, dt$, where *c* is the instantaneous pollutant concentration at an exposure location, *dt* is the instantaneous time interval of exposure, and *T* is the total exposure averaging period, which equals $\int dt$ (24 hours for the person-day records here).

The first method uses only the residence location to estimate daily exposure concentration for each person-day. We call this the residence-based exposure concentration (C_R); it represents conventional exposure concentration estimation using only residence address information. Since the spatial location of

exposure does not change with this approach, the discretized exposure concentration during each personday varies in time only, not in space. Using the ArcGIS intersect tool, we extracted concentrations from the 24 dispersion modeling concentration maps (each representing one hour of the day with 1 km spatial resolution), resulting in ambient concentrations (c_{τ}) for each hour of the day (Δt_{τ} , equal to 1 hour) at each residence location. We then numerically integrated these data using time weighting in SPSS to estimate the daily residence-based exposure $C_R = (\sum c_{\tau} \Delta t_{\tau}) / T$ for each person-day in the study sample.

Second, we estimated daily exposure concentrations by matching the spatiotemporal locations in each person-day activity-travel record with modeled concentration at those locations; we call this the activity-based exposure concentration estimate (C_A). Specifically, we extracted concentrations from the modeled data for each discrete location along each person-day activity-travel path. This results in ambient pollutant concentration (c_{σ}) and time spent (Δt_{σ}) for each discretized spatiotemporal activitytravel path location, $\sigma =$ (latitude, longitude, time). Note that concentration for the same hour of day changes due to movement in space. The daily activity-based exposure concentration was then numerically estimated as $C_A = (\sum c_{\sigma} \Delta t_{\sigma}) / T$ for each person-day in the study sample. For explanatory analyses, we also estimated exposures, $E_A = \sum c_{\sigma} \Delta t_{\sigma}$ for sub-daily periods.

To compare the two measures of exposure concentration, we calculated the relative percent difference between the activity-based and residence-based exposure concentration as $(C_A - C_R)/C_A$ for each person-day in the sample. We call this the exposure error, as it estimates the error associated with using residence location only to calculate exposure. Frequency distributions for the study sample of daily exposure concentration (estimated using both methods) and of exposure error were compared to describe differences. A paired-sample t-test was used to quantify the significance of differences in the means for each sample distribution. Finally, we calculated bias factors to quantify the potential bias in relative risk estimates (based on simple linear models) due to use of the residence-based exposure estimate, following the method outlined by Setton et al. (2011).

3.3.5 Analysis of Exposure Distributions and Inequality

A second goal was to characterize disparities between groups in activity-based exposure concentration and in potential exposure error, including identification of factors impacting both. To do this, we first categorized daily exposure concentrations and exposure errors by population subgroup. We focused on subgroup types representing characteristics that have previously been found to experience exposure disparities or air pollution susceptibility. Specifically, the person-day exposure concentration estimates were categorized by age (5-18, 19-45, 46-65, and older than 65 years), race/ethnicity (Asian, white, Hispanic, black), and household income/poverty (below poverty, above poverty to below \$75,000, above \$75,000). Age less than 5 could not be considered, as no survey data are available for this category. To define the poverty threshold, we used the 2009 federal poverty guidelines that are based on household size (Department of Health and Human Services, 2009). The above \$75,000 threshold was chosen to capture approximately the highest third of the income distribution in the study area. After categorization, group frequency distribution summary statistics (e.g., mean, percentiles) were calculated and compared. To measure the significance of differences between groups of the same type, we used 95% confidence intervals around each mean and performed one-way ANOVA, followed by post-hoc Games-Howell testing (Hayes, 2005). A similar analysis was performed for differences in exposure errors between groups.

To investigate impacts of urban design, travel, and activity factors on exposure and exposure disparities, we performed a few additional analyses. As a proxy for urban design, we first categorized exposure concentrations by the urbanicity of the residence location (urban, suburban, second city, rural), as provided with the survey data (Claritas, Inc. [2004] provides urbanicity category definitions; we use the term rural for the town and country category, to avoid confusion with the proper name – Town 'N' Country – of a region in the study area. See Figure 3.1). Next, we categorized exposure concentrations by daily personal travel time. A preliminary analysis of the time spent at different activity location types

shows that mean daily personal travel time is 62 minutes (see Table 3.1); thus, we formulated three daily travel time categories, i.e., no travel, daily travel less than 1 hour, and daily travel greater than 1 hour to understand the distribution of exposures by daily personal travel time. Distributions of activity-based exposure (concentration x time) were also compared between different activity location types to explore the contribution of activity-location type to exposures. Specifically, activity location types were divided into three categories—at-residence, non-residential, and in-travel. Exposures were also compared against daily travel time. Similar analyses were performed for exposure error. Finally, we performed a multivariate linear regression analysis to assess the impact of urban design and activity factors on exposure concentration. Specifically, we used a hierarchical stepwise approach, in which the sociodemographic predictors (gender, age, racioethnicity), followed by the income categories, were introduced first, to control for their impacts on exposure concentration. The urbanicity categories, followed by the activity time variable, were entered subsequently. All predictor variables were introduced into the model as categorical binary variables (e.g., male/female, black/non-black), except the time variable, which was introduced as a continuous variable. A 95% confidence (p < 0.05) statistical significance criteria for each predictor variable was used to discard or retain variables at each modeling step. All statistical analyses were performed in SPSS.

3.4 Results and Discussion

3.4.1 Distributions of Human Activity in the Tampa Area

Table 3.1 provides a summary of the average temporal distributions of activity types observed by the 2009 national household travel survey for the filtered study sample in Hillsborough County. Time activity data are also provided from two well-known historical human activity surveys used for exposure analysis, the National Human Activity Pattern Survey (NHAPS) (Klepeis et al., 2001) and the Canadian Human Activity Pattern Survey (CHAPS) (Leech et al., 2002). As in the historical surveys, residents in the study sample spent the majority of their time at home (about 80%), although the percentage of time spent at home is about 13% higher here. The order (from highest to lowest percentage of time spent) of activity location types is also the same here as in the NHAPS and CHAPS. However, the quantitative distribution of time is somewhat different; the population sampled here spent more time on average at work, and less time travelling, at meals, and at other activities. Some of these differences may be due to differences in spatial scale, geography, and demographics. Specifically, the NHTS results are for Hillsborough County, FL, while the NHAPS and CHAPS results are for the entire US and Canada, respectively. Further, Florida is a state with large older-adult population, which may contribute to more time spent at home. Finally, different definitions of the activity type categories between the surveys could also have led to some differences.

	This V	Work ¹	NHAPS ²	CHAPS ³			
Activity Type ⁴	min.	(%)	(%)	(%)			
Home	1151	80	67	67			
Other	116	8.0	19	20			
Work	98	6.8	5.9	6.0			
Travel	62	4.3	5.7	5.3			
Meals	13	0.9	1.9	1.8			

Table 3.1 Average time spent per day by activity location type

¹Filtered sample from 2009 NHTS for Hillsborough County.

²National Human Activity Pattern Survey (Klepeis et al, 2001).

³Canadian Human Activity Pattern Survey (Leech et al, 2002).

⁴The following specific categories from each study were included under each label. Home – NHTS Home category; NHAPS and CHAPS categories of Indoor at Home and Outdoor at Home. Work – NHTS Work category; NHAPS and CHAPS Office/Factory category. Travel – NHTS categories of Travel and Transport Someone; NHAPS and CHAPS categories of In Vehicles and Near Vehicles - Outside. Meals – NHTS Meals category; NHAPS and CHAPS Bar/Restaurant category. Other – NHTS categories of School/Daycare/Religious Activity, and Medical/Dental Services, Shopping/Errands, Family Personal/Business Obligations, Social/Recreational Activities, and Other categories; NHAPS and CHAPS categories; NHAPS and CHAPS categories of School/Public Building, Indoors-Other, Outdoors-Other, and Mall/Store.

Figure 3.2 provides the spatial distribution of activity time from the study sample (the percentage

of total time spent in each block group area (subplots a–c), along with urbanicity (subplot d) of each block group). To our knowledge, this presentation of a spatially distributed activity time density map applied to exposure analysis is novel. Subplot d indicates that urbanicity generally decreases from central Tampa, surrounded by suburbs (including Citrus Park and Temple Terrace). A few pockets classified as second city areas (Sun City Center, Brandon, Plant City, New Tampa, Town 'N' Country) are farther from central Tampa and are surrounded by areas classified as the rural urbanicity category. The block group areas in the largest time density category (with at least 0.4% of the total time in the sample) are located in areas

that are categorized as rural (e.g., Fish Hawk), suburban, and some second city locations (e.g., in Brandon and Sun City Center) and largely correspond to the areas with the highest percentage of residential activity time (not shown). The block groups with the highest densities of non-residential time are largely special locations (Tampa International Airport, University of South Florida) or, for the second highest category (containing from 0.1% to 0.4% of total time in the sample), in rural, second city, or suburban locations. As a whole, the population of the Hillsborough County sample spent little time in urban block groups, though the time densities are larger for non-residential activities.



Figure 3.2 Spatial distribution of sample population activity-time (% of time spent) and urbanicity in study area. a) % of total time spent in all activity types within block group, b)% of total time spent in non-residential activities within block group, c) difference (%) between residential (r) and non-residential (nr) activity-times spent in each block group, d) urbanicity category of block group.

3.4.2 Diurnally Varying Spatial Distributions of NO_x Concentration

The average diurnal cycle of modeled NO_x concentration for the study area is shown in Figure 3.3. See Yu and Stuart (2013) for a detailed discussion of the spatial distribution of concentrations in the study area and results from evaluation of model performance. For our purposes here, note that for many hours of the day the concentrations are highest along the major roadways in the area with a broad peak apparent over central Tampa and near Tampa International Airport. A high is also often visible near a major port facility (Port Sutton) to the south of Downtown. Diurnally, concentrations exhibit morning (6:00–8:00 am) and evening (5:00–9:00 pm) peaks, consistent with increased NO_x emissions from traffic during commute hours. The evening peak is more spread out in time than the morning peak, consistent with both a larger meteorological mixing height in the evening and typical commute behaviors; specifically, the morning commute is known to be largely driven by work-related activities, while the evening commute may include maintenance, social, recreational, and other activities (Jou & Mahmassani, 1997; Kim et al., 2008). A detailed evaluation of modeled estimates against measured data is provided in Yu and Stuart (2013).



Figure 3.3 Estimated diurnal cycle of hourly average ambient NO_x concentrations ($\mu g/m^3$) in study area, from dispersion modeling results

3.4.3 Daily Time-weighted Activity-based Exposure Concentrations and their Social Distribution

The cumulative distribution of estimated daily (24-hour) activity-based NO_x exposure concentration is shown in Figure 3.4 (left side). The mean exposure concentration for the study sample is $17 \ \mu g/m^3$, with values for individual person-day records ranging from 7.0 to 43 $\mu g/m^3$. Using a typical fraction of NO₂ in NO_x estimated for the Tampa area of 0.8 (Poor, 2008), the values found here roughly correspond to daily NO₂ exposure concentrations of 7.4 ppb and 18 ppb for the sample mean and maximum, respectively. Although these values are on the low end of 24-h NO₂ exposure concentrations measured elsewhere (e.g., Delfino et al. (2008); Kim et al. (2006)), they are in the range of 24-hour average NO₂ exposure concentrations that have been found to be associated with a variety of respiratoryrelated health outcomes (US Environmental Protection Agency, 2008).



Figure 3.4 Cumulative distributions for daily exposure concentrations. Activity-based daily exposure concentration (left), residence-based daily exposure concentration (middle), and daily exposure error between the two as a percent difference, $(C_A - C_R)/C_A$ (right). Boxplot whiskers indicate 5th and 95th percentile values, (x) indicates mean value. Summary statistics provided below each box plot; 95% confidence intervals around each mean in parentheses.

The distribution statistics of the activity-based daily NO_x exposure concentrations for a few subgroups are provided in Table 3.2. Cumulative distributions are provided in Figure 3.5. Apparent differences in exposure concentrations among subgroups in the racioethnic and income categories are seen. Among the racioethnic groups, estimated mean daily exposure concentration is highest for the black group ($20 \mu g/m^3$), followed by the Hispanic group; mean exposures were lowest for whites (16 $\mu g/m^3$). Results for the Asian subgroup are not shown due to the small sample size (14 person-days). Although within-group variations increase with increasing group mean concentrations, the 95% confidence intervals around the means for the black and white categories are far apart, and one-way ANOVA with post-hoc Games Howell testing also indicated high significance for the difference (p = $6x10^{-8}$). Differences between the other categories were not significant, as the confidence intervals overlap. Among the income categories, the mean daily exposure concentration was highest for the group

characterized by household income below the poverty level ($18 \mu g/m^3$). This value is slightly lower than that estimated for the black group. Mean exposure concentration decreases with the income category, to $16 \mu g/m^3$ for the group characterized by higher incomes (household annual incomes above \$75,000). The confidence intervals and post-hoc testing indicate statistically significant differences between means for the below poverty versus highest income group, and between the two above poverty groups, but not between the below poverty versus middle income group. Differences in mean exposure among the age-based groups are not apparent.

		Exposure Concentration (µg/m ³)								Exposure Error (%)							
Group	Ν	Mean		Confide	ence	Interva	ıl	Min	Max	Mean	(Confide	nce l	nterva	al	Min	Max
Race/ethnicity ¹	1173																
Black	115	20	(19.0	-	21.5)	8.5	43	1	(-1.1	-	3.3)	-64	34
Hispanic	29	18	(15.9	-	19.2)	11	28	5	(1.3	-	9.6)	-8.9	45
White	1029	16	(16.0	-	16.6)	7.0	41	4	(3.3	-	4.4)	-32	58
Income ²	1131																
Below poverty	137	18	(17.1	-	19.2)	7.4	43	1	(-0.6	-	2.6)	-64	45
Middle income	577	17	(16.8	-	17.6)	7.0	41	3	(1.9	-	3.5)	-52	53
Higher income	417	16	(15.6	-	16.5)	7.4	32	6	(4.6	-	6.5)	-31	58
Age	1224																
5-18	148	17	(16.1	-	17.9)	8.5	29	3	(0.6	-	4.6)	-64	53
19–65	665	17	(16.6	-	17.4)	7.0	41	5	(4.2	-	6.0)	-48	58
Over 65	411	16	(15.9	-	16.9)	7.4	43	1	(1.0	-	1.8)	-17	23
Urbanicity	1224																
Urban	267	22	(21.2	-	22.4)	12	43	0	(-1.3	-	0.9)	-64	25
Suburban	387	17	(16.3	-	17.2)	10	35	4	(3.3	-	5.2)	-31	42
Second city	287	16	(15.5	-	16.2)	8.8	25	4	(3.1	-	4.9)	-31	38
Town & rural	283	13	(12.6	-	13.6)	7.0	27	6	(4.3	-	7.1)	-17	58
Daily travel time	1224																
More than 60 min	452	17	(17.0	-	17.8)	8.5	41	8	(6.5	-	8.8)	-32	58
Up to 60 min No travel	533 239	17 16	(16.2 15.3	-	17.1 16.7)	7.6 7.0	43 35	2 0	(1.0 0	-	2.4 0))	-64 0	42 0

Table 3.2 Group distribution statistics for daily activity-based exposure concentration and exposure error

¹Racioethnic labels used here are shortened forms of Race and Origin category labels used by US Census. Category descriptions are available at www.census.gov. Note that placement in a category is by self-selection, and individuals may be categorized in multiple or no categories.

²Below Poverty, Middle Income, and Higher Income labels refer to households with income below poverty threshold, above poverty threshold but less than \$75,000, and \$75,000 or above.



exposure concentration (µg/m³)

Figure 3.5 Cumulative distributions of daily activity-based exposure concentration for population subgroups. (a) personal attributes and (b) urban characteristics. Category definitions provided in text. Note that racioethnic subgroup populations are not exclusive; populations have overlapping individuals.

Differences in daily activity-based exposure concentrations observed here between the racioethnic groups are consistent with our previous studies that have estimated exposures in the Tampa area using only residence location (Stuart et al., 2009; Yu & Stuart, 2013) or school location (Stuart & Zeager, 2011). Specifically, we found greater exposures for the black, Hispanic, and low-income (below poverty) groups than the white and higher-income groups, respectively. Hence, regardless of the use of individual-level activity information in the exposure estimation, the qualitative direction of the disparities found for the Tampa area appears to be robust.

Furthermore, results are consistent with other findings from the study area and elsewhere. Specifically, in a study of the Tampa area using 1999 National-scale Air Toxics Assessment concentration data and the population distributions from the 2000 US census, Chakraborty (2009) found that the black, Hispanic, and below poverty groups are subject to disproportionate cancer risks and respiratory hazards, while no conclusive inequalities were found for individuals above age 65. Overall, results here contribute to the body of literature across localities in the US and elsewhere (Green et al., 2004; Houston et al., 2004; Linder et al., 2008; Marshall, 2008; Marshall et al., 2006; Mitchell & Dorling, 2003; O'Neill et al., 2003; Pearce et al., 2006), largely finding typically higher exposures to primary pollutants for socially and economically disadvantaged groups, with some exceptions (Buzzelli & Jerrett, 2007) and reverse finding for secondary pollutants (Marshall, 2008).

It is worthy of mention that the use of spatiotemporal activity information in this study did not change the relative ranking of mean disparities between racioethnic versus income groups. The mean difference between the black and white category was larger than the difference between the below poverty and highest income group; blacks had the highest estimated average exposure of all racioethnic or income groups. However, this result is complicated by the results of regression analysis (discussed below), for which income below poverty was associated with a larger independent increase in exposure concentration (1.7 μ g/m³), than being black (versus non-black, 1.2 μ g/m³). However, the comparative difference in group mean disparities found here is consistent with results from other study areas. Specifically, in a study in southern California, Marshall et al. (2006) found that exposure levels differed more among ethnic groups than between high- and low-income households, while Clark et al. (2014) found a similar result through a national level analysis. We note that there are many aspects of social disadvantage that are not captured by race, ethnicity, or income alone. Further, it is well established that there are interactions between factors that affect exposure disparities (e.g., Apelberg et al. (2005); Perlin et al. (2001)) with many studies in the air pollution field now using multi-factor indices that can also include education, occupation, employment status, family size, and home ownership (Forastiere et al., 2007).

Although differences in daily exposure concentrations are evident in our results, their importance to health outcomes is not necessarily clear. To explore the potential importance, we applied literature estimates of increased risk (primarily as reported by US Environmental Protection Agency 2008) to estimate possible health impacts. Neuberger et al. (2007) found a 2.9% increase in risk of total mortality associated with a 10 μ g/m³ increase in 24-hour mean NO₂ concentrations. Applying this to the differences in group means found here would suggest an increased risk of 1% for blacks versus whites (on

average), and an increased risk of 0.5% for those living in poverty versus in households with annual incomes over \$75, 000 (on average). Even higher differences in health risks may be present between groups, when considering susceptible people, such as children and older adults. Application of the 61.3% increased risk for cough incidence per 20 ppb increase in 24-hour NO₂ concentration found by Schwartz et al. (1994) in a study of children, would result in an approximately 5% excess risk for black compared to white children here, on average. Similarly, applying the ratio of 6.8% increased risk of all respiratory hospitalizations per 20 ppb increase in daily NO₂ concentrations found in a study by Fung et al. (2006) of adults aged 65 and older, suggests a 0.6% higher risk for elderly blacks compared to elderly whites here, on average. Note that for any individual, the comparative risks may be higher or lower due to individual risk factors (smoking, diet, exercise, occupation, access to health care, etc.) (Dockery et al., 1993; Pope III et al., 2002). Additionally, since differences in harmful health effects have been found even when differences in exposures are not clear (Deguen & Zmirou-Navier, 2010), small differences in exposures between groups may be important.

Overall, our results suggest that to attain the policy goal of reducing disparities in health outcomes (Healthy People 2020 & US Department of Health and Human Services, 2010), interventions that reduce existent disparities in exposure between socioeconomic groups may be helpful. Further, the methods used here provide an approach for estimating activity-based exposures specific to individual person-days, but for a large sample. This could be helpful for the study of factors affecting population health outcomes and for estimation of expected risks, without the intractably large costs of personal exposure concentration sampling for a large population.

3.4.4 Urban Form, Activity, and Exposure Relationships

We are interested in understanding how factors related to urban form may impact the magnitude of exposures and their social distribution in the Tampa area. Figure 3.5b provides the distributions of estimated daily activity-based exposure concentrations categorized by the urbanicity of residence location, with statistics provided in Table 3.2. Substantial differences in NO_x exposure concentrations are seen between residence urbanicity types. The highest mean daily exposure concentration $(22 \mu g/m^3)$ was
found for records with urban residence location, while that for records with rural residence location was 40% lower (13 μ g/m³). Mean exposures were intermediate for the suburban and second city categories. The confidence intervals and post-hoc testing indicate that all differences between the category means are significant. The largest difference in means between urban versus rural residence urbanicity categories (9 μ g/m³) is also more than twice as large as the largest difference among the social categories discussed above (4 μ g/m³ for the black versus white subgroup mean difference). Hence, residence urbanicity likely influences exposure and its social distribution among groups. This is broadly consistent with results of previous studies comparing exposures for populations in urban versus rural areas. For example, in the EXPOLIS-Helsinki study, individuals living in Downtown had 23% higher exposures than suburban residents (Rotko et al., 2001). Similarly, in a study of school children, Rijnders et al. (2001) found that both outdoor and personal NO₂ exposures for the highest versus lowest urbanicity (and traffic density), with a mean difference in personal exposures for the highest versus lowest urbanicity category of 14.6 μ g/m³.

Results of the multivariate linear regression analysis (Table 3.3) also indicate that urbanicity was the strongest predictor of exposure concentrations (with the highest coefficient value and t-statistic) among the factors studied. A model using only the urbanicity variables as predictors (not shown) captures about 35% of variance in the individual exposures, a substantial portion of the total model variability captured in the more complex model shown. Consideration of interaction terms between the sociodemographic and urbanicity variables provides further insight on the influence of residence urbanicity on the disparities in exposure found above between sociodemographic groups. Specifically, interaction terms urban*black, suburban*black, and suburban*below poverty all had significant t-statistics (significance values of 0.002, 0.005, and 0.004, respectively) and high coefficients (3.2, 3.1, and 2.5 μ g/m³, respectively) when added to the base model shown. Furthermore, with the interactions terms added, the black explanatory variable (which now represents blacks living in second city and rural regions) was no longer significant, and the below poverty variable had substantially reduced significance (0.04 with a reduced coefficient value of 0.9 μ g/m³). Additional comparisons of subgroup exposures (not shown) also indicate that, for those living in second city and rural regions, the difference in the mean

exposure for blacks versus others is not significant. That is, the exposure disparity (higher group mean daily exposure concentration) found here for the black group, on average, above, is due to both higher exposures for the urban and suburban black population, and higher residence urbanicity for the black population in the study area. Residence urbanicity classification also explains some of the disparity between economic groups, but the result is more complicated, as an urban below-poverty interaction term was not found to contribute significantly, but the suburban below-poverty term was.

Explanatory Variable	Coefficient ¹ (β_i, γ_i)	t statistic	Significance	
constant ² (β_0)	11.5	35	3e-180	
Black	1.2	2.9	3e-3	
Below poverty	1.7	4.3	2e-5	
Middle income ³	1.2	4.7	2e-6	
Urban	8.3	23	1e-98	
Suburban	3.4	11	8e-25	
Second city	2.6	7.5	1e-13	
Time away from residence	0.2	6.8	1e-11	
Goodness of fit				
R^2	0.40			
Adjusted R ²	0.39			
Number of cases	1120			

Table 3.3 Linear regression model for activity-based exposure concentration

¹Regression model is $C_A (\mu g/m^3) = \beta_0 + \beta_1 X_{c1} + \beta_2 X_{c2} + ... \gamma_1 X_{t1} + \gamma_2 X_{t2} + ... + \varepsilon$, where $X_{ci} \rightarrow (1,0)$ are binary variables, and X_{ti} are continuous variables. Only time variable entered as continuous variable (with units of hours). β_i have units of $\mu g/m^3$, γ_i have units of $(\mu g/m^3)/hr$.

²Constant concentration represents exposures for people who are non-black with incomes over \$75,000, who live in town/rural areas and did not travel on sample day.

³Middle income refers to households with income above poverty threshold but less than \$75,000.

Although it is known that residence urbanicity is associated with increased exposure, the reasons for this are not well understood. One contributing reason that has been explored extensively is the presence of higher concentrations of pollutants in urban versus rural areas. We can clearly see in comparing Figure 3.3 and Figure 3.2d that NO_x concentration is generally higher in urban versus rural areas throughout the day. However, we look here at the additional role of activity, with a focus on travel activity. Figure 3.6 provides NO_x concentration and exposure distributions categorized by activity-location types (at-residence, non-residential, and in-travel) for the subsample (n = 975) of person-day

records that included some activity in each category on the survey day. The mean (time-weighted) concentration is highest ($19 \ \mu g/m^3$) for the in-travel activity category and lowest ($16 \ \mu g/m^3$) for the atresidence category. That is, ambient concentrations were higher at the locations of non-residential and travel activities (at least during the times when our sample population was located there). However, mean exposures ($\mu g \ hr/m^3$) are lower for travel and non-residential activities, as less time is spent in these activities (see Table 3.1). Overall, the group mean daily exposure concentration increases for those who travel more (Figure 3.5b). Confidence intervals for the categorical means (Table 3.2) indicate significantly different group mean exposure concentrations for daily activity records with more than 60 minutes of travel time versus those with no travel. Our multivariate linear regression (Table 3.3) also indicates a small increase in exposure concentration with increased daily time away from the residence location (travel time plus time at non-residential locations), with concentrations increasing by $0.2 \ \mu g/m^3$ per hour of total daily time. Hence, although residence location remains a better predictor of daily exposure concentration than does time away from the residence (or time travelling), these activity times may play a role.



Figure 3.6 Cumulative distributions of time-weighted NO_x concentration ($\mu g/m^3$) and NO_x exposure (μg -hr/m³) by activity type location for all sampled daily records, including some activity away from residence. NO_x concentrations and exposures shown in left and right figures, respectively. Summary statistics provided below each box plot; 95% confidence intervals around each mean are in parentheses.

These results are consistent with those of other recent studies indicating the importance of exposures during travel. de Nazelle et al. (2013) found that travel activities contributed 24% of the total daily intake of NO₂. Dons et al. (2011b) and Dons et al. (2012) found that transport time accounted for 21% of black carbon exposures, and identified transport activity as a primary reason for differences in exposure between family members. Zhang and Batterman (2013) also recently found that increased traffic congestion led to greater population health risks; for the on-road population, this was due in part to increased transport times. We found the contribution of time in travel to be less for our study area, accounting for 6% of the total daily exposure on average, but time at nonresidential locations accounted for 24%.

From an exposure mitigation perspective, it is known that higher concentrations of many pollutants in urban areas are due largely to the proximity and spatial concentration of air pollution sources in urban areas, including car exhaust on congested roadways, combustion emissions from home heating, and nearby industrial emissions. Hence, mitigation policies have focused on reducing emissions from sources (e.g., through engineering control technologies). However, reduction in exposures requires reduction in emissions at a rate outpacing economic and population growth, which has proved difficult to sustain. Another popular strategy has been urban design that displaces sources away from where people live via urban planning and zoning policies (South Coast Air Quality Management District (AQMD), 2005). However, this strategy has resulted in collocation of sources with socially-disadvantaged population groups who cannot afford to live in less polluted areas (Perlin et al., 2001; Pulido, 2000) and with increases in emissions-producing travel necessary for people to access their homes, places of employment, and services.

Hence, "smart growth" urban design strategies are now being promoted as potentially mitigating exposures (Office of Sustainable Communities et al., 2013). Previous work has suggested that highdensity urban growth can potentially help in reducing the vehicle miles travelled and the overall emissions (Hankey & Marshall, 2010; Stone et al., 2009). However, simply applying land use intensification (or densification) strategies without making modifications to the existing transportation

infrastructure might increase congestion and lead to higher concentrations in urban areas (Farber et al., 2009). This could also exacerbate social disparities in exposures, as many disadvantaged groups disproportionately live in more dense urban areas (Baum et al., 1999). This underscores the need for caution in implementing high-density developments alone. However, another informative viewpoint may be differences in activity behavior that place people in spatiotemporal locations of high concentrations. Particularly interesting from a policy viewpoint are activity behaviors that are impacted by civic infrastructure. In this study, we found that average concentrations were higher in travel and non-residential activities, and estimated daily exposures were higher for those who travel more. Hence, a focus on civic infrastructure that reduces time travelling (and other non-residential activities) as well as emissions at those locations may be warranted. Implementation of transit infrastructure is one such approach, as it can reduce congestion (with concomitant reductions in emissions) and can reduce the time spent travelling on congested roadways. However, cost-competitive transit infrastructure requires high-density development (Kenworthy & Laube, 1999).

3.4.5 Exposure Error

The cumulative distributions of estimated residence-based daily NO_x exposure concentration and exposure error (due to the use of residence versus activity-based approach) are shown in Figure 3.4. Overall, we found the mean exposure error [$(C_A - C_R)/C_A$] to be 3.6%, with a range of -64% to 58%. Additionally, for the majority of the sample (56%), the error is positive (the activity-based exposure estimate is larger than the residence-based estimate). There is a small amount of overlap in the confidence intervals around each mean, though a paired samples t-test suggests statistically significant differences (p = 3e-22). Additionally, for the subsample of person-day records (n = 985) that included at least some travel away from the residence on the survey day, the mean error is slightly increased (4.4%). The calculated bias factors for the full sample and for travelling subsample, were 0.85 and 0.82, respectively, indicating that in a health impact study using residence-based daily exposure estimates, the relative risk may be underestimated by 15%, or 18% for the traveling sample.

Mean exposure bias (or error) values observed in our study are consistent with previously reported results, and suggest the importance of consideration of activity and travel patterns for exposure estimation. For Metro Vancouver, Setton et al. (2011) reported an exposure bias for residence-based versus activity-based exposure estimates of 0.70 to 0.84 for NO₂ (depending on the method used for concentration interpolation). Further, in a study of Flanders and Brussels, Dhondt et al. (2012) found small but significant differences between the mean dynamic (i.e., activity-based) exposures and residential exposures (21.6 versus 20.98 μ g/m³), with a resulting exposure error of 2.9%. Similarly, in a Belgian study, Dons et al. (2011b) found that time-activity patterns could account for approximately 30% of weekly personal mean exposure differences between a worker and a homemaker from the same household. While their study does not consider exposure error explicitly, their findings underscore the importance of time-activity patterns and their impact on exposures.

Our results suggest that a residence-based approach likely underestimates exposures for a large proportion of the population, resulting in underestimated risks of health impacts of air pollution. However, for almost half (46%) of the population, exposures and risks may be overestimated using a residence-based approach. Additionally, although the average error was found to be 3.6%, the maximum (absolute) error was 64%. Hence, exposure estimation methods that account for spatiotemporal changes in location and concentration may be needed for more accurate estimation of exposure and better health impact assessments. Nonetheless, this does not discount the importance of exposures at the residence location. Our results above on the large percentage of time spent at the residence location (on average) and on the predictive value of residence urbanicity are consistent with epidemiological studies that continue to suggest the value of exposures at residence location as a predictor for health responses (Brauer et al., 2008; Gan et al., 2011).

3.4.6 Social Distribution of Exposure Error

It is interesting to inquire whether estimated exposure error differs between demographic groups, i.e., whether residence-based estimates may be systematically biased for specific segments of the population; systematic biases could lead to systematic misclassification of exposures by group during

health impact analyses. To address this question, Figure 3.7 provides the cumulative distributions of exposure error for each of the sociodemographic groups studied above, with statistics provided in Table 3.2.



exposure error (%)

Figure 3.7 Cumulative distributions of exposure error for population subgroups. Exposure errors presented by (a) personal attributes, and (b) urban characteristics. Above-poverty refers to households with income above poverty threshold but with incomes less than \$75,000. Note that racioethnic subgroup populations are not exclusive; populations have overlapping individuals.

Among the racioethnic groups, exposure errors are largely positive (underestimation) for the Hispanic and white subpopulations, with the highest variation seen for the Hispanic group. Results are mixed for the black subgroup, with largely positive errors, but a substantial proportion in the negative (overestimation) range. Mean exposure errors were not found to be significantly different between any of the racioethnic groups considered. With regard to income, exposure errors are largely positive (underestimation) for the higher-income (annual income above \$75,000) and middle-income groups, with some negative (overestimation) errors in the below-poverty group. We found the mean exposure errors between the higher income group significantly different from those for both the below poverty group and the middle income group, but difference between the low and middle income groups was not significant. Mean exposure error is positive for all age groups, but is highest (most underestimation of exposures) for

active adults (ages 19–65), and lowest, with least variation, for older adults (over age 65). Mean errors were significantly different between these two age groups, but not between either of these groups and the child (ages 5–18) group.

Residential location appears to be a major determinant of the direction and the extent of exposure error. As can be observed from the box plots, the residence-based exposure concentrations are almost equal to the activity-based exposure concentrations for a large proportion (50%) of those living in urban areas. This suggests that pollutant concentrations at residential and activity locations may be similar for those living in urban areas. Moreover, with decreasing density, the variability in the exposure error increases. Specifically, there is a greater incidence (and magnitude) of under-estimation of exposures by residence-based estimates in rural regions (and over-estimation in urban regions). Further, the mean exposure error was found to be significantly different for the individuals residing in urban regions and the individuals residing in the suburban, second city and rural regions. These results suggest that using residence-based estimates may lead to underestimation of NO_x exposures (and resulting health effects) for those living in low-density regions, when compared to those in high-density urban areas.

Exposure error also increases, both in magnitude and variability, with an increase in the travel time. Further, an increase in the travel time leads to higher potential for under-estimation of exposures. As such, ignoring activity and travel patterns for individuals who travel for a significant portion of their daily time, could lead to the underestimation of health effect estimates.

In summary, the mean exposure errors are high for age groups 19–65, above-poverty groups, Hispanics, rural residents, and groups with travel time greater than 60 minutes. Specifically, the agebased differences in the exposure error may be a manifestation of the differences in the propensity to travel among the different age groups (children and older adults are likely to travel less). Within the context of income groups and rural residents, their travel patterns may be a contributing factor for the high exposure error (their daily activity patterns may lead them into more polluted areas compared to their residential locations). For the groups with longer travel times, spatial variation in concentrations could be a contributing factor for such large exposure error. These results suggest that residence-based estimates

may underestimate exposures for the advantaged population groups, rather than vulnerable groups (with the exception of Hispanics).

Further, our results suggest that the residence-based approach may not necessarily lead to severely flawed exposure estimates for most vulnerable subgroups of the population. This provides support for previous studies that did not consider activity and travel patterns in exposure analysis. In absence of data on activity and travel patterns, such residence-based approaches may not necessarily lead to significantly biased exposure estimates, at least for a majority of the most vulnerable population segments. However, there are individuals within the susceptible groups who are still prone to either under or over estimation of exposures using the residence-based approach. Additionally, the error may be important for people whose occupations require them to travel or be present for significant portions of time on roadways (e.g., sales personnel, highway workers etc.).

To our knowledge, there is little previous literature on the socioeconomic distributions of exposure error within the US. Limited evidence on this topic is available from Europe (Dhondt et al., 2012). Although it is difficult to compare the social distributions of exposure error between these studies (as groupwise exposure errors are not reported in their study), we are able to observe a few similarities. Specifically, they also report that exposure error in rural locations is significantly higher compared to urban locations. Dhondt et al. (2012) also reported that rural zones had dynamic NO₂ exposure values that could be 15% higher than the static values. Our results above provide differences in the variability of exposure error between urban and rural regions and the distribution of exposure error among population subgroups.

3.5 Limitations

Some limitations affect the robustness of these findings. First, the travel survey data used here may not be representative of the true spatiotemporal distribution of activities. Although the survey sample size is quite large, the county sample may not be large enough to capture the full spatial coverage necessary. Use of activity-based travel demand models for exposure analysis (Beckx et al., 2009b; Dhondt et al., 2013; Dons et al., 2014; Hatzopoulou & Miller, 2010) is one promising approach for

generating the larger sample sizes that are needed. Second, exposures during travel activity were estimated using concentrations along the shortest route, as path data were not available. Although this is a reasonable approach, computed routes may not coincide with the actual travel paths on the particular sampled person-day. Third, due to limitations in the temporal availability of the input data sets, the travel data are from a 2009 survey, but the concentrations are based on 2002 data. Hence, results are not expected to represent exposures for a particular year. Fourth, findings are limited by the use of estimated ambient pollutant concentrations for exposure analysis, rather than indoor, microenvironmental, or personal measurements. In the case of important indoor or personal sources, this could poorly represent exposures. Fifth, we have directly considered only one pollutant (NO_x) in the analysis here. It is well known that spatial and temporal concentration patterns and scales of variability differ by pollutant (Bhugwant & Brémaud, 2001). These differences could result in different distributions of exposure and exposure error. We expect the result here to be somewhat informative to understanding exposures to primary pollutants with substantial traffic emissions, but not to pollutants with substantial secondary formation or important emissions sources that are not collocated with traffic (such as ozone and formaldehyde). Sixth, defining urbanicity based on a single contextual population density measure could limit our findings. Whereas this definition incorporates a few key characteristics of urban form, there is a need to consider additional measures including transportation infrastructure characteristics in defining urbanicity. Seventh, this work has focused on investigating inequality in exposures (and potential health outcomes) between population groups characterized by race, ethnicity, income (and residence urbanicity). However, we note that there are many indices of social disadvantage and inequality that have been used in air pollution exposure and health impact studies; appropriate indicators likely depend on the social and political context. Further, there are many individual and group factors other than differences in exposures that can lead to differences in health outcomes (O'Neill et al., 2003); some of these are access to health care, overall health, smoking, diet, exercise, occupation, and genetics. Finally, it is well established that group averages do not necessarily represent the exposures of individuals in that group. Hence, the social disparity findings and implications can only address group level differences.

3.6 Conclusion

In this study, we estimated ambient NO_x exposures for residents of Hillsborough County, Florida, using activity and travel data (from the national household travel survey) matched to the spatiallyresolved diurnal cycle of NO_x concentrations. Travel routes were estimated based on the shortest-time path. We examined the social distribution of these daily activity-based exposures. Finally, we compared our activity-based estimates with those that result from using only residence location.

The findings of this work include the following:

- The Hillsborough County travel survey sample population spent little time in urban block groups. The time densities in urban block groups are larger for non-residential than residential activities.
- The diurnal cycle of NO_x concentration in the study area exhibited typical morning and evening peaks, consistent with increased NO_x emissions from traffic during commute hours. Spatially, concentrations were highest near roadways and in urban areas throughout the day.
- The mean daily activity-based exposure concentration for the study sample was found to be $17 \ \mu g/m^3$, with values for individual person-day records ranging from 7.0 to $43 \ \mu g/m^3$.
- The black, Hispanic, and low-income subgroups had higher mean estimated activitybased exposures than comparison groups. The mean disparity in exposure between the black and white groups is larger $(4 \ \mu g/m^3)$ than that between the below-poverty and highincome groups $(2 \ \mu g/m^3)$. However, regression results show that income below poverty is associated with a higher increase in exposure than black heritage alone, whereas Hispanic status was not found to be a significant predictor.
- The highest group mean exposure concentrations $(22 \ \mu g/m^3)$ were seen for those living in urban regions. Having an urban versus rural residence was also associated with the largest increase in exposure concentration in the regression $(8.3 \ \mu g/m^3)$. Furthermore, the

residence urbanicity interaction variables largely explained the largest disparities found between sociodemographic groups. Being black while living in urban or suburban areas and living below poverty in suburban locations were each associated with higher exposures.

- Time in travel and other non-residential activities was also associated with higher activity-based exposure concentrations, specifically $0.2 \ \mu g/m^3$ per hour spent away from home. This is due to the higher concentrations at these locations.
- The overall mean exposure error resulting from using residence-based versus activitybased estimation was 3.6% here, with residence-based estimate lower for most of the sample population.
- The mean group exposure errors were highest for person-days with more than an hour of travel, people with higher household income, people living in rural areas, adults aged 19–65, and Hispanics. This suggests that studies that use residence-based exposure estimation may not be severely misclassifying exposures for disadvantaged and susceptible groups including blacks, low-income households, and older adults, at least on average.

In summary, this work demonstrates an approach for using available travel survey data and concentration modeling results for spatiotemporally-resolved estimation of activity-based exposures. Novel contributions include the presentation and use of a spatially distributed activity time density map applied to exposure analysis, and the examination of the social distribution of errors in exposure. Our results suggest that activity-based exposure estimation may be important for assessing exposures of individuals, but a residence-based approach may not necessarily lead to substantially biased exposure estimates for the most vulnerable groups, on average. Within the context of previous work, the results here continue to reveal the presence of social disparities in exposure and, possibly, exposure-related health risks, in the study area, even after accounting for spatiotemporal population movement. Further, they confirm the importance of the urbanicity of residence location (and to a lesser degree, travel time) in

influencing exposures and their social distribution. This supports the need for urban design policies that ensure that densification is accompanied by civil infrastructure (e.g., public transit) that decreases emissions in urban areas as well as time spent traveling, particularly for disadvantaged groups.

CHAPTER 4: AN INTEGRATED MODELING FRAMEWORK TO ESTIMATE EMISSIONS, CONCENTRATIONS, AND EXPOSURES UNDER ALTERNATE TRANSPORTATION DESIGN SCENARIOS IN THE TAMPA AREA

4.1 Introduction

Human exposure to traffic-related pollution is of specific interest due to its linkages to adverse health impacts (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Past research efforts have focused on accurately estimating exposure to traffic-related pollution using a variety of personal monitoring techniques in which individuals wear exposure measurement devices (Dons et al., 2011b; Kousa et al., 2001); however, personal monitoring campaigns are often limited to small sample sizes due to high costs. Alternatively, studies also used fixed monitoring station measurement data (Sarnat et al., 2010) to estimate individual exposures; however, this approach cannot capture important spatial variations in pollutant concentrations, especially for traffic-related air pollutants (Monn, 2001), potentially leading to exposure misclassification. To address this, locally in Tampa, travel-survey data was used to estimate personal exposures to traffic-related pollution (Gurram et al., 2015). However, these sample-based studies are generally inadequate for exploring impacts of policy scenarios that seek to lower human exposure levels.

Previously, researchers pursued several policy initiatives including pollution control technologies (Kimura et al., 2001), congestion taxation (Johansson et al., 2009), and smart growth approaches such as compact growth urban design and vehicle electrification (Yu & Stuart, 2017) that seek to improve air quality. Within these options, the smart growth approaches are appealing as they seek to improve public health by providing individuals with active forms of transportation while improving air quality (Frank et al., 2006a). However, there is a considerable ambiguity about the utility of smart growth-based urban design policies for alleviating air pollution and reducing population exposure. Specifically, studies

argued that employing a compact growth or high density development would lead to higher primary pollutant exposures for individuals within the high dense zones (Frank & Engelke, 2005; Hixson et al., 2009). Considering this, it is important to gather further evidence by evaluating the impact of alternative urban design policies on air quality and population exposure.

To address these issues, a few recent studies used a variety of transport and land use modeling approaches to estimate exposures to traffic-related pollution and answer policy questions (Beckx et al., 2009a; Hatzopoulou & Miller, 2010; Hixson et al., 2009; Stone et al., 2009). Within these tools, the activity-based travel demand models are equipped to provide high-resolution human activity information both in time and space. This information may be combined with pollutant concentration data to obtain disaggregate exposure measures for a hypothetical population.

A European study led by Beckx et al. (2009c) and a Canadian study led by Hatzopoulou and Miller (2010) produced the seminal work in the area of exploiting activity-based travel demand models to estimate personal exposure to traffic-related pollution. Beckx et al. (2009c) used the activity-based model ALBATROSS in conjunction with the emission model MIMOSA and the dispersion model AURORA to estimate roadway link-specific emissions, ambient concentrations, and personal exposures. Hatzopoulou and Miller (2010) used a similar model setup initially and enhanced this approach later by including an agent-based dynamic traffic assignment model (MATSim) to improve the sensitivity of vehicular emissions to congestion (Hao et al., 2010; Hatzopoulou et al., 2011). Despite the important advances made by these two studies, they have a few limitations. Both studies simulated individual activities at low spatial resolution; Beckx et al. (2009c) used postal codes (with average size of 8.8 km²), and Hatzopoulou and Miller (2010) used traffic analysis zones (TAZ). Typically, the spatial size of a TAZ varies, but it holds under 3000 individuals. Additionally, concentrations were simulated for a short time period in both studies. Moreover, both studies used a receptor grid with low spatial resolution to estimate pollutant concentrations; Beckx et al. (2009c) used a grid size of 9 km², whereas Hatzopoulou and Miller (2010) estimated concentrations at the TAZ centroids. Finally, neither study explicitly modeled individual exposures during travel. The European group tried to address this in a later study by

incorporating domain average near-roadway concentrations (Dhondt et al., 2012). Within the context of the US, a similar framework based on the above two studies was used recently to estimate population exposure to traffic pollution (Vallamsundar et al., 2016). However, the limitations discussed in the earlier studies carry over to this study. For a detailed comparison of the activity-based exposure modeling frameworks, refer to Table 4.1.

The limitations from the earlier studies raise an important point about the spatial resolution that is needed to accurately predict exposure to traffic-related pollution. Since traffic-related pollutant (e.g., NO₂) levels exhibit substantial small-scale spatial variation, use of a high-resolution concentration receptor network in conjunction with high-resolution population activity data may be necessary. Otherwise, the small-scale spatial variations in pollutant concentrations could go undetected, leading to biased exposure estimates. Moreover, exposures during travel have been found to contribute significantly toward overall daily exposures; ignoring them could lead to exposure misclassification (de Nazelle et al., 2013; Dons et al., 2012; Dons et al., 2011b; Gurram et al., 2015).

	Spatial Scope	Pollutant(s)	Daily Activity schedules				Traffic Assignment	
Study			Name and Type	Spatial Resolution	Temporal Resolution	Population Sample Size	Model	Vehicle Population Sample Size
Beckx et al. (2009)	Utrecht, Netherlands (Urban city center with area of 7.8 km ²)	PM ₁₀ and PM _{2.5}	A Learning-Based, Transportation- Oriented Simulation System (ALBATROSS); rule-based computational process model Postcode area (3987 postcode areas with average size of 8.8 km ²) 1-hour		1-hour	30% expanded to 100% using extrapolation	All-or-nothing assignment (TRANSCAD)	100%
Beckx et al. (2009b)	Netherlands (Country with area of 42,000 km ²)	NO ₂	A Learning-Based, Transportation- Oriented Simulation System (ALBATROSS); rule-based computational process model	Postcode area (3987 postcode areas with average size of 8.8 km ²)	1-hour	30% expanded to 100% using extrapolation	All-or-nothing assignment (TRANSCAD)	100%
Hatzopoulou and Miller (2010)	Greater Toronto (Metropolitan area that covers 7,200 km ²)	NO _x	Toronto Area Scheduling model for Household Agents (TASHA); rule- based computational process model	Traffic analysis zone (463 TAZs)	5-minutes	5% expanded to 100% using extrapolation	Static traffic assignment (EMME/2)	5%
Hatzopoulou and Miller (2011)	Greater Toronto (Metropolitan area that covers 7,200 km ²)	NO _x	Toronto Area Scheduling model for Household Agents (TASHA); rule- based computational process model	Traffic analysis zone (463 TAZs)	5-minutes	5% expanded to 100% using extrapolation	Dynamic traffic assignment (MATSim)	5%
Dhondt et al. (2012)	Flanders and Brussels, Belgium (Regional cities covering 13,750 km ²)	NO ₂	Forecasting Evolutionary Activity- Travel of Households and their Environmental RepercussionS (FEATHERS); uses the activity- scheduler based on ALBATROSS	Zone-level (1145 zones with average size of 12 km ²)	1-hour	100%	Equilibrium traffic assignment (TRANSCAD)	100%
Vallamsundar et al. (2016)	Maricopa County, AZ (600 km ²)	PM _{2.5}	open-source Activity Mobility Simulator (OpenAMOS); composite of a rule-based computational process and utility-maximization based model	Traffic analysis zone (175 TAZs)	1-minute	5%	Dynamic traffic assignment (DTALite)	5%
This study	Hillsborough County, FL (3280 km ²)	NO _x	The Person Day Activity and Travel Simulator (DaySim); utility- maximization based model	Parcels (1.02 million)	30-minutes (adjusted to 1-minute)	100%	Dynamic traffic assignment (MATSim)	100%

Table 4.1 Comparison of activity-based exposure modeling frameworks

Table 4.1 (continued)

Study	Vehicle Emissions			Pollutant Concentrations			Exposure During Travel
	Model	Spatial Resolution	Temporal Resolution	Model	Spatial Resolution	Temporal Resolution	I G
Beckx et al. (2009)	MIMOSA	Link-level	1-hour	AURORA	9 km ²	1-hour	Domain-average concentrations measured at traffic-related monitoring stations
Beckx et al. (2009b)	MIMOSA	Link-level	1-hour	AURORA	9 km ²	1-hour	Domain-average concentrations measured at traffic-related monitoring stations
Hatzopoulou and Miller (2010)	Mobile 6.2	Link-level	1-hour	CALPUFF	Traffic analysis zone (463 TAZs)	1-hour	Not modeled
Hatzopoulou and Miller (2011)	Mobile 6.2	Link-level	1-hour	CALPUFF	Traffic analysis zone (463 TAZs)	1-hour	Not modeled
Dhondt et al. (2012)	MIMOSA4	Link-level	1-hour	IFDM	0.02 km^2 near roads and 1 km ² elsewhere	1-hour	Averaged the hourly concentrations near the roadway receptors for the entire study area
Vallamsundar et al. (2016)	MOVES	Link-level (only 3 major corridors totaling 55 km was used)	1-hour	AERMOD	0.01 to 0.09 km ² (receptor concentrations are averaged over TAZs)	1-hour	Not modeled
This study	MOVES	Link-level	1-hour	R-LINE	0.25 km^2	1-hour	Estimates intermediate locations along travel routes at every 5-second interval and uses the nearest receptor concentration

In this study, an integrated modeling framework built using an activity-based travel demand model (DaySim), dynamic traffic assignment model (MATSim), mobile source emission model (MOVES), and line-source pollution dispersion model (R-LINE) is presented. This modeling framework may be used to estimate human activity patterns, mobile source emissions, ambient traffic-related pollutant concentrations, and human exposure. Whereas the European and Canadian studies use a similar setup to explore human exposures to traffic pollution, this study is warranted for the following reasons. First, there is a need to gather further evidence on the utility of such frameworks that use the activitybased travel demand, dynamic traffic assignment, and dispersion modeling (ABM-DTA-dispersion) to estimate human exposure to traffic-related pollution. Second, the activity-based models that have been used for exposure estimation in the previous studies are rule-based computational process models that follow a set of if-then condition-action rules to estimate the activity and travel behavior of individuals. Exposure estimation frameworks that use utility maximization-based econometric models are currently non-existent. The utility maximization-based models originate from the economic theories of consumer choice and operate under the philosophy that individuals choose the alternatives that maximize their utility; utility refers to the level of satisfaction that an individual achieves by choosing an alternative. Thus, this study demonstrates the applicability of a utility maximization-based econometric model for exposure estimation. Third, this study simulates the activity and travel patterns of individuals at high spatial resolution. Specifically, activities of individuals are simulated at a parcel level instead of a TAZ level. Here, parcels refer to the detailed coordinates of the centroids of physical structures including office buildings and housing units. Fourth, to my knowledge, no other study that uses an ABM-DTAdispersion framework explicitly simulates personal exposures during travel. Fifth, unlike previous studies, this study uses the full hypothetical population rather than a sample to simulate detailed spatiotemporal activities of individuals and their exposures. Finally, this study can provide insights on the impact of spatial resolution of modeled activity and pollutant data on individual and population exposures. In essence, this study can add to the current body of work on the utility of transportation

modeling for estimating individual-level population exposure to traffic-related pollutants, using a tool that offers high spatiotemporal resolution.

4.2 Methods

This section provides an overview of the study area and demographics, followed by the integrated modeling framework along with the data sources used.

4.2.1 Study Area, Demographics, and Pollutant Focus

The study area is Hillsborough County, Florida, which is a part of the Tampa Bay region. The geographic context of the county is presented in Figure 4.1 (Google, 2017). Interstate highway 275 acts as a major commuter corridor connecting north of Tampa to the central business district at the south. I-75 and I-4 run along the north-south and east-west directions, respectively, and serve as major highways for intra-city, inter-city, and inter-state travel.

According to the US Census, county population in 2010 was approximately 1.2 million, with 51.2% female. The age categories of under 5 years, 5–19, 20–44, 45–64, and over 65 represented 6.7%, 20.6%, 36.3%, 24.8%, and 11.5% of population, respectively. Within the context of race, the white, black, and Asian categories were the largest, with 74.2%, 16.4%, and 3.4% of the population, respectively. Additionally, 23.9% of individuals identified with Hispanic or Latino origin. Finally, the county household income categories of below \$25,000, \$25,000 to \$75,000, and above \$75,000 correspond to 23.5%, 45.8%, and 30.8% of the population, respectively (US Census Bureau, 2010a).



Figure 4.1 Study area of Hillsborough County, Florida (Source: Google 2017)

In addition to a diverse mix of people, the county has few public transportation options, an unsatisfactory air quality record (American Lung Association, 2011), and a sprawling urban form (Smart Growth America, 2014). These attributes make it a good testbed for investigating alternate transportation design scenarios that may improve air quality in the region. The study focused on oxides of nitrogen (NO_x) as a surrogate for traffic-related air pollution (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010).

4.2.2 Transportation and Air Quality Modeling Framework

The integrated transportation and air quality modeling framework used to estimate personal

exposures for the entire population in the study region is shown in Figure 4.2.



Figure 4.2 Integrated transportation and air quality modeling framework for population exposure estimation

4.2.2.1 Estimation of Travel Demand using DaySim

The Person Day Activity and Travel Simulator (DaySim) shown in Figure 4.2 is an econometric travel-demand model system based on the principle of utility maximization. The theory of utility maximization suggests that individuals choose the activity and travel alternatives that maximize their utility; utility is a representation of the level of satisfaction that individuals achieve by pursuing an

alternative. DaySim predicts the travel demand for a region by estimating the long-term and short-term choices of individuals. Long-term choices include work and school locations for individuals and auto ownership levels for households, and short-term choices include the location and scheduling of daily activities and mode of travel between activities (Bradley et al., 2010).

The DaySim framework requires inputs including synthetic population and parcel information as well as level of service measures including travel time and travel cost for different times-of-day and travel modes to estimate the daily activity and travel behavior of individuals (Bowman & Ben-Akiva, 2001; Bradley et al., 2010). Synthetic population refers to the population records generated from the census data using an Iterative Proportional Fitting approach (Beckman et al., 1996). Specifically, individual records from the Public Use Microdata Area (PUMA) sample are replicated a sufficient number of times so that the resulting aggregate distributions at the level of the census block group match those provided by the census for a few control categories. Similarly, parcel information refers to a high-resolution dataset that provides information including the spatial location, land use, and property value for each land parcel in a region. Using these inputs, DaySim provides the activity and travel information including the location, sequence, and timing of activities and the mode of travel between activity locations, for each and every hypothetical individual in the study area.

For the purpose of this study, the Tampa Bay Activity-based Model (TBABM) was used; TBABM developed for the Tampa Bay area as part of the Federal Highway SHRP2 project, was based on the DaySim modeling framework (Gliebe et al., 2014). The Tampa Bay area consists of five counties— Hillsborough, Pinellas, Pasco, Hernando, and Citrus; thus, the geographic scope of TBABM extends beyond Hillsborough County, which is the primary geographic focus for estimating population exposure. Considering the activity and travel behavior of the adjacent county residents is important to account for the travel related to inter-county activities including work. The demographic inputs based on the 2010 census were developed by using POPGen 1.1 (Ye et al., 2009) for an estimated 3.1 million individuals in the Tampa Bay region, including 1.2 million in Hillsborough County. The parcel-based land use and highway and transit network inputs for the model were created by the staff of the District 7 office of the

Florida Department of Transportation (FDOT District 7). The model structure and parameters initially developed for the Sacramento region were retained in the Tampa Bay ABM, as the consultant team found that local data were insufficient to estimate many key models in the system (Gliebe et al., 2014). Results of a model run of the TBABM provide the daily activity and travel information from 3:00 AM to 2:59 AM on a typical weekday for the entire population in the Tampa Bay region; the activity and travel information include the purpose of each travel trip (work, shopping, recreation, etc.), trip start and end times, travel times and travel distances, and the detailed geo-coded locations of activities.

4.2.2.2 Estimation of Dynamic Traffic Patterns using MATSim

Whereas DaySim provides detailed spatiotemporal information pertaining to the fixed-activity locations of individuals, it is not capable of providing their whereabouts during travel. This knowledge is crucial to accurately estimate exposures because exposures during travel typically were found to be higher compared to other activities (Dons et al., 2011b; Gurram et al., 2015). Therefore, to estimate spatiotemporal locations of individuals for the entire 24 hours, the Multi-Agent Transport Simulation (MATSim) was used, as shown in Figure 4.2. MATSim is an iterative agent-based micro simulation of traffic systems and provides a framework for optimizing the travel demand of each modeled individual or simply an agent (Balmer et al., 2009; Raney & Nagel, 2006). To initiate the optimization process, the program requires the travel plans (i.e., travel-demand) for each agent and the characteristics of the transportation network. Specifically, a plan refers to the activity and travel information including detailed origin-destination location coordinates, activity types, activity start and end times, and travel modes between activities. The network characteristics describe the transportation infrastructure by providing information including geographic coordinates, length, free-flow speed, capacity, number of lanes, and allowed travel modes for each roadway link. Initial plans are the activity and travel patterns obtained from the TBABM. MATSim then simulates the agents' daily schedule (i.e., their initial plan) subject to the network capacity and travel time constraints. Each agent was allowed to store a maximum of three different plans in memory. In every iteration, MATSim selects a plan, executes it, and then calculates a score for the simulated plan based on its utility (Nagel et al., 2016). The utility of a plan S_{plan} with N

activities is obtained by a summation of all utilities from activity participation ($S_{act,q}$) and travel disutilities ($S_{trav,mode(q)}$) and is based on the scoring function developed by Charypar and Nagel (2005) as shown in equation 4.1. Generally, participating in an activity is associated with positive utility while activities like travel are associated with negative utility or disutility.

$$S_{plan} = \sum_{q=0}^{N-1} S_{act,q} + \sum_{q=0}^{N-1} S_{trav,mode(q)}$$
 4.1

In this study, the default parameters proposed by Charypar and Nagel (2005) were used for scoring; activity participation, late arrival, and traveling are scored at 6, -18, and -6 utilities/hour, respectively. Thus, plans that maximize activity participation times while minimizing the travel times and avoiding late arrivals have higher scores.

Following the scoring at the end of every iteration, the scores of the executed plan are compared with that of the plans in memory, and the plan with the least score is dropped. Following this, for a fixed percentage of agents, the program replans (i.e., modifies) the activity schedules by either rerouting or adjusting the departing times by 15 minutes. Specifically, in every iteration, 10% of the agents were rerouted, departure times were adjusted for 10% agents, and the rest of the agents stuck to their previous best plans. These percentages were based on computational feasibility and guidance from previous literature (Waraich et al., 2015). Subsequently, the program proceeds to the next iteration. To completely exploit the high computing resources available to us, the simulation was allowed to continue until 300 iterations although only a minimum of 60 iterations is generally needed according to the authors of MATSim (Waraich et al., 2015).

Only those trips that use the automobile mode of travel were simulated in this study as they have the largest mode share (close to 90%) in the study region based on the DaySim modeling results; the total number of automobile trips is approximately 9.7 million. To successfully simulate this large number of trips made by 2.3 million individuals (the rest of the population did not travel), a cluster setup of 48 processors each with 25 GB of RAM was used. The outputs from MATSim include updated activity and travel patterns and travel route information for each and every agent along with roadway link-specific traffic volumes and links-specific average speeds for each hour of a typical weekday.

4.2.2.3 Estimation of Mobile-Source Emissions using MOVES

The roadway-link traffic volumes and speeds simulated using MATSim can be used to generate link-specific mobile source emissions for each hour of the day. To estimate the mobile source emissions of pollutants, the United States Environmental Protection Agency's Motor Vehicle Emission Simulator 2014a (EPA MOVES) model was used (Koupal et al., 2003; US Environmental Protection Agency & Office of Transportation and Air Quality, 2015), as shown under the air pollution modeling component in Figure 4.2. MOVES estimates emission factors (e.g., mass per km per vehicle) or emission totals for both on-road motor vehicles and non-road equipment (Koupal et al., 2003).

Although Hillsborough County, FL, is the area of focus for air quality and exposure estimates in this study, the roadway links that fell within a five-kilometer buffer around Hillsborough County were also included for air quality analysis as impacts of emissions near the county edges on air quality cannot be ignored. The diurnal cycle of hourly emissions was estimated for an average winter day by running MOVES in a batch mode at the project scale for 14,025 roadway links. Specifically, MATSim-generated hourly car volumes and average speeds for each roadway link were input to the MOVES model. MOVES provides default data for the diurnal cycle of hourly temperature and relative humidity for each month of 2010. These observations were aggregated across November through March to generate an average diurnal cycle of hourly temperature and relative humidity for a representative winter day. County-specific default fuel formulation data and the national default vehicle age distribution data for 2010 were also used. The output from MOVES includes roadway link-specific running emissions of NO_x in grams for each hour of a typical winter weekday.

4.2.2.4 Estimation of Pollutant Concentrations using R-LINE

To estimate the pollutant concentrations from mobile sources, R-LINE was used, as shown in Figure 4.2. R-LINE is a line source model based on steady-state Gaussian formulation, which is consistent with current regulatory models including AERMOD and is used to simulate the mobile-source pollutant dispersion of near-surface releases (Snyder et al., 2013). The model includes new treatments for vertical and horizontal plume spread of near-surface releases and incorporates new tracer field and wind tunnel study data (Snyder et al., 2013). Additionally, it has been formulated to reduce estimation errors under light and variable wind conditions.

As presented by Snyder et al. (2013), R-LINE simulates concentrations from roadway sources by integrating the Gaussian plume formulation as a line segment. Given the coordinates (x_r, y_r, z_r) for the receptor r and a plume with origin at $(0, Y_s)$, the concentration at receptor r due to a line source of length L with origin Y1 is given by the summation of contribution from a differential element (dC_{pt}) as shown in equation 4.2 (Snyder et al., 2013).

$$C(x_r, y_r, z_r) = \int_{Y_1}^{Y_1+L} dC_{pt}$$
4.2

The element's contribution is further a function of plume (pl) and meander (m) components which are added using a weighting factor f as shown in equation 4.3. Here, f is a function of lateral turbulence and mean wind (Snyder et al., 2013).

$$dC_{pt} = (1 - f) * dC_{pl} + f * dC_m$$
4.3

The plume concentration and the meander component are composed of vertical (VERT) and horizontal (HORZ) dispersion terms, emission rate of q (mass/(time*length)), and effective wind speed of U_e (length/time) and are represented using equations 4.4 and 4.5, respectively (Snyder et al., 2013).

$$dC_{pl} = \frac{qdY_s}{U_s} \left[VERT * HORZ_{pl} \right]$$

$$4.4$$

$$dC_m = \frac{qdY_s}{U_a} [VERT * HORZ_m]$$

$$4.5$$

The vertical component for both plume and meander and the horizontal component for plume, and the horizontal component for meander are shown in equation 4.6, 4.7, and 4.8, respectively. Under low wind speeds, the horizontal plume is assumed to move equally in all directions, thus giving rise to the horizontal meander equation 4.8 (Snyder et al., 2013).

$$VERT = \frac{1}{\sqrt{2\pi}\sigma_z} \left[exp\left(-\frac{1}{2} \left(\frac{z_s - z_r}{\sigma_z} \right)^2 \right) + exp\left(-\frac{1}{2} \left(\frac{z_s + z_r}{\sigma_z} \right)^2 \right) \right]$$

$$4.6$$

$$HORZ_{pl} = \frac{1}{\sqrt{2\pi}\sigma_y} exp\left(-\frac{1}{2}\left(\frac{y_r - y_s}{\sigma_y}\right)^2\right)$$

$$4.7$$

$$HORZ_m = \frac{1}{2\pi\sqrt{(x_r - x_s)^2 + (y_r - y_s)^2}}$$
4.8

The diurnal cycle of winter time hourly link-specific emission outputs from MOVES were used as inputs to R-LINE to calculate the diurnal cycle of concentrations throughout the spatial domain, i.e., Hillsborough County, FL. Specifically, link-level emissions were modeled as line sources using the roadway length and width characteristics obtained from the roadway network file provided by FDOT District 7 as part of TBABM. Based on Grimmond and Oke (1999), the ratio of displacement height to roughness length was assumed to be 5. Additionally, the initial dispersion length for the plumes created from the line sources was assumed to be 1.2 m based on an average vehicle height of 1.5 m and in accordance with the US EPA's guidance for hot-spot analysis (US Environmental Protection Agency et al., 2010). Hourly meteorological surface data for November through March 2010 were prepared using the AERMET program by using raw data from the National Climatic Data Center for the Tampa International Airport. A total of 15% of the hours in 2010 had missing meteorological data fields, thus resulting in 3060 hours with valid meteorological data. Concentrations were generated for each hour of this record for a regular grid of receptor locations with 500 m resolution throughout the study area; the number of receptors totaled 13,806. Output values from the simulation were averaged to generate the diurnal cycle of concentrations for each hour of an average winter day.

4.2.2.5 Analysis of Exposure

The exposure modeling step involves combining the spatiotemporal locations of simulated hypothetical individuals with the spatiotemporal distribution of pollutant concentrations to estimate person-level exposures, as shown in Figure 4.2. The outputs from DaySim and MATSim were merged to create a sequential activity-record for each Hillsborough County resident for a 24-hour period. Specifically, the activity records contain the location coordinates, time-of-day, and activity durations both for fixed-location activities and the travel activity, for each individual. This information is combined with the diurnal cycle of winter-average concentrations to generate time-weighted exposure measures for all the representative individuals in the county. The daily activity-based exposure concentration was

numerically estimated as $C_A = (\sum c_{\sigma} \Delta t_{\sigma})/T$ for each person-day in the study sample, with c_{σ} and Δt_{σ} representing the ambient concentration and the time spent at each discretized spatiotemporal activity location; σ represents the latitude, longitude, time; and *T* the total exposure averaging period.

To understand the impact of using high-resolution versus low-resolution data on sub-group and population exposures, two different sets of data were used to model exposures. The high-resolution activity data refers to the activity and travel patterns of individuals at the parcel level for fixed-activity locations and every five seconds during their travel; the low-resolution activity data refers to activity and travel patterns estimated at the block group-level with even assignment of the travel times to the origin and destination block groups. Thus, in the low-resolution scenario, individuals are assumed to stay at the block group centroid instead of the parcel location. Similarly, the high-resolution concentration data were estimated for a 500-meter regularly-spaced receptor network, whereas the low-resolution data were estimated for receptors located at the block group centroids.

4.3 Results

4.3.1 Spatiotemporal Distributions of Activities

Analysis of the time spent by county residents in different locations reveal that they spent little time collectively in the county during active working hours, predominantly from 8:00 AM until 5:00 PM, as shown in Figure 4.3. Conversely, they spent a lot of time collectively in the county during the off-work hours, predominantly from 8:00 PM until 7:00 AM. More specifically, the collective time spent within the county remains highest and fairly constant from 11:00 PM until 6:00 AM, after which it falls steeply until 10:00 AM. It remains fairly constant and at its lowest level from 10:00 AM until 3:00 PM. Finally, the collective time spent in the county increases from 3:00 PM until the end of the day, with a steep rise from 4:00 PM until 8:00 PM.



Figure 4.3 Diurnal cycle of daily cumulative duration spent by all county residents

The spatiotemporal distributions of the activity and travel patterns of individuals at block groupresolution are shown in Figure 4.4. Spatially, for much of the day, high activity durations were observed in the residential areas of New Tampa, Westchase, Riverview, Sun City Center, Fish Hawk, Brandon, and Plant City; it should be noted that activity durations at these locations were more pronounced in the offwork hours. As the workday begins, activities spill over into the adjacent block groups and/or block groups that are (often) categorized as employment and business generators. Specifically, starting at 7:00 AM, higher activity durations were observed in the University area, the industrial area behind Tampa International Airport, at the intersection of I-4 and I-75 west of Brandon, MacDill Airforce Base, and the corridor along Dale Mabry Highway. The higher activity durations in most of these employment and business-generator block groups persist until 8:00 PM. Unlike for the high activity durations, the block groups with low activity duration are generally scattered throughout the county. The only exception to this is the set of block groups near Downtown Tampa that appear to have lower activity durations during the off-work hours from 7:00 PM until 7:00 AM.

Although the raw activity durations provide information on the collective locations of individuals, they may overemphasize results toward larger geographies. To account for the impact of block group size on activity duration, the activity durations were normalized by the block group area to create an activity-duration density map, as shown in Figure 4.5. Not surprisingly, high activity-duration densities coincided with smaller block groups concentrated in the urban core of Tampa. High activity-duration densities were

observed throughout the day in smaller block groups that are interspersed near the University area, Westchase, and Egypt Lake. Similarly, high activity duration densities were observed near Downtown from 7:00 AM until 8:00 PM. Conversely, low activity duration densities were predominantly restricted to the larger block groups that form the exoskeleton of Tampa.



Figure 4.4 Total activity duration (person-hours) by block group and hour of day for all individuals in Hillsborough County



Diurnal patterns of link-level passenger car volumes and travel speeds for Hillsborough County are presented in Figure 4.6 (in the form of bi-hourly averages). As expected, traffic volumes, as shown in Figure 4.6a), were higher during the morning (7:00–9:00 AM) and the evening (4:00–7:00 PM) peak hours than the rest of the day. Additionally, traffic volumes during evening peak hours were higher than volumes during morning peak hours. Travel speeds, as shown in Figure 4.6b, correspond to the diurnal pattern of traffic volumes, with lower speeds during the morning and evening peak hours. Spatially, higher volumes were observed along major freeway corridors—I-75, I-275, and I-4. This is expected, as these freeway corridors experience high traffic volumes, which also were observed along the road network near suburban locations including Brandon, Citrus Park, and Town 'N' Country. Accordingly, travel speeds were lower in these suburban locations along with the North Tampa area, the University area, and a few sections of the freeway corridors.

Traffic-count data for eight different locations in the Tampa Bay area were available in the transportation network input file of TBABM (Gliebe et al., 2014). Root Mean Squared Error (RMSE) between the estimated daily traffic volumes and observed annual average daily traffic (AADT) volumes, for these eight locations, was found to be 0.41. Further, the error between estimated and observed traffic flows for inter-city roads was higher than those for intra-city roads, presumably because the current model system does not consider long-distance (or inter-city) travel, visitor travel, and freight movement in detail.



4.3.2 Spatiotemporal Distributions of Emissions and Concentrations

The diurnal cycle of spatially-distributed NO_x emissions for a representative winter day (obtained by estimating a mean diurnal cycle of temperature and humidity for November through March) is shown in Figure 4.7. The total emissions resulting from passenger car travel in Hillsborough County was approximately 20.4 tonnes/day. The highest link-specific hourly emission of approximately 9800 grams was observed on the I-275 section in North Tampa from 7:00–8:00 AM. Temporally, emissions were higher during the morning (7:00–10:00 AM) and evening (3:00–7:00 PM) peak hours compared to the rest of the day. Emissions during the morning and evening peak hours make up more than 50% of daily total NO_x emissions. Additionally, emissions during the evening peak hours were higher compared to the emissions during the morning peak hours; specifically, the evening peak hour contributed 30.8% towards daily total emissions, whereas the morning peak contributed 21.8% to total emissions. This could be because the evening commute has a higher propensity for stopping to participate in other activities when compared to the morning commute (Chu, 2003).

Spatially, the highest emissions were observed along the major freeway corridors including I-75, I-275, and I-4; this is expected, as these corridors experience high traffic volumes—specifically, the roadway links on I-275 and I-75 in North Tampa, leading to the I-4 and I-75 intersection point on I-275 to the south of Brandon near Sun City Center, and on I-275 and Courtney Campbell Causeway, which connects Tampa with Clearwater. These roadway sections contributed roughly 7.5% towards daily NO_x emissions. Similarly, high emissions also were observed along the Veterans Expressway and the road network near Brandon and Town 'N' Country. Emissions were somewhat low in the Tampa Downtown area.

The diurnal cycle of NO_x concentrations resulting from passenger cars is presented in Figure 4.8. Overall, the predicted winter-average NO_x concentration for Tampa was approximately 4.7 μ g/m³. NO_x concentrations have two temporal peaks; one during the morning (5:00–9:00 AM) and the other during the evening (5:00–10:00 PM) peak hours. Note that emission estimates during the evening peak hours were


Figure 4.7 NO_x emissions by hour of day from cars for representative winter day in Hillsborough County



found to be higher than emissions during the morning peak hours, which is in contrast to the peak hour concentration estimates. Lower mixing heights could be a reason for the higher morning concentrations.

Similar to the emissions patterns, higher concentrations were observed along the major freeway corridors, including I-75, I-275, and I-4, as shown in Figure 4.9. This is primarily because automobiles were the only pollutant sources modeled here. High concentrations also were observed along the road network near the University area, the Downtown area, and Brandon, a suburban location near Tampa.



Figure 4.9 NO_x concentration distribution by hour of day due to on-road passenger cars in Hillsborough County for winter months 2010

4.3.3 Distribution of Exposures by Demographics, Urbanicity, and Travel Activity

To understand the potential disparities in exposure to traffic-related pollution, distributions of exposures for a few demographics were obtained; to understand the impact of residence location and transport on exposures, distributions of exposures for urbanicity and travel categories were obtained. The corresponding cumulative distributions of subgroup exposures along with the overall population exposure in Hillsborough County are shown in Figure 4.10. Figure 4.10a shows distributions of NO_x exposures by race, Hispanic origin status, income, and age, and Figure 4.10b shows NO_x exposures by residence location urbanicity and daily travel time in minutes.



Figure 4.10 Cumulative distributions of personal exposure concentration for NO_x resulting from passenger cars. Exposures are shown by a) demographics and b) urbanicity and travel activity. "Other" racial category includes American Indian or Alaskan Native, Native Hawaiian, multiracial, or other races. Income categories based on the household income. Middle income refers to individuals from households above poverty level but income below \$75,000.

The mean and median daily average population exposure concentration for NO_x resulting from passenger cars was 10.2 and 9 μ g/m³, respectively; exposures range from 0.2 to 145 μ g/m³. Compared to

the mean population exposure, subgroup mean exposure was lower for whites (lower by 1.4%), non-Hispanics (1%), individuals with household income above \$75, 000 (0.6%), children ages 0–5 (4.4%) and 6–8 (9%), and older adults above age 65 (11.2%). Conversely, mean exposure was higher for blacks (higher by 5.9%), Asians (2.1%), other racial subgroups (2.1%), Hispanics (3.7%), individuals from households living below poverty (2.7%) and at middle incomes (2.9%), and adult ages 19–45 (7.2%) and 46–65 (0.5%) compared to the mean population exposure. Similarly, for the urbanicity and travel categories, group mean exposure was lower for individuals living in rural areas (51%) who did not travel (9.7%) and whose daily travel time is less than 60 minutes (1.1%) compared to the mean population exposure; group mean exposure was higher for individuals living in urban areas (1.8%) whose daily travel time is greater than 60 minutes (8.3%) than the mean population exposure. Thus, on average, exposures were higher for the black, Asian, lower-income, middle-income, and active age (19–65) subgroups. Similarly, exposures were higher for individuals residing in urban zones and with higher daily travel times.

Although the distributions of group exposures describe the exposure disparities between groups of interest, they provide few details. To identify the strength of exposure-disproportionality at each exposure level, a subgroup inequality index shown in equation 4.9 was used (Stuart et al., 2009).

$$F_{ij} = \log(Z_{ij}/T_i) \tag{4.9}$$

 F_{ij} quantifies the degree to which members of a specific population subgroup *i* are disproportionately exposed to a particular level of pollutant, *j*, Z_{ij} is the fraction of the total population with exposures above level that is the specific subgroup, and T_i is the fraction of the total population of Hillsborough County that is the specific subgroup. Thus, positive and negative index values suggest disproportionately high and low representation of that subgroup at any exposure level. Exposure level can be quantified by a number of measures. Here, the percentile values of daily personal exposure concentration were used.

The subgroup inequality indices by race, Hispanic origin, income, and age are shown in Figure 4.11. Blacks, Asians, and other groups have disproportionately high representation at most exposure percentiles. Alarmingly, this trend accentuates as the exposure levels increases, i.e., minority groups were

more disproportionately affected at higher exposure levels. Conversely, a disproportionately low percentage of the white subgroup was affected at most exposure levels. As the exposure levels increased, the low representation of the white subgroup was further accentuated. Similar trends were observed for the Hispanic and non-Hispanic groups in which Hispanics make up for a disproportionately high percentage of exposed individuals at higher exposure levels. Among the income categories, middle-income group made up for a disproportionately high percentage of exposed individuals at higher exposure levels. Among the income categories, middle-income group made up for a disproportionately high percentage of exposed individuals at most of the high exposure levels when compared to the below-poverty and above \$75,000 income groups. However, when the below-poverty group was further separated into two groups—i.e., below-poverty white and below-poverty non-white—the below-poverty non-white group had disproportionately high representation at almost all the exposure levels as opposed to the disproportionately low representation of the below-poverty white group at most exposure levels. Further, the inequality index for the below-poverty non-white group increases steeply as the exposure levels rise. Finally, the active age groups—i.e., ages 18–65 years—appear to be disproportionately affected by high exposure levels compared to the relatively inactive age groups.



Figure 4.11 Subgroup inequality index by demographics versus percentile of cumulative distribution of individual daily exposure concentrations

4.3.4 Spatiotemporal Distributions of Exposure and Exposure Density

The spatiotemporal distributions of cumulative personal exposure concentration for NO_x aggregated by block groups are shown in Figure 4.12. The diurnal trend of spatially-aggregated exposures aligned very closely with the diurnal trend of estimated NO_x concentrations. The morning (6:00–8:00 AM) and evening (5:00–7:00 PM) commute hours accounted for about 48% of the daily population exposure.



Figure 4.12 Total cumulative exposure ($\mu g/m^3$ person-hr) to NO_x by block group and hour for sample of Hillsborough County residents

Spatially, high exposures were observed near the greater Carrollwood area, along the I-275 corridor, Riverview, and Sun City Center during off-work hours. However, during working hours, high exposures were concentrated in those block groups that form the urban core of Tampa; predominantly, Downtown and the corridors along I-275 and Dale Mabry Highway. The University area appears to have high exposures throughout the day. Low exposures were more prominent within the block groups that form the outer skeleton of Tampa, possibly due to the low NO_x concentrations in these areas; interestingly, unlike the activity durations, the Downtown area does not figure in the group of low exposure regions, thus pointing to the impact of ambient concentrations on exposures.

The spatiotemporal distributions of exposure densities (normalized by block group area) are shown in Figure 4.13. Spatially, during off-work hours, high exposure densities were concentrated in the residential pockets between the University area and I-275, along I-275, and near Downtown. During the morning commute from 5:00–8:00 AM, the high-exposure density areas shifted from the residential pockets and spread out across the urban core of Tampa, especially along major highways and suburban areas such as Brandon and Plant City. The spatial distribution of the exposure density remained fairly constant from 8:00 AM until 5:00 PM. During these work hours, the Downtown appeared to have the highest exposure densities. In addition, the University area, areas along I-275, the airport area, and the Carrollwood business area also featured high exposure density pockets; very few pockets in the Brandon and Plant City areas featured high exposure densities. During the evening commute of 5:00–7:00 PM, similar to the morning commute, high exposure densities were spread out across the urban core of Tampa, Brandon, and Plant City. However, it should be noted that the exposure densities during the evening commute.



Figure 4.13 Cumulative exposure density ($\mu g/m^3 * person hr/km^2$) for Hillsborough County resident sample 2010

4.3.5 Impact of Spatiotemporal Resolution of Modeled Data on Human Exposures

The differences in exposures obtained using highly spatiotemporally-resolved activity and concentration data and low-resolution data are shown in Figure 4.14. A positive value in the figure indicates higher estimates for the high-resolution approach compared to the low-resolution approach. A negative value indicates lower estimates of exposures using high-resolution activity and concentration data.



Figure 4.14 Relative percent difference in exposures obtained using highly spatiotemporally-resolved activity and concentration data and low-resolution activity and travel, and concentration data

Overall, the use of low-resolution activity and concentration data over high-resolution data led to 10% lower exposures, on average; the differences range from about -800% to 90%. This difference in exposure resulting from the use of low-resolution activity and concentration data versus high-resolution data can inform our understanding of exposure error resulting from using low-resolution data. On average, use of low-resolution data resulted in underestimation of exposures (positive differences) for all the demographic groups. The distribution of this exposure error among the various racial and ethnic

groups showed little to no variation from the distribution of population-level exposure error. Similarly, the below-poverty income category showed little difference from the population-level exposure errors. However, the exposure errors for the middle income and the above \$75,000 income groups were slightly larger compared to the population-level exposure error. The mean (and median) exposure errors for the middle and above \$75,000 income groups were 0.4% (1.3%) and 2.2% (3.2%) higher compared to the population-level exposure error; this suggests that, on average, these two groups have a greater propensity for underestimation of exposure swith the use of low-resolution data, compared to the population. Comparing age groups, the under 5, 6–18, and over 65 categories had slightly smaller exposure errors compared to the population; the mean (and median) exposure errors for the under 5, 6–18, and over 65 categories were 0.8% (2.1%), 2% (3.2%), and 3.3% (2.3%) lower compared to population exposure errors suggesting a lower propensity for underestimation of exposures on average. In contrast, the more active population groups of age 19–45 and 46–65 had slightly greater propensities for underestimation of exposure compared to the population; the mean (and median) exposure errors for these two groups were 1.6% (1.7%) and 0.4% (1.1%) greater than the population exposure error.

Similar to the demographic groups, the use of low-resolution data resulted in underestimation of exposures for the urbanicity groups on average. Moreover, the exposure error distributions for the urban and rural categories were very similar to the population-level exposure error distribution. The only caveat is the slightly higher (by 2.9%) mean exposure error for the rural category compared to the population suggesting a greater propensity for underestimation on average. In contrast to the demographic and urbanicity categories, the exposure error distributions for the travel categories showed a larger variation compared to the population. On average, use of low-resolution data for the groups traveling up to 30 minutes, 31–60 minutes, and above 60 minutes per day resulted in overestimation by 4% and underestimation by 11% and 21%, respectively. In addition, the mean (and median) exposure error for the group with low travel time was lower by 14.3% (8.8%) compared with population-level error. However, the mean (and median) exposure error for the group with the highest travel time was 11% (7.6%) greater than the population. For the group with travel time between 30 and 60 minutes, the mean

and median errors were respectively greater (0.2%) and lower (2.3%) compared to the population-level exposure error.

The use of low-resolution data also affected the ranking of mean exposures for a few demographic and travel categories. Specifically, use of low-resolution data resulted in a higher rank of mean exposure for medium travelers (31–60 min. per day), whites, other racial subgroups, non-Hispanics, and low-income individuals in their respective subgroups. In contrast, the ranking of mean exposure for heavy travelers (more than 60 minutes per day), medium and high income individuals, and the age group of 45–65 dropped with the use of low-resolution data.

4.4 Discussion

The spatiotemporal distributions of population activities in this study are consistent with observations from other study areas. Specifically, a shift in the activity locations from residential to non-residential (i.e., business, educational, airport, and work) locations between work hours (8:00 AM to 5:00 PM) and non-work hours (9:00 PM to 6:00 AM) was observed. Dhondt et al. (2012) and Vallamsundar et al. (2016) reported similar findings of higher person-hours in industrial or business zones during day and vice versa during nights. Moreover, morning and evening commute from 6:00–8:00 AM and 5:00–7:00 PM, respectively, show slightly different spatial activity patterns compared to the rest of the day predominantly due to individuals' commute.

The spatiotemporal trends of exhaust emissions resulting from passenger car travel in Tampa are comparable to emission estimates from other regions, but the aggregate emissions were slightly lower. Specifically, Hatzopoulou and Miller (2010) found that exhaust NO_x emissions were the highest along the major roads and the emission peaks correspond with the peaks in travel, as in this study. Additionally, they also observed higher emissions during the evening peak. However, this study estimated the daily total NO_x emissions for a typical weekday in 2010 to be 20.4 tonnes, whereas Hao et al. (2010) estimated the daily total NO_x emissions for the Greater Toronto Area in 2001 to be 75.62 tonnes, Beckx et al. (2009d) reported 70,210 tonnes for Netherlands in 2000, Batterman et al. (2014) reported 14,715 tonnes/yr (which roughly translates to a daily total of 40.3 tonnes) for Detroit in 2010, and Yu and Stuart

(2016) reported 44.4 tonnes on-road emissions for the same study region. The lower emissions totals here could be attributed to a variety of factors including the geographic scale, vehicle mix, and emissions processes considered. Specifically, both the Greater Toronto area and Netherlands feature much larger transportation networks compared to Hillsborough County. Further, the Netherlands and Detroit studies estimated emissions from heavy-duty diesel and gasoline vehicles, and the Greater Toronto Area and Netherlands studies estimated NO_x emissions from exhaust, hot, and cold start processes, whereas this study estimated on-road exhaust emissions for passenger cars only; moreover, this study is for 2010 whereas both the Greater Toronto Area and Netherlands studies were conducted for 2000 suggesting a temporal mismatch.

Due to the focus on car emissions only, the predicted NO_x concentrations in this study were also generally lower than the observed concentrations near a roadway monitor, and our research group's previous studies in this region using a more comprehensive emissions inventory (including point and area sources). Specifically, this study found some differences in the diurnal cycles of the estimated and observed hourly concentrations for winter, as shown in Figure 4.15. Generally, estimated NO_x values were lower than the observed values except for the peak hours. Following the peak, modeled concentrations drop rapidly, as opposed to a slow decline in the observed values. This is probably due to the non-inclusion of additional sources of pollution other than passenger cars. The spatially-averaged winter mean NO_x level in this study is $4.7 \ \mu g/m^3$, and Yu and Stuart (2013) reported a domain average annual mean NO_x concentration for 2002 of $12 \ \mu g/m^3$. Using a similar activity-based, emission, and dispersion modeling approach like this study, Beckx et al. (2009c) predicted NO₂ levels of $38 \ \mu g/m^3$ in 2005 for Netherlands.



Figure 4.15 Diurnal cycle of hourly observed and estimated concentrations at near-roadway monitor

Although the winter-average NO_x levels in this study appear to be low, the hourly NO_x concentrations at a few locations in Tampa may be high. For example, the NO_x concentration near the intersection of I-275 and I-4 on January 7, 2010 from 5:00–6:00 PM was 4405 μ g/m³. Using a NO₂–NO_x ratio of 0.4 for near-roadway locations (Batterman et al., 2014), this translates to 1762 μ g/m³ of NO₂. Moreover, the 98th percentile of one-hour daily maximum for 2010 winter measured at the same location was 1642 μ g/m³, which is about 9 times higher than the one-hour NO₂ standard (185 μ g/m³) set according to the National Ambient Air Quality Standards (NAAQS), defined as the 98th percentile of one-hour daily maximum concentrations averaged over three years. Although this study does not report a three-year average, uses a non-regulatory air quality model, and predicts NO_x instead of NO₂, the high one-hour NO₂ value estimated at this specific location makes a strong case for exploring human exposure to traffic-related pollution.

On average, exposures to NO_x were lower but high-end exposures in this study are greater than maximum exposures from earlier studies. Specifically, mean (and maximum) exposure concentration to NO_x in this study is $10.2 \ \mu g/m^3$ (and $145 \ \mu g/m^3$); a previous study for the same location that included additional sources of emissions reported activity-based NO_x exposure concentration of $17 \ \mu g/m^3$ (and 43

 μ g/m³) (Gurram et al., 2015). Similarly, activity-based exposure concentration of NO₂ for Flanders was estimated to be 21.6 μ g/m³ (and 44.3 μ g/m³) (Dhondt et al., 2012). Finally, Hatzopoulou and Miller (2010) estimated the maximum activity-based NO_x exposure concentration of 81 μ g/m³ for the city of Toronto. These findings show that although exposures were low on average in this study, the high-end exposures are either comparable to or greater than the maximum exposures reported in other locations. Additionally, the spatiotemporal exposure plot provides a novel way to interpret population exposures. Specifically, morning and evening commute largely influences the population exposure levels and the highest exposure densities were generally concentrated in the urban core of Tampa. In such a scenario, urban design policies that seek to reduce the travel times and distances, especially during the commute, and encourage non-auto modes of transport may potentially mitigate concentrations and population exposures.

Demographical and urbanicity-related exposure analysis in this study confirms the existence of exposure inequalities at the population-level. Specifically, findings of higher mean exposures for blacks, Hispanics, low-income groups, urban residents, and individuals with higher travel times, in this study, are consistent with previous investigations in the same study area and elsewhere (Gurram et al., 2015; Marshall, 2008; Yu & Stuart, 2013). Although consistent patterns of exposure variations with age were not observed in Chapter 3, this study found that mean and median exposures for active-age groups were higher. This could be of public health significance for working individuals who may be more susceptible (e.g., asthma patients) to the effects of traffic-related pollutants. Additionally, the finding of greater exposure inequalities (shown in Figure 4.11) for the below-poverty non-white group as opposed to the below-poverty white group is consistent with earlier studies (Clark et al., 2014; Gurram et al., 2015; Marshall, 2008). This shows that race may be a stronger predictor of individual exposure inequalities than income in some cases. But more importantly, this confirms that traffic-related exposure inequalities are persistent in Tampa and are propagated by the spatiotemporal distributions of the individuals and pollutants under the existing layout of transportation infrastructure. Thus, it is important to explore urban design policies that not only seek to mitigate exposures but also exposure inequalities.

The spatiotemporal resolution of the modeled activity and travel patterns, and pollutant concentrations has a substantial impact on the accuracy of estimated exposures. This suggests that the disaggregated exposure estimates in previous studies that used low-resolution activity and travel patterns and (or) pollutant concentrations (Beckx et al., 2009a; Beckx et al., 2009c; Dhondt et al., 2012; Hatzopoulou & Miller, 2010; Vallamsundar et al., 2016) and did not explicitly model exposures during travel (Hatzopoulou & Miller, 2010; Vallamsundar et al., 2016) could potentially be underestimated or overestimated in some cases. This also explains the larger disaggregate exposure levels, despite low mean NO_x concentrations, in this study compared to previous studies. Moreover, use of low-resolution data modifies the relative ranking of group-wise mean exposures. It is worthy of mention that earlier studies showed differences in overall (Gurram et al., 2015; Setton et al., 2011) and group-wise mean exposures (Gurram et al., 2015) when activity and travel patterns were included in exposure analysis. This study further demonstrates that the resolution of activity and travel, and pollutant concentration data is important, especially for disaggregated exposure analysis.

4.5 Conclusion

Activity-based travel demand modeling provides a unique opportunity to exploit the rich set of disaggregate spatiotemporal activity and travel data to inform on subgroup and population-level exposures to traffic-related pollution. This study used a framework based on activity-based travel demand modeling (DaySim), dynamic traffic assignment (MATSim), mobile-source emissions estimation (MOVES), and dispersion modeling (R-LINE) to estimate disaggregate and subgroup exposures to NO_x. Passenger-car-related NO_x concentrations at a few near-roadway locations could potentially exceed the one-hour NAAQS standard for NO₂. Additionally, persistent exposure inequalities were observed in the study area. Finally, the spatial resolution of activity and travel, and concentration data was found to influence exposure estimation and use of low-resolution data may lead to both underestimation and overestimation of exposures. Thus, this study adds to the body of literature on exposure modeling frameworks that use ABM-DTA-Dispersion paradigms.

CHAPTER 5: IMPACT OF TRANSIT-ORIENTED COMPACT GROWTH ON AIR QUALITY AND EXPOSURES TO TRAFFIC-RELATED AIR POLLUTION IN TAMPA AREA 5.1 Introduction

Exposure to traffic-related air pollution poses major health risks. A wide spectrum of studies associated exposure to traffic-related air pollution with autism (Volk et al., 2013), negative birth outcomes (Brauer et al., 2008), diminished cognitive development (Sunyer et al., 2015), lung cancer incidence (Beelen et al., 2008b), mortality (Beelen et al., 2008a; Hoek et al., 2002), and respiratory symptoms, atopic diseases, and allergic sensitization in children (Kim et al., 2004; Morgenstern et al., 2008). Understanding the pathways that lead to population exposure to traffic pollution may help in controlling the negative health outcomes.

Urban land use and design and transport planning are considered to be among the important factors that influence population exposure to traffic pollution. Frank et al. (2006b) used a walkability index that characterizes the urban form by quantifying the compactness, connectedness, and diversity of neighborhoods and found that increase in walkability leads to reductions in vehicular travel and emissions. Similarly, (Clark et al., 2011) found from an examination of 111 US urban areas that urban form characteristics such as population density and centrality along with transit supply may influence the urban air quality and corresponding population exposures. Although these studies reported associations between urban form, transport, and air quality, they were mainly observational and did not provide insights on the air quality and exposure effects of pursuing alternate urban forms for future development in a region.

To address this, a few studies modeled the impact of alternate urban forms and/or investment in transit infrastructure on vehicular emissions, concentrations, and population exposure. Stone et al. (2007) simulated vehicular activity in alternate hypothetical urban forms and found that compact urban forms

lead to less vehicular travel and emissions. Hixson et al. (2009) used a GIS-based land use planning tool, a four-step travel demand model, and a source-oriented three-dimensional photochemical air quality grid model to estimate air quality and population-weighted exposure in the San Joaquin Valley. They found that compact growth urban forms, when pursued along with investments in high speed rail and adoption of clean technologies, result in lower emissions of non-methane organic gases, NO_x, and PM_{2.5} when compared to sprawling or business-as-usual urban forms. Additionally, they showed that compact urban form helps in reducing the $PM_{2.5}$ concentrations over most of their study region (except for urban centers) but increases the population-weighted exposure by 10–15% when compared with low-density development.

Similarly, De Ridder et al. (2008a) combined spatial land use data obtained from satellite imagery with a four-step travel demand model and an atmospheric chemical transport dispersion model to study the impact of sprawling urban form on regional air quality and population exposure. They found that relocating 12% of the urban population to the greener peripheries results in a 17% increase in traffic volume, approximately 4% increase in ozone and PM_{10} levels, and 13% reduction and 1.2% increase in exposures for the group of individuals who moved out and who stayed, respectively.

More recently, Shekarrizfard et al. (2017) combined a travel demand model with EPA MOVES and the dispersion model CALPUFF to estimate the impact of transit and vehicle technology improvements on air quality and population exposure. Overall, they found that a large portion of reductions in vehicular emissions in the future transit investment scenario is due to improvements in vehicular technology, with transit investment accounting for an additional 3% reduction in the 2031 NO₂ levels; similarly, transit investment resulted in an additional 10% reduction in future-year population exposure to NO₂ (Shekarrizfard et al., 2017).

Locally in Tampa, Yu and Stuart (2017) found that compact urban form development along with vehicle fleet electrification could have varied (in both strength and direction) impacts on air quality and population exposure depending upon the type of primary pollutant being studied. Finally, Stevenson et al. (2016) modeled the health benefits of compact cities and found that such cities can achieve overall

health gains of 420–826 disability-adjusted life-years (DALYs) per 100,000 population. However, they also cautioned that their study quantified the linkages between land use, transport, and population health from a macro-level perspective and argued for the need to look at these linkages using agent-based modeling approaches.

In fact, most of the modeling studies mentioned above use transportation models that rely on aggregated demographic information to estimate travel demand; these models may not be sensitive enough to predict the shifts in the daily activity and travel patterns of individuals, including their travel mode, departure time, and activity-participation preferences. These activity and travel choices may have a significant impact on the distributions of on-road vehicles, emissions from those vehicles, concentrations, and population exposure. Thus, it is important to understand the linkages between urban land use and design, transport, and air quality through the use of highly resolved agent-based modeling approaches.

Previously, studies pioneered this approach by building frameworks that integrate activity-based travel demand models (ABM), dynamic traffic assignment models (DTA), mobile-source emission models, and dispersion models to estimate population-level exposures to traffic pollution (Beckx et al., 2009c; Dhondt et al., 2012; Hatzopoulou & Miller, 2010; Vallamsundar et al., 2016). The activity-based travel demand models, in particular, offer the capability to simulate the daily activity and travel patterns of individuals and their exposures to traffic-related pollution under different policy scenarios. Specifically, using the above ABM-DTA-emissions-dispersion framework, Dons et al. (2011a) studied the impact of altering shopping hours and Dhondt et al. (2013) explored the impact of fuel price increase on population exposures. Whereas these studies provide valuable insights into the effects of local policies on exposures, they did not fully exploit the land use and transportation-related features of this framework to understand the relationship between urban land use, transport design, and population exposure. This is a significant gap, especially considering that such transportation and air pollution frameworks are well-suited for simulating the impacts of alternate land use and transportation infrastructure scenarios on air quality and population exposures. In addition, as discussed in Chapter 4, the integrated transportation and

air pollution modeling framework in this study has more desirable features (over other such modeling frameworks) such as higher spatial and temporal resolution, inclusion of meteorological conditions for an entire season (as opposed to only a few days in a year), and explicit modeling of exposures during travel.

This study—part of an overarching study in Tampa—seeks to use an activity-based travel demand modeling approach to understand the impact of transit-oriented compact-growth strategies on local air quality and exposure levels; the multi-year ongoing project in Tampa is focused on understanding the linkages between urban form, transportation infrastructure design, exposures to traffic-related air pollution, and its social distribution (Evans & Stuart, 2011; Fridh & Stuart, 2014; Gurram et al., 2015; Stuart et al., 2009; Stuart & Zeager, 2011; Yu & Stuart, 2013, 2016, 2017). Specifically, this study uses the ABM-DTA-emissions-dispersion framework to understand the impact of implementing a future-year transit vision in conjunction with population reassignment strategies that reduce the distances between residences and work locations; the daily activity and travel patterns of individuals, vehicular emissions, air quality levels, and population exposure for different urban design scenarios are predicted in this study. Thus, this study will further add to the body of literature on sustainable urban forms that seek to improve public health through policy interventions focusing on land use/urban form and transportation design.

5.2 Methods

5.2.1 Study Area and Pollutant Focus

This study is focused on Hillsborough County, Florida, a county with an estimated population of 1.3 million, and its largest city is Tampa. It is a predominantly urban county, with an estimated 96.5% of the population residing in the urbanized areas (US Census Bureau, 2010b). The county provides an interesting setting to conduct this research due to the limited transit availability, dependence on automobile for travel, and unsatisfactory air quality record (American Lung Association, 2011). Additionally, the metropolitan area of Tampa-St. Petersburg-Clearwater figures in the top 100 sprawling metro areas in the US (Smart Growth America, 2014). More recently, the county is planning to expand the current interstate system by adding express toll lanes (Florida Department of Transportation, 2017).

The impact of these automobile-oriented expansions on the county's air quality and population exposures, especially for the vulnerable population groups, is largely unclear.

HEI identified NO_x as a potential surrogate for traffic-related pollution (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Thus, this study chose NO_x as a surrogate for the more complex mix of traffic-related pollution in the study area. Additionally, NO_x is associated with a variety of adverse health outcomes including reduced lung function, wheezing, and asthma (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010).

5.2.2 Modeling Framework

The integrated modeling framework that comprises activity-based travel demand simulation, dynamic-traffic assignment simulation, emissions estimation model, and pollutant dispersion, and described in Chapter 4, was used to simulate the effect of alternate land use and transportation scenarios on regional travel, air quality, and population exposure. Briefly, the activity-based travel demand model DaySim was used to estimate the initial travel demand of the study region. DaySim employs the principle of utility-maximization and estimates individual daily activity and travel patterns using a suite of econometric models including multinomial and nested logit models. Since this initial travel demand from DaySim does not provide the travel route information for individuals, the dynamic traffic-assignment model MATSim was used to estimate the specific route of travel. In this process, MATSim also provides an updated set of activity and travel information that is consistent with the network travel conditions during the simulation. In contrast to Chapter 4, which focused on simulating car and ride mode trips, this study includes simulation of additional modes of travel such as public transit, walk, and bicycle. Thus, MATSim provides the updated activity and travel information along with the distribution of automobile and public transit vehicular volumes on the roadway network.

Following this, the vehicular volumes on the roadway network were input to MOVES to estimate the hourly roadway link-level emissions. Similar to Chapter 4, the emissions were estimated for an average winter day and the default vehicular distribution on the roadways was used; however, in this study, public transit vehicles were additionally considered for the estimation of emissions. These link-

level emissions were then input to R-LINE to estimate the hourly concentrations for the winter months. Once the hourly concentrations were estimated for the winter months, they were processed to obtain a diurnal average. To estimate the population exposure to NO_x , these diurnal average concentrations were spatially and temporally matched with the locations of individuals. Exposures during travel were explicitly calculated using the travel route information from MATSim. For a more detailed description of the modeling framework, refer to Chapter 4.

5.2.2.1 Specifications for the Transportation Models

To accurately represent the vehicular emissions resulting from daily activity and travel patterns, it is important to consider the inter-regional travel. Thus, this study focused on characterizing the travel within and between Hillsborough County and its surrounding counties using the Tampa Bay activitybased travel demand model (TBABM) developed for the Florida Department of Transportation's (FDOT) District 7 jurisdiction (Gliebe et al., 2014). District 7 includes Hillsborough, Pinellas, Pasco, Hernando, and Citrus counties. Hence, the travel demand was derived for the full projected population in 2040 using TBABM.

Consequently, this initial travel demand was input to MATSim to obtain an updated set of daily activity and travel information along with detailed route information for individuals in the District 7. Due to computational feasibility, MATSim runs were performed using a randomly-chosen 10% population as opposed to the use of full population. Since the simulation used only a sample of the population, the capacities of the highway infrastructure and the transit vehicle sizes were proportionately reduced to simulate real-world conditions (Horni et al., 2016). This was operationalized by setting the flow capacity and storage capacity factors to 0.1 and 0.18, respectively. Similarly, the passenger car equivalent (PCE) value for the transit services was proportionately scaled down using a factor of 0.1.

As mentioned previously, this study includes the simulation of travel modes including car, public transit, shared ride, walk, bicycle, and school bus. To facilitate the simulation of car mode, a hypothetical 2040 transportation roadway network prepared by the FDOT was used (Florida Department of Transportation, 2015). To simulate public transit, MATSim requires an additional set of transit-related

input files that describe the spatial distribution of the stop locations, presence of bus bays, route, schedule, and the physical characteristics of vehicles (e.g., seating and standing capacity, vehicle length) for each transit line. These transit-related input files were created based on the 2040 transit-schedule information provided by FDOT (Florida Department of Transportation, 2015). Further details about the transit inputs are provided in Section 5.2.3.2, as these inputs vary for the low and enhanced-transit infrastructure scenarios. Ride mode users correspond to the individuals who travel via the car mode as passengers. Therefore, ride trips ideally should make route choices similar to that of car trips but without using the roadway capacity. To facilitate the simulation of ride mode trips, the maximum travel speed for the ride mode was set equal to that of the car mode, and the PCE value was set to zero. To accurately simulate the route choices for the bicycle and school bus modes, information on the availability of bicycle paths and school bus routes and schedules is needed. However, this information is not readily available for the FDOT-supplied transportation network. Therefore, bicycle and school bus trips were assumed to use the same roadway network and travel routes as car. The PCE for these two modes was reduced sufficiently so as to not impact roadway capacity. Moreover, travel speed for the bicycle mode was set as 15 km/h, and the travel speed for school bus was set equal to the car mode. Finally, walk mode trips were assumed to travel 1.3 times the beeline-path distance between the origin and destination at a speed of 5 km/h.

MATSim provides a variety of strategies that focus on time, route, and mode innovation to simulate individual daily activity and travel patterns (Horni et al., 2016). This study used the mode innovation, time-allocation-mutator, and reroute strategies. Collectively, these strategies help to optimize individual daily activity and travel patterns by minimizing their daily travel time. More specifically, the travel time reductions are achieved through the substitution of car mode with alternate travel modes such as public transit and bicycle for sub-tours, alteration of trip departure times, and exploration of alternate travel routes. In each iteration, the mode innovation strategy was applied for 20% of the population, the time mutation and reroute strategies were simultaneously applied for 20% of the population, and the remaining 60% of the population stick to their initial (or previously-optimized) activity and travel schedules.

5.2.2.2 Specifications for the Air Pollution Models

The activity-based travel demand outputs from TBABM and MATSim pertain mainly to daily personal travel. Thus, the non-personal or commercial travel including freight was not considered for emissions estimation. To estimate the roadway link-level NO_x emissions, three MOVES onroad source vehicle types, i.e., passenger cars, passenger trucks, and transit buses, were used. Here, passenger cars refer to any coupes, compacts, sedans, or station wagons whose primary purpose is to carry passengers (US Environmental Protection Agency et al., 2015). Passenger trucks refer to light-duty trucks including pickups, sport utility vehicles (SUVs), and vans that are mainly used for the purpose of personal travel (US Environmental Protection Agency et al., 2015). The percentage of transit buses on a roadway link was determined by analyzing the hourly vehicle volumes output from MATSim. However, for car mode trips, separating passenger car volumes from passenger truck volumes is slightly more challenging, because neither TBABM nor MATSim delineate passenger car trips by vehicle type. Therefore, passenger car and passenger truck share for every roadway link was assumed to be 56% and 44% of the automobile volumes on the corresponding link. This share is based on the distributions of VMT by vehicle type in the US for 2010 (Davis et al., 2016).

For the R-LINE dispersion modeling, the surface roughness and displacement height for Tampa were chosen based on guidelines in Grimmond and Oke (1999); specifically, the ratio of displacement height to roughness length is assumed to be 5. Additionally, the initial dispersion for the plumes created from the line sources is assumed to be 1.2 based on an average vehicle height of 1.5 m and in accordance with the US EPA's guidance for hot-spot analysis (US Environmental Protection Agency et al., 2010). Using these parameters, NO_x concentrations were estimated for the winter months, i.e., November through March. The receptor grid is made of 13,806 receptors evenly spaced at 500 meters. Meteorological data for Tampa International Airport for 2010 were obtained from the National Climatic Data Center. Further modeling details pertaining to the specific urban design scenarios are presented below.

5.2.3 Alternate Urban Design Scenarios

Three alternate urban land use, population redistribution, and transportation infrastructure scenarios were used in this study to understand the impact of transit-oriented compact-growth strategies on population exposure to NO_x . All scenarios were implemented for the 2040 model year. The three scenarios include a low-bus service (low-transit) scenario that implements the 2010 bus-transit infrastructure, an enhanced-bus service (enhanced-transit) scenario that uses the planned 2040 bus-transit infrastructure, and a transit-oriented compact (compact-growth) scenario that uses the 2040 bus-transit infrastructure and increases residential density. To control for potential confounding factors that impact vehicular emissions and NO_x concentrations and to systematically identify the impact of urban transportation, population redistribution, and land use characteristics on population exposure, the modeling specifications discussed in sections 5.2.2.1 and 5.2.2.2 were held constant across the three scenarios.

5.2.3.1 Spatial Distribution of Population

Figure 5.1a shows the spatial distribution of the 2040 base residential density (used in both the low-transit and enhanced-transit scenarios); Figure 5.1b shows the spatial distribution of the difference in residential density between the compact-growth scenario and the low-transit/enhanced-transit scenarios. The distribution of population demographics for 2040 is done by the Hillsborough County Planning Commission (Hillsborough Metropolitan Planning Organization, 2014), which used 2010 as the base year and updated the population distribution every five years until 2040. Specifically, the population growth projections made by the Florida Bureau of Economic and Business Research were used as the control totals for the future year. For each TAZ, they developed an attractiveness index based on the vacant developable acres and inverse-weighted it by the square of distance between activity centroids and the vacant developable land. In the compact-growth scenario, the residential distribution is densified. Specifically, an attractiveness index was developed to redistribute the households in the study region. The attractiveness index (AI) for every parcel *i* in the study region was calculated as shown in equation 5.1.

$$AI_i = \frac{t_i}{\log D_{t_i}} \sum_{k=1}^n \frac{r_k}{\log D_{r_k}}$$
5.1

k represents a parcel within a 0.5-mile buffer around the origin parcel, r_k is the number of retail and service type of jobs in the k^{th} parcel, t_i is 1 if no bus stops are present in a 0.5 mile buffer around the i^{th} parcel and 0 otherwise, D_{r_k} is the distance in feet between the i^{th} parcel and the k^{th} parcel, and D_{t_i} is the distance in feet between the i^{th} parcel and the nearest bus stop. Overall, the attractiveness index assigns weight to a parcel based on the number of service and retail jobs available near it, availability of a walkaccessible bus stop, and the distance to job locations and the nearest bus stop; the parcels that are closest to both locations with a high number of jobs and a bus stop have higher weights. In essence, the attractiveness index developed aims at achieving compaction by directly capturing some of the key D variables including density, diversity, and distance to transit as identified by Ewing and Cervero (2010).



1 16 256 4096 35155 -3297 0 5000 10000 15000 21219 Figure 5.1 Spatial distribution of block group-level residential density in 2040 base and compact-growth scenarios. a) base residential density for 2040, b) difference in residential density between hypothetical compact growth scenario and base scenario. Note that low-transit and enhanced-transit scenarios use base residential density.

Following the development of the attractiveness index, 50% of households that fall in parcels with an attractiveness index below 75th percentile were randomly chosen for reallocation. New parcels were randomly chosen from the set of all parcels with probability (p_i) given in equation 5.2.

$$p_i = \frac{AI_i}{\sum_{i=1}^n AI_i}$$
 5.2

Thus, about 37.5% of households in the study region were reallocated from parcels with a low attractiveness index to parcels with a high attractiveness index. As shown in Figure 5.1, residences were more spread out in the 2040 base case compared with the compact-growth scenario. Due to the population reallocation, the residence density of several block groups that form the urban core of Hillsborough County has increased. The mean residential density in the compact-growth scenario is 1199 households/km² and represents an increase of 27% compared to the base residence density in 2040. The highest increase in residence density of 250% is observed for a block group in Downtown near the Selmon Expressway. Conversely, the largest drop in residence density of 49% is observed in the Town 'N' Country area.

The high-density block groups resulting from population reallocation fall primarily along I-275, Dale Mabry Highway, Selmon Expressway, near the University area, Downtown Tampa, Brandon, Mango, and Plant City. Particularly, the highest increase in residential density is observed near Downtown Tampa, the University area, and Tampa International Airport. Consequently, the block groups that surround the urban core of Tampa, Brandon, Mango, and Plant City witness a drop in residential density.

5.2.3.2 Bus-Transit Infrastructure

Figures 5.2a and 5.2b show the spatial distribution of 2010 bus service and 2040 bus service, respectively. The 2010 bus service is used in the low-transit scenario, and the 2040 bus service is used in the enhanced-transit and compact-growth scenarios. The county plans to migrate its entire bus-fleet to compressed natural gas (CNG) by 2040. However, to elicit the impact of transportation infrastructure and land use on air quality while controlling for the effect of vehicle and fuel technologies, diesel-powered

buses were simulated in the three alternative urban design scenarios. The bus infrastructure plan for 2010 was created by reducing the frequency of services and removing the additional bus routes from the FDOT-supplied 2040 bus infrastructure plan. Thus, after this adjustment, the newly-created 2010 bus infrastructure plan closely resembles District 7's original transit scheme for 2010. The 2010 bus services comprise 6284 bus stops, 94 routes, and 2811 km of bus-serviced roadways, and 2040 bus services include 8754 bus stops, 195 routes, and 5413 km of bus-serviced roadways.



roadway type — highway only — highway and transit

Figure 5.2 Highway and transit infrastructure in 2040 for low-transit and enhanced-transit scenarios

An overall summary of the scenarios and their urban form and transportation characteristics is provided in Table 5.1. The enhanced-transit scenario captures the impact of additional bus services on the local air quality and population exposure; similarly, the compact-growth scenario captures the impact of both additional bus services and compact urban development on the regional air quality and population exposure. Table 5.1 Summary of urban land use and transportation infrastructure characteristics for three alternate urban design scenarios

Urban Form and	Scenario		
Transportation Characteristics	Low Transit	Enhanced Transit	Compact Growth
Urban form	2040 base population distribution		Reallocated base population
	Lower residential density		Higher residential density
Transportation	2040 highway		
	2010 bus service		2040 bus service

5.3 Results

5.3.1 Mode Shares and Travel Characteristics for Alternative Urban Design Scenarios

The travel mode shares of daily personal trips for the three urban design scenarios are shown in Figure 5.3. The initial mode shares resulting from the DaySim model and the updated shares following the MATSim model are presented separately. The relative ranking of most of the mode shares was same in both DaySim and MATSim models, with the exception of the bicycle mode, with MATSim comparatively lower than DaySim for the three scenarios.



Figure 5.3 Mode shares for low-transit, enhanced-transit, and compact-growth scenarios. Mode shares shown follow simulation in a) DaySim and b) MATSim. Low-transit simulates 2010 transit bus services; enhanced-transit simulates 2040 bus services; compact-growth simulates both 2040 bus services and population compaction.

Overall, in all of the scenarios, the car mode drew the highest share; however, its share dropped from the low-transit scenario to the enhanced-transit scenario and further dropped for the compact-growth scenario. This decline was more discernible in the MATSim model results, with the drop amounting to 2.3% and 9% from low-transit to enhanced-transit and compact-growth, respectively. In contrast to the car mode, both the walk and transit modes experienced a rise in their shares from the low-transit to the compact-growth scenarios; the mode share gain for walk was much higher compared to transit. Specifically, the increase in the share of walk mode from low-transit to enhanced-transit and compact-growth was 1.1% and 7.1%, respectively; the increase in transit share from low-transit to enhanced-transit and compact-growth was 1.2% and 1.8%, respectively. Similar to the walk and transit mode shares, the mode share for bicycle also generally increased from low-transit to compact-growth, although this increase was relatively low. The mode share for the school bus remained relatively constant across all the scenarios.

In addition to shifts in mode shares, the three urban design scenarios led to changes of other travel measures, including travel times and distances. The percent change in the travel measures for the enhanced-transit and compact-growth scenarios when compared with the low-transit scenario are shown in Figure 5.4. The total daily trips predicted in the enhanced-transit scenario was less than that in the low-transit scenario by 0.5%; however, the total daily trips in the compact-growth scenario was very similar to the low-transit scenario. Compared to the low-transit scenario, both the cumulative daily travel time and travel distance for the enhanced-transit and compact-growth scenarios were low, although the reductions in the enhanced-transit scenario were more muted compared to the compact-growth scenario. It should be noted that despite no reduction in the overall number of trips, the compact-growth scenario led to reductions in the travel distances and times.



Figure 5.4 Percent change in cumulative travel distance, travel time, and number of trips for enhanced-transit and compact-growth scenarios compared with low-transit scenario. Low-transit simulates 2010 transit bus services; enhanced-transit simulates 2040 bus services; compact-growth simulates both 2040 bus services and population compaction.

5.3.2 Distributions of Emissions and Concentrations of NO_x

Figure 5.5 shows the diurnal emissions for the alternate urban design scenarios. Emissions in all scenarios displayed a similar diurnal trend with a morning peak from 7:00–9:00 AM and an evening peak from 4:00–6:00 PM. The peak emissions in the evening were higher compared to the morning by 15% for the low-transit and enhanced-transit scenarios and 12% for the compact-growth scenario. The daily aggregate emissions in the low-transit, enhanced-transit, and compact-growth scenario were 47.9, 48.7, and 42.8 tonnes, respectively; thus, the total emissions in the low-transit scenario were 2% less compared to the enhanced-transit scenario and 11% more compared to the compact-growth scenario. The emissions in all scenarios were higher compared to the daily auto-only emissions (20.4 metric tonnes) for 2010 estimated in Chapter 4. The higher emissions in the 2040 scenarios compared to 2010 can predominantly be attributed to an increase in auto-driver trips by 42%, 40%, and 30% for the low-transit, enhanced-transit, and compact-growth scenarios, respectively. Additionally, emissions from bus-transit were also included in the 2040 scenarios.



Figure 5.5 Diurnal NO_x emissions for low-transit, enhanced-transit, and compact-growth scenarios. Low-transit simulates 2010 transit bus services; enhanced-transit simulates 2040 bus services; compact-growth simulates both 2040 bus services and population compaction.

Figures 5.6 and 5.7 show the diurnal cycle of the domain-average NO_x concentrations and the distribution of hourly NO_x concentrations for the three urban design scenarios, respectively. The morning peak for the diurnal concentrations led by 1 hour compared to the emissions; thus, the highest mean concentrations were observed from 6:00–8:00 AM. Similarly, the peak hour concentrations in the evening were observed from 5:00–6:00 PM as opposed to 4:00–6:00 PM for the emissions. The peak concentrations in the morning were higher compared to the evening; this trend was in contrast with the diurnal trend for emissions.

The domain-average hourly-mean concentration in the winter season for the low-transit scenario was 10.7 μ g/m³. The hourly-mean concentrations in the enhanced-transit and compact-growth scenarios were 2% higher and 9% lower than the low-transit scenario, respectively. The maximum concentrations for the low-transit, enhanced-transit, and compact-growth scenarios were 5072, 5314, and 7321 μ g/m³, respectively, and were observed along the insterstate corrirdors of I-275 and I-4 between 5:00–6:00 PM, as shown in Figure 5.8.



Figure 5.6 Diurnal cycle of domain-average NO_x concentrations for low-transit, enhanced-transit, and compact-growth scenarios. Low-transit simulates 2010 transit bus services; enhanced-transit simulates 2040 bus services; compact-growth simulates both 2040 bus services and population compaction.



Figure 5.7 Distribution of hourly NO_x concentration for low-transit, enhanced-transit, and compactgrowth scenarios. Lower whisker given by max(min(x), Q1-1.5*IQR), upper whisker given by min(max(x), Q3+1.5*IQR), where x represents vector of concentrations, Q1 is 25th percentile, Q3 is 75th percentile, IQR is Q3-Q1.



Figure 5.8 Spatial locations of maximum NO_x concentrations for low-transit, enhanced-transit, and compact-growth scenarios

Additionally, Figures 5.9, 5.10, and 5.11 show the spatial distributions of the differences in NO_x concentration between the enhanced-transit and low-transit scenarios, the compact-growth and low-transit scenarios, and compact-growth and enhanced-transit scenarios, respectively. Overall, NO_x concentrations in the low-transit scenario were higher compared to the enhanced-transit scenario in a few outer geography pockets surrounding Tampa's urban core. The concentrations in the enhanced-transit scenario were higher than the low-transit scenario within the urban core of Tampa, especially along the I-275 commute corridor. A similar and more accentuated trend was observed for the concentration differences between the compact-growth and low-transit scenarios. Concentrations in the compact-growth scenario were higher than the low-transit scenario almost entirely within Tampa's urban core along the I-275 starting from the University area, I-4, and Dale Mabry Highway. For the rest of the county, the concentrations in the compact-growth scenario were lower compared with the low-transit scenario. The concentrations in the compact-growth scenario. The concentration differences between the compact-growth scenario were lower compared with the low-transit scenario. The concentration differences between the compact-growth scenario were lower compared with the low-transit scenario.

to those between the compact-growth and low-transit scenarios. The only difference was that the urban core, where the concentrations in the compact-growth scenario were higher is spatially smaller when compared with the enhanced-transit scenario (Figure 5.11) instead of the low-transit scenario (Figure 5.10).



Figure 5.9 Spatial distribution of the difference in NO_x concentrations between enhanced-transit and low-transit scenarios (enhanced transit-low transit) for morning and evening peaks hours


Figure 5.10 Spatial distribution of difference in NO_x concentrations between compact-growth and low-transit scenarios (compact growth-low transit) for morning and evening peaks hours



Figure 5.11 Spatial distribution of difference in NO_x concentrations between compact-growth and enhanced-transit scenarios (compact growth-enhanced transit) for morning and evening peaks hours

5.3.3 Population Exposure

Figure 5.12 shows the distribution of the exposures to NO_x for the individuals in the low-transit, enhanced-transit, and compact-growth scenarios. The mean population exposure concentration in the low-transit scenario was 22.7 μ g/m³, and the mean exposure concentrations in the enhanced-transit and compact-growth scenarios were higher than the low-transit scenario by 3.3% and 29%, respectively. The spatial distribution of the differences in daily exposure density between the enhanced-transit and lowtransit scenarios and compact-growth and low-transit scenarios is shown in Figure 5.13. The mean exposure density for the enhanced-transit and compact-growth scenarios was approximately 3.3% and 33.3% higher than the low-transit scenario, respectively. The block groups with high exposure density in the enhanced-transit scenario compared with the low-transit scenario were interspersed throughout Tampa's urban core and the suburban areas. In contrast, the high exposure density block groups in the compact-growth scenario were concentrated primarily in the urban core of Tampa along I-275, I-4, and Dale Mabry Highway. The highest increase in exposure density in the compact-growth scenario was observed in block groups near the Downtown, especially those between the Selmon Expressway and I-275. High exposure density was also observed in the block group below Tampa International Airport. Low-exposure densities were observed along the I-75 corridor in the southern part of the county.



Figure 5.12 Distribution of population exposure for low-transit, enhanced-transit, and compact-growth scenarios. Lower whisker given by max(min(x), Q1-1.5*IQR), upper whisker given by min(max(x), Q3+1.5*IQR), where x represents vector of concentrations, Q1 is 25th percentile, Q3 is 75th percentile, and IQR is Q3-Q1.



Figure 5.13 Differences in block group-level aggregated exposure densities between different scenarios. Exposure density differences are shown between a) enhanced-transit and low-transit and b) compact-growth and low-transit scenarios.

5.4 Discussion

This study provides complementary evidence on the impact of urban design featuring transitoriented compact-growth policies on population distribution, traffic emissions, concentrations, and population exposure. Transportation and air pollution models were used to estimate high resolution spatiotemporal distributions of individuals, vehicular activity, and pollutant concentrations. In the study, an increase in household (and population) density was observed in the compact-growth scenario, which employs transit-oriented population compaction policies; the population density in the compact-growth scenario was 7146 people/km², which represents an 8% increase compared to the 2040 base population distribution in the low-transit and enhanced-transit scenarios. This is similar to the findings of Stone et al. (2007), who reported a mean increase in density between 6.6 and 26.8% for different metropolitan statistical areas in their compact growth scenario; similarly, Hixson et al. (2009) created a high-density transit-oriented scenario with an estimated population density of 3935 people/km².

The drop in VMT in this study as a result of simulating transit-oriented compact-growth development is about 10%. This is consistent with the findings of Gim (2012), who performed a metaanalysis on the relationship between density and travel behavior and concluded that higher densities lead to reduced auto travel in the US (although muted compared to Europe). Additionally, Stone et al. (2007) estimated a median drop in VMT of 6% for a compact-growth scenario when compared to projected business-as-usual growth. Similar reductions in VMT due to increases in residential density were reported by Chattopadhyay and Taylor (2012).

Compact and mixed-use urban forms reduce VMT and boost alternate modes of travel, including walk, transit, and bicycling (National Research Council et al., 2009). In this study, lower share for the auto mode was observed with a concomitant increase in the share for the walk mode in the compact-growth scenario. Only a marginal increase in the share for the transit mode was observed in the compact-growth scenario (3.1% and 2.5% in the compact-growth and enhanced-transit scenarios, respectively, as opposed to 1.3% in the low-transit scenario). Additionally, the shares for the bicycle mode for the three scenarios remained the same. Primarily, two reasons are hypothesized for the lower shares of the transit mode—one, the 2040 hypothetical transit envisioned by the county is simply inadequate for attracting additional transit riders, and two, the attractiveness index developed in this study controls for the presence of transit at individual residences but did not consider the availability of transit at the travel destinations. Previously, it has been shown that transit ridership is primarily dependent on the connectivity between origins and destinations (Arrington & Cervero, 2008). The reason for low bicycle mode shares is unclear.

Overall, air quality in the transit-oriented compact-growth scenario slightly improved. Emissions and concentrations in the compact-growth scenario were lower by 11% and 9%, respectively, compared to the low-transit scenario. This is consistent with the findings of Yu and Stuart (2017), who looked into the effects of compact-growth on the regional emissions, concentration, and population exposure for the Tampa Bay area. They found that regional on-road NO_x emissions in the compact scenario were reduced by 29% compared to the sprawled-growth scenario. However, in their compact-growth scenario, a significant portion of the region-wide future population was reallocated to Hillsborough County; this resulted in 20% higher on-road NO_x emissions for the county in the compact-growth scenario compared to the sprawled-growth scenario. Similarly, Schweitzer and Zhou (2010) studied 80 metropolitan areas and reported lower ozone concentrations in the compact urban forms. Finally, Hixson et al. (2009) also

reported reductions in NO_x emissions when pursuing a compact-growth scenario. However, the emissions and concentrations in the enhanced-transit scenario were higher compared to those in the lowtransit scenario. This could be due to the insufficient offset of emissions as a result of lower travel mode shifts from car to bus. In addition to the low mode shift, the increased bus frequencies and the addition of new diesel-powered buses seems to have led to higher emissions. For example, the daily total NO_x emissions for the bus-only roadway links (i.e., only buses travel on these links) is 796 grams/meter for the enhanced-transit scenario as opposed to 73 grams/meter for the low-transit scenario, an increase of almost 1000%. Similarly, the enhanced-transit scenario records daily total emissions of 58,740 grams/meter (an increase of 68% compared to low-transit scenario) for bus links (i.e., other travel modes were allowed on these links apart from bus). However, for non-bus links (i.e., no buses travel on these links), the daily total emissions in the enhanced-transit scenario is 34,018 grams/meter, i.e., 38% lower compared to the low-transit scenario. This suggests that transit intensification strategies, if not targeted precisely, may lead to the deterioration of air quality; hence, transit investment in itself, which several studies use as a predictor for increased share of the transit mode (for example, Hixson et al. (2009)), may not always be a reliable indicator for increased transit use. Additionally, it is not clear if the air quality results in this study will hold with other types of transit, such as CNG-powered buses, light rail, and heavy rail. Nonetheless, compact urban design policies in conjunction with competent transit plans that displace a significant portion of auto drivers to the transit mode may hold the key for improving air quality.

Although the compact-growth scenario marginally improves the urban air quality in the study area, the population exposure is higher compared to the low-transit and enhanced-transit scenarios. This seems to be in contrast with Yu and Stuart (2017), who reported lower population exposure to NO_x in compact scenarios compared to sprawl scenarios for the same study region. However, they also reported higher exposures under compact scenarios for butadiene and benzene, thus arguing that compact forms may have differential effects on population exposure depending on the mix of pollutant sources. Similarly, Schweitzer and Zhou (2010) reported higher neighborhood exposures to ozone and $PM_{2.5}$ in compact regions. Hixson et al. (2009) found 10–15% higher exposure to primary $PM_{2.5}$ components such

as elemental carbon and organic carbon in high-density development scenarios. Thus, compact urban forms by themselves may not always lead to reductions in population exposure. Perhaps they need to be combined with other strategies such as development of public transit infrastructure that improves accessibility between activity locations, urban design that encourages alternate modes of travel including walk and bicycle, fuel and vehicle technologies that lead to lesser life-cycle emissions, and displacing pollutant sources from high-density population zones. A combination of these strategies may lead to lower exposures and better health outcomes especially for the vulnerable population groups.

5.5 Limitations

This study has several limitations, one of which arises from the use of parameters for the activitybased travel demand model from the Sacramento region instead of Tampa. The available sample sizes to estimate the travel demand model parameters for Tampa were insufficient; thus, model parameters were borrowed from the Sacramento region by the developers of the model (Gliebe et al., 2014). Although the model developers concluded that it is preferable to borrow parameters from regions with large sample sizes than estimating parameters with insufficient local data, estimating travel demand based on parameters from a different urban region may introduce some uncertainty and variability.

Although the traffic on the roadways was simulated using MATSim, information on toll roads was not included in the simulation. This could lead to biased estimates of the spatial distribution of traffic in the urban region. The study did not include the emissions from commercial traffic such as freight, shipping, and other on-road sources such as school buses. Further, emissions from point and area sources were not included. Thus, it is not clear if the observed trends in concentrations and population exposure will remain the same even after the inclusion of these additional sources of pollution.

The attractiveness index developed in this study solely considers transit and job accessibility at the residence locations of individuals. However, Arrington and Cervero (2008) argued that transit accessibility between origin and destination is important for improving transit mode share. Additionally, accessibility to other activity locations such as shops, hospitals, and entertainment places was not

considered. Thus, the compact urban form employed in this study may not entirely represent a mixed-use development.

Finally, the transit infrastructure simulated here entirely comprises diesel buses. It is highly unlikely that the county will pursue diesel buses in 2040. Additionally, Hillsborough County's Long Range Transportation Plan includes light rail for 2040 (Tampa Bay Area Regional transportation Authority, 2015). However, the rail mode was not included in the activity-based model by the model developers. As such, the impact of this hypothetical light rail transit on the county's air quality and population exposure was not simulated.

5.6 Conclusion

This study investigated the impact of a transit-oriented compact-growth scenario on population distribution, vehicular travel and emissions, concentrations, and population exposure. Adding more diesel-powered bus routes and improving bus frequencies increased NO_x emissions, leading to higher exposures. Thus, the bus-transit plan adopted for Tampa may not be adequate to cause sufficient travel mode shifts and may, in fact, deteriorate the air quality. Additionally, the compact urban forms co-located individuals near major roadway sources, thus exacerbating their exposures. Hence, there is a need for collaborative solutions from public health and urban design professionals that seek to improve air quality and population health. Future research efforts should consider alternate modes of transit, including light and heavy rail, which improve accessibility between locations and urban design plans that proliferate mixed-use neighborhoods.

CHAPTER 6: SYNTHESIS

6.1 Introduction

Over the past six decades, the world has witnessed continued migration from the rural regions to the urban regions, giving rise to a phenomenon known as urbanization. This urbanization trend is expected to continue at least until 2050 (United Nations et al., 2015). Recognizing the potential impacts of this large-scale population movement on urban regions, the heads of governments across the world have come together to adopt policies consistent with the 2030 agenda for sustainable development (UN General Assembly, 2016). Sustainable development is defined as the type of development that meets the needs of the current generation without taking away from the resources of future generations; thus, sustainable development features elements of intra-generational and inter-generational equity (Brundtland Commission, 1987). The 2030 agenda for sustainable development stipulates targets that include safe and sustainable transport for all, betterment of population health through improvement of air quality, and reduction of inequalities to shift the world onto a more sustainable and resilient path (UN General Assembly, 2015).

Although the 2030 sustainability agenda does not identify curbing air pollution as a standalone goal, it establishes clean air as a desirable target under multiple goals (Lode et al., 2016; UN General Assembly, 2015). This suggests the importance of clean air for improving human health and reducing environmental degradation. Within the urban context, the transportation sector emits a significant proportion of pollutants that are of interest to human health and climate change. Exposure to traffic-related air pollution assumes special interest due to the growing body of evidence on its associations with mortality, morbidity, and environmental justice concerns (Gurram et al., 2015; HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010; Hystad et al., 2015; Marshall, 2008; Schultz et al., 2015; Stuart et al., 2009; Stuart & Zeager, 2011; Yu & Stuart, 2013, 2016). Thus, improving air quality and

mitigating population exposure to traffic-related air pollution and the associated exposure inequalities directly helps to achieve urban sustainability, including the health and inequality-related targets from the 2030 agenda for sustainable development.

Within this context, an emerging field of research suggests that the design of urban form may have an impact on travel demand, regional and local air quality, and population exposure. Moreover, the US EPA's report on the built environment suggests that about two thirds of new development in the next 40–45 years is yet to be built (US Environmental Protection Agency, 2013). This provides the necessary impetus for transportation engineers and urban planners, air quality specialists, and public health policy experts to collaborate in the design of healthy and sustainable cities that improve air quality and population health. However, the underlying mechanisms that govern the relationships between urban transportation infrastructure, land use, air quality, and population exposure are not yet well understood.

To this end, a few major gaps in the literature were identified in this study. First, the importance of consideration of detailed spatiotemporal activity and travel patterns of individuals for the estimation of exposure to traffic-related air pollution is not extensively investigated in the current literature. Also, previous literature did not extensively study the distributions of exposure to traffic-related air pollution for different socioeconomic and urbanicity-related subgroups of the population. Moreover, several previous studies used human activity and travel patterns derived from travel surveys for exposure estimation. Given the limited intra-urban spatial representation of the activity and travel patterns of individuals in survey samples, it is not entirely clear if the exposure results from a survey-based sample analysis are robust compared to a population analysis. Similarly, the importance of using high-resolution activity and travel, and pollutant concentration data in modeling frameworks that estimate exposures has received little to no attention. Finally, previous studies that developed transportation and air pollution modeling frameworks for exposure estimation have not extensively used such frameworks to simulate air quality levels and personal exposures under alternate transportation and land use design scenarios. In view of this discussion, the overarching goal of this dissertation was to understand the linkages between urban transportation infrastructure and land use design, and population exposure to traffic-related air

pollution using a transportation and air pollution modeling framework. Understanding these linkages may aid the design of healthy and sustainable cities and communities.

Section 6.2 provides a summary of each of the aims and their corresponding results in this research. Section 6.3 discusses the policy implications of these findings, and Section 6.4 discusses the limitations of this work along with the directions for future research.

6.2 Summary of Results

6.2.1 Impact of Activity and Travel Patterns, and Urbanicity on Exposures to NO_x

In the study presented in Chapter 3, the activity and travel patterns of individuals generated using the 2009 National Household Travel Survey (NHTS) were combined with CALPUFF-modeled NO_x concentrations for 2002 to estimate the exposures and their social distribution for the residents of Hillsborough County, FL. To understand the importance of human activity and travel patterns for estimation of personal exposures, two exposure measures were estimated—activity-based exposure and residence-based exposure. The activity-based approach considered the movements of individuals in time and space along with time-dependent NO_x concentrations at those locations, whereas the residence-based approach estimated the exposures based on the concentrations at individuals' residences. Finally, to understand the important predictors of exposure, a multivariate regression model was estimated.

It was found that the population from the county's travel survey sample spent more time in suburban and rural areas compared to urban areas. However, time densities for nonresidential activities in urban areas were higher than those of residential activities. Although exposures in this study were found to be generally lower than the National Ambient Air Quality Standards (NAAQS), a large range of exposures was found, from 7.0 to $43 \,\mu g/m^3$. Additionally, disproportionately high mean exposures were found for blacks, Hispanics, low-income households or individuals, urban residents, and individuals with daily travel time above one hour.

On average, exclusion of the activity and travel patterns of individuals (i.e., use of a residencebased exposure measure) led to an underestimation of exposure by 3.6%. However, exposure errors were found to be lower for vulnerable population subgroups including blacks, Hispanics, low-income

individuals, and urban residents. NO_x concentrations during travel and at nonresidential locations were generally high compared to the concentrations at residential locations. However, on average, the concentrations at nonresidential locations made up a small portion of daily total personal exposure. Finally, urban form variables, time spent away from home, income, and race (Black category) were found to be significant predictors of exposure.

Overall, the results suggest the importance of considering the activity and travel patterns of individuals for exposure estimation. Additionally, the study found associations between urban form, activity and travel variables, and exposures. Thus, the study has implications for improving air quality and population health through urban design interventions.

6.2.2 Integration of Models for Travel Demand and Air Pollution to Predict Population Exposures

In the study presented in Chapter 4, an activity-based travel demand model (DaySim) was combined with a dynamic traffic assignment simulator (MATSim), mobile-source emissions estimator (MOVES), and a dispersion model (R-LINE) to estimate population exposure to traffic-related air pollution and its social distribution for the Tampa area. This modeling framework was developed to improve tools for understanding of the relationships between urban transportation and land use design, air quality, and population exposure. Whereas Chapter 3 focused on the estimation of the exposures for a population sample based on a travel survey, Chapter 4 refined the exposure estimation methodology (by computing travel paths of individuals that take congestion effects into consideration) and expanded the exposure analysis to the full population. This modeling framework was used for Hillsborough County for a typical winter day and was applied to explore the impact of using high-resolution data on population and subgroup exposures by computing exposure measures using high-resolution and low-resolution data.

Higher activity durations and densities (normalized by block group area) were found in the urban core of Tampa as opposed to the outer areas. By 7:00 AM, activities began spilling over into the University area, intersections near I-4 and I-75, and locations that generate employment and business in the urban core of Tampa; the high activity durations in these locations persisted until 8:00 PM. A morning peak (7:00–9:00 AM) and an evening peak (4:00–7:00 PM) were observed in the traffic volumes and NO_x

emissions that largely coincided with the commute patterns. The peak-hour emissions contributed more than 50% towards the daily total passenger car emissions in the county. Emissions and the resulting concentrations were high along the major freeway corridors, especially near the zones including Downtown, the airport, and University.

Although concentrations in this study were generally lower than other studies, evidence was found for high NO₂ concentrations (obtained using a NO₂/NO_x ratio) that potentially may be greater than the one-hour National Ambient Air Quality Standard (NAAQS) for NO₂ in some areas. Regarding demographics, higher mean exposures were observed for blacks, Asians, and other racial subgroups compared to whites and Hispanics compared to non-Hispanics, below-poverty and middle-income groups compared to the high-income group, and the active age groups (19–65 years old) compared to the rest of the age categories. Disproportionately higher mean exposures also were found for individuals residing in the urban regions compared with those living in rural areas and those whose daily total travel time exceeded one hour as opposed to individuals whose daily travel time was less than one hour. Moreover, it was found that exposure disparities increased as the levels of exposure to NO_x rose for blacks, Asians, other racial minorities, Hispanics, and below-poverty non-whites.

On average, use of low-resolution activity and travel, and concentration data appears to have resulted in the underestimation of exposures for the population and most of the studied subgroups. The mean underestimation of exposure due to the use of low-resolution data was about 10% for the population; the population subgroup that traveled for more than one hour per day had the greatest underestimation.

Overall, this study suggests the possible harmful impact of traffic-related air pollution on population health due to the current urban transportation infrastructure design, thus providing an impetus for further work on the complex linkages between urban design, air quality, and population health.

6.2.3 Impact of Transit-Oriented Compact-Growth Policies on Population Exposure

In Chapter 5, three alternate urban design scenarios were simulated for 2040 using an integrated modeling framework to understand the relationship between urban transportation infrastructure and land

use design, air quality, and population exposure. Specifically, a low-transit scenario was simulated with the 2010 diesel bus infrastructure and the 2040 default household distribution. The enhanced-transit scenario improved upon this by simulating the 2040 diesel bus infrastructure while maintaining the 2040 default household distribution. The compact-growth scenario used the same bus infrastructure as the enhanced-transit scenario but compactly redistributed the 2040 households to reduce the spatial distance between households, and jobs and transit stops. All the scenarios used the same 2040 highway network.

Compared to the low-transit scenario, a slight drop was observed in the share of auto mode in the enhanced-transit scenario, with a concomitant increase in the shares of walk and transit modes. The highest drop in the share of auto mode was observed in the compact-growth scenario. Most of this reduction in the share of auto mode was reallocated to the walk mode, with a small portion allocated to transit. Thus, the share of transit was rather low in both the enhanced-transit and compact-growth scenarios. Additionally, the enhanced-transit scenario had slightly decreased aggregate travel distances and travel times and total daily trips compared to the low-transit scenario. In contrast, the compact-growth scenario had lower aggregate travel distances and times, despite no apparent change in the daily total number of trips, compared to the low-transit scenario.

The NO_x emissions in 2040 for all the scenarios were higher than the emissions in 2010 (from Chapter 4). Additionally, the lowest and highest total emissions were observed in the compact-growth and enhanced-transit scenarios, respectively. Similarly, the lowest and highest domain-average hourlymean NO_x concentration was recorded for the compact-growth and enhanced-transit scenarios, respectively. Although the domain-average concentration for the compact-growth scenario was lower than the low-transit scenario (and the enhanced-transit scenario), spatial distribution of the differences in concentrations between the two scenarios revealed that concentrations were higher in the urban core of Tampa for the compact-growth scenario. Conversely, concentrations in the compact-growth scenario were lower compared to both the low-transit and enhanced-transit scenarios for a majority of the outer Tampa area surrounding the urban core. This is likely due to relocation of mobile-pollutant sources from outer/isolated peripheries to inner areas. Mean population exposure in the compact-growth scenario was higher compared to the enhanced-transit and low-transit scenarios. This is likely due to high exposure densities in the urban core.

Results suggest that compact urban forms potentially may lead to improved air quality; however, to realize similar exposure-related benefits, they may need to be implemented with transportation infrastructure designs that encourage transit-use and active modes of transport including walk and bicycle.

6.3 Implications

Overall, this dissertation seeks to add to the body of knowledge on the relationships between urban transportation infrastructure and land use design, air quality, and population exposure. Specific answers to the science questions determined through this dissertation along with a discussion of their potential science and policy implications are provided below.

6.3.1 Impacts of Activity and Travel Patterns on Exposure Estimates

Through a travel survey-based analysis, it was found that exclusion of activity and travel patterns for exposure estimation significantly underestimated individual exposures by about 3.6%. This is consistent with the findings of Setton et al. (2011). Although this exposure error may appear to be low, it potentially could result in inaccurate/biased health outcome assessments, especially at the high end of exposures. Additionally, this finding may have implications for developing nations where pollutant concentrations are orders of magnitude higher compared to those in developed nations. Thus, assessment of exposures using detailed activity and travel patterns of individuals may be important.

Exposure errors were found to be generally lower for vulnerable population groups, including blacks and below-poverty and middle-income groups. With regard to urban characteristics, high exposure errors were found for suburban and rural residents and individuals whose daily travel time is greater than one hour. Thus, using coarser approaches may particularly affect the health outcome assessments for individuals who travel for a significant portion of the day or whose activity radius is large.

6.3.2 Exposure Disparities

Statistically significant exposure disparities were found for minorities and low-income subgroups in the Hillsborough County. Higher exposures were found for urban residents and individuals whose

daily travel time is greater than one hour. Although NO_x levels observed in this study were somewhat low, a recent analysis on a cohort of about 61 million Medicare beneficiaries suggested that exposure to pollutant levels lower than the current NAAQS may still have an adverse impact on the overall population health, with pronounced effects on racial minorities and lower economic groups (Di et al., 2017); it should be noted that this study focused on exposures to $PM_{2.5}$ and ozone, although similar observations were made for exposures to NO_2 previously (Young et al., 2014). This challenges policy-makers to take a lead on intervention strategies that seek to reduce exposure disparities along with the overall exposure levels.

6.3.3 Robustness of Exposure Estimates using a Sample-Based and Population-Level Analysis

Distributions of exposure by sociodemographic and urbanicity characteristics had similar exposure disparity trends for the sample-based analysis (using travel survey data) and a population-level analysis. However, there was a substantive difference in the maximum personal exposure levels $(43 \ \mu g/m^3 \text{ for sample vs } 145 \ \mu g/m^3 \text{ for population})$. This is important considering that NO_x concentrations in the population analysis were estimated based only on passenger car emissions, whereas the sample analysis used additional emissions from other point and area sources to estimate NO_x levels. Thus, conducting a population-level analysis may be of help in identifying the high percentile exposures in both developed countries and the highly-polluted regions in developing countries where exposures generally exceed American and European standards.

6.3.4 Impacts of using Low-Resolution Data Versus High-Resolution Data on Exposure Estimates

The integrated transportation and air pollution modeling framework that was adopted in this study to estimate population exposure is computationally more intensive than similar frameworks from other studies (Beckx et al., 2009a; Hatzopoulou & Miller, 2010; Vallamsundar et al., 2016). Specifically, the framework used in this study estimated fixed-activity locations at a high spatial resolution of parcels and the locations of individuals during travel at every five seconds along the roadways; hence, the framework in this study is of higher spatial resolution compared to those in other studies. Therefore, one might naturally ask what are the benefits of such a computationally-intensive, high spatial resolution approach

used in this study? However, it was found in this dissertation research that use of low-resolution modeled data resulted in the underestimation of exposures on average, with a large range of error in exposures. This finding is consistent with previous studies that found that exposures during travel are of high importance to daily total exposure (de Nazelle et al., 2013; Dons et al., 2012; Gurram et al., 2015). Therefore, exposure modeling frameworks that do not incorporate highly resolved fixed-activity locations and travel path information may substantially underestimate exposures, thus potentially leading to inaccurate health assessments.

6.3.5 Impact of Transit-Oriented Compact-Growth on Air Quality and Population Exposure

Neither bus-transit intensification nor compact development, independently or collectively, were found to shift individuals to the transit mode beyond a very small increase in transit share. However, similar to previous studies that investigated the relationships between urban form and transport, it was found that compact-growth increased the travel share for the walk mode. Thus, compact-growth strategies may improve population health by inducing individuals to choose active modes of travel.

Air quality generally improved for the compact-growth scenario, except in the urban core areas, compared to the remaining two scenarios. This could be due to a combination of lower travel distances and travel times and a shift to active modes of travel that resulted in lower daily total NO_x emissions and diurnal concentrations. Thus, compact-growth scenarios may help to improve urban air quality levels.

Although air quality improved in the compact-growth scenario, on average, the mean population exposure was highest in the compact-growth scenario due to the relocation of households into the urban core of the study region. Thus, compact growth may exacerbate population exposure. However, de Hartog et al. (2010) suggest that shifting to actives modes of transport such as bicycling could potentially offset health risks due to exposure to air pollution. Thus, although compact-growth policies by themselves may not be a panacea for the negative outcomes due to exposure to traffic-related pollution, they may be combined with other policy initiatives such as active campaigning for bicycle and walk modes, incentivizing the use of active travel modes, and clean-fuel technologies to mitigate emissions and, therefore, population exposure.

Overall, bus-transit intensification did not result in substantial shifts to transit mode in the study area. The low shares for the transit mode in all scenarios suggest that transit planning may need to be rethought for the area. Moreover, several previous studies used transit investment as a surrogate for higher transit shares. However, as demonstrated in this study, that may not always be the case. This is not to say that transit investment is not important; in fact, transit investment plays a very significant role in expanding transit services and improving the mobility of individuals. But equal consideration must be given to transit planning such that both trip origins and destinations are equally serviced. Moreover, the increase of diesel bus services in the scenarios led to higher NO_x emissions and concentrations and a general deterioration of air quality. Thus, transit agencies may also need to give appropriate consideration to the type of fuel used in their services. For example, using compressed natural gas (CNG) powered buses in a compact-growth scenario may lead to the reduction of emissions from transit vehicles and improve the air quality.

6.4 Limitations and Future Research

This research has several limitations. A few important limitations along with the directions for future research are discussed below.

6.4.1 Travel-Survey Samples

Chapter 3 used a travel-survey sample to estimate the activity and travel patterns of individuals, which were then combined with NO_x concentrations to estimate exposures. Although the survey sample is fairly large, it may be limited in capturing the true spatiotemporal distributions of human activities. Additionally, simulating the vehicles from the travel survey on roadways would generally result in uncongested conditions, since the simulated volumes substantially underestimate the actual volumes on the roadways. Thus, estimating routes based on uncongested conditions may result in poor/biased travel path estimates. Overall, the use of travel surveys may result in spatially non-representative and inaccurate activity and travel patterns of individuals. Despite these limitations, it was found that the trends of social distribution of exposures for survey-based analysis was robust compared to a population analysis.

6.4.2 Single Pollutant Analysis

Throughout this dissertation, NO_x was the pollutant of interest. NO_x is an appropriate choice to represent the traffic-related air pollution, as the HEI panel identified it as a potential surrogate for exposures to traffic-related pollution (HEI Panel on the Health Effects of Traffic-Related Air Pollution, 2010). Moreover, NO_2 is a component of NO_x and a criteria pollutant regulated by the US EPA and has known health effects. However, the HEI panel also recognized that there is no ideal surrogate that accurately represents the complex mix of traffic-related air pollution. Thus, analysis based on a single pollutant may not fully represent the type of pollution in question. Considering this, future studies that use this framework may consider additional primary traffic-related pollutants including fine particulate matter, benzene, and carbon monoxide. Moreover, to understand the relationships between urban transportation and land use design and population exposure to secondary formations, pollutants including ozone and formaldehyde may be considered (Yu & Stuart, 2016).

6.4.3 Microenvironmental Exposure Factors

In this study, the effect of exposure adjustment factors such as indoor infiltration fractions or inhalation rates on human exposure to traffic-related air pollution were not considered. Furthermore, additive exposure to pollutants due to sources in particular microenvironments including the kitchen or car and exposure to other sources including smoking were excluded in this analysis. Only exposures to ambient NO_x concentrations from traffic were estimated. This may result in overestimation or underestimation of exposures. Since this study focused on exposure to ambient traffic pollution, the microenvironmental sources of pollution were ignored. However, future studies may address this issue by including microenvironmental adjustment factors along with inhalation intakes.

6.4.4 Data Requirements for the Integrated Modeling Framework

The integrated modeling framework used in this study needs significantly large computational resources and detailed data inputs. Given that many local planning agencies may not have such resources, application of this framework for smaller planning agencies may be a challenge. Thus, it is of interest to look further into the question of appropriate spatial and temporal resolution of the activity and travel and

air quality data that may be needed for the estimation of population exposure. On a related note, it is of interest to understand how exposure estimates vary as the sample size increases from one percent to the full population. Such an exercise may give an understanding of the appropriate sample size needed to accurately predict population exposure while meeting the resource constraints.

6.4.5 Health Impact Assessments

Although the modeling framework used in this study integrated several models that use highly spatially- and temporally-resolved data to estimate agent-based exposures, it was not extended to estimate the health outcomes of these exposures. Moreover, the additional health benefits resulting from pursuing active modes of travel were not simulated. Thus, coupling the framework presented herein with health impact assessment tools such as US EPA's Benefits Mapping and Analysis Program (BenMAP) may provide further insights into the potential health impacts of the simulated alternate urban design scenarios.

6.4.6 Model Uncertainty

The integrated modeling framework used in this study comprises a variety of transportation and air pollution models. Each model comes with its own set of uncertainties. Perhaps the greatest uncertainty in the framework comes from the activity-based travel demand model (ABM) due to the use of modeling parameters from the Sacramento region instead of those from the Tampa Bay area. It was found that Tampa had a higher proportion of retirees and non-work-trips compared to Sacramento, which may add uncertainty to the model predictions. Additionally, since the activity-based models were estimated using observed data and since it is difficult to accurately collect all the information pertinent for model estimations, this may add further uncertainty.

Following this, MATSim simulation results in further uncertainties, as evidenced by the differences in the observed and estimated volumes at count stations. Roadway link-specific volumes were used to estimate the link-specific NO_x emissions. Since an average speed method was used instead of the second-by-second vehicle specific power data, emission estimates may be prone to uncertainty. Additional uncertainty may result due to the use of national default vehicle age distributions instead of the local data for the county. Finally, all passenger cars were assumed to be of the same type in the MOVES

analysis, which is unrealistic and may result in uncertainties. Finally, uncertainties in exposure estimates from R-LINE may be due to the use of meteorological data from a single location, uniform values of surface roughness and displacement height for all of Tampa, and an assumed value for the initial dispersion of plumes.

Although the above-mentioned modeling parameters may result in uncertainties, they may be controlled by implementing the same parameters across multiple scenarios. Thus, the relative differences in travel, air quality, and exposure outcomes between the scenarios may be attributed to the differences in the transportation and land use design. Additionally, in future studies, the biggest source of uncertainty for the travel demand models may be addressed by re-estimating the model components using local data for Tampa. Finally, characterization and quantification of uncertainty in exposure estimates of such integrated models is an important avenue for future research.

6.5 Conclusion

This study investigated the relationship between urban transportation and land use design, air quality, and population exposure and its social distribution. Use of highly spatially- and temporally-resolved activity and travel, and pollution data led to more accurate exposure estimates. It was found that lower socioeconomic subgroups, minorities, urban residents, and individuals whose daily travel was greater than one hour were subject to disproportionately high NO_x exposures. Using an integrated modeling framework, it was found that compact development led to improved air quality for a substantial area of the urban region. However, mean population exposure increased under the compact development scenario compared to a scenario with lower population density. Similarly, intensification of diesel bus services led to higher emissions, concentrations, and population exposure, suggesting that the type of fuel used in transit may influence air quality levels. Overall, this study found associations between urban design, air quality, and exposures. Thus, urban design interventions may potentially improve air quality and population health.

REFERENCES

- [1] Abdulazim, Abdelgawad, Habib, & Abdulhai. (2013). Using smartphones and sensor technologies to automate collection of travel data. *Transportation Research Record: Journal of the Transportation Research Board*, 2383, 44-52. doi:10.3141/2383-06
- [2] Alig, Kline, & Lichtenstein. (2004). Urbanization on the US landscape: Looking ahead in the 21st century. *Landscape and Urban Planning*, 69(2-3), 219-234.
- [3] American Lung Association. (2011). State of the air 2011. Retrieved from http://www.stateoftheair.org/2011/assets/SOTA2011.pdf. Accessed on December 2, 2013.
- [4] Anderson, Favarato, & Atkinson. (2013). Long-term exposure to air pollution and the incidence of asthma: Meta-analysis of cohort studies. *Air Quality, Atmosphere & Health, 6*(1), 47-56. doi:10.1007/s11869-011-0144-5
- [5] Anderton, Anderson, Oakes, & Fraser. (1994). Environmental equity: The demographics of dumping. *Demography*, *31*(2), 229-248.
- [6] Apelberg, Buckley, & White. (2005). Socioeconomic and racial disparities in cancer risk from air toxics in Maryland. *Environmental Health Perspectives*, *113*(6), 693-699. doi:10.1289/ehp.7609
- [7] Arentze, & Timmermans. (2004). A learning-based transportation oriented simulation system. *Transportation Research Part B: Methodological*, 38(7), 613-633. doi:<u>http://dx.doi.org/10.1016/j.trb.2002.10.001</u>
- [8] Arrington, & Cervero. (2008). Effects of TOD on housing, parking, and travel. Washington, DC. Retrieved from: <u>http://www.nctcog.org/trans/sustdev/tod/TCRP%20128%20TOD%20effects%20on%20housing, %20parking,%20and%20travel.pdf</u>
- [9] Axhausen, Zimmermann, Schönfelder, Rindsfüser, & Haupt. (2002). Observing the rhythms of daily life: A six-week travel diary. *Transportation*, 29(2), 95-124. doi:10.1023/a:1014247822322
- [10] Baklanov, Hänninen, Slørdal, Kukkonen, Bjergene, Fay, ... Ødegaard. (2007). Integrated systems for forecasting urban meteorology, air pollution, and population exposure. *Atmos. Chem. Phys.*, 7(3), 855-874. doi:10.5194/acp-7-855-2007
- [11] Balmer, Rieser, Meister, Charypar, Lefebvre, Nagel, & Axhausen. (2009). MATSim-T: Architecture and simulation times. In *Multi-agent systems for traffic and transportation engineering*. Bazzan & Klügl (Eds.).(pp. 57-78). Hershey, PA: Information Science Reference.

- [12] Batterman, Chambliss, & Isakov. (2014). Spatial resolution requirements for traffic-related air pollutant exposure evaluations. *Atmospheric Environment*, 94, 518-528. doi:<u>http://dx.doi.org/10.1016/j.atmosenv.2014.05.065</u>
- [13] Baum, Garofalo, & Yali. (1999). Socioeconomic status and chronic stress. Does stress account for SES effects on health? *Ann N Y Acad Sci*, 896, 131-144.
- [14] Beckman, Baggerly, & McKay. (1996). Creating synthetic baseline populations. *Transportation Research Part A: Policy and Practice*, 30(6), 415-429.
- [15] Beckx, Int Panis, Arentze, Janssens, Torfs, Broekx, & Wets. (2009a). A dynamic activity-based population modelling approach to evaluate exposure to air pollution: Methods and application to a Dutch urban area. *Environmental Impact Assessment Review*, 29(3), 179-185. doi:<u>http://dx.doi.org/10.1016/j.eiar.2008.10.001</u>
- [16] Beckx, Int Panis, Uljee, Arentze, Janssens, & Wets. (2009b). Disaggregation of nation-wide dynamic population exposure estimates in the Netherlands: Applications of activity-based transport models. *Atmospheric Environment*, 43(34), 5454-5462. doi:<u>http://dx.doi.org/10.1016/j.atmosenv.2009.07.035</u>
- [17] Beckx, Int Panis, Van De Vel, Arentze, Lefebvre, Janssens, & Wets. (2009c). The contribution of activity-based transport models to air quality modelling: A validation of the ALBATROSS– AURORA model chain. *Science of the Total Environment*, 407(12), 3814-3822. doi:<u>http://dx.doi.org/10.1016/j.scitotenv.2009.03.015</u>
- [18] Beckx, Panis, Vankerkom, Janssens, Wets, & Arentze. (2009d). An integrated activity-based modelling framework to assess vehicle emissions: Approach and application. *Environment and Planning B: Planning and Design*, *36*(6), 1086-1102. doi:10.1068/b35044
- [19] Beelen, Hoek, Brandt, Goldbohm, Fischer, Schouten, . . . Brunekreef. (2008a). Long-term effects of traffic-related air pollution on mortality in a Dutch cohort (NLCS-AIR study). *Environmental Health Perspectives*, *116*(2), 196-202.
- [20] Beelen, Hoek, van den Brandt, Goldbohm, Fischer, Schouten, . . . Brunekreef. (2008b). Longterm exposure to traffic-related air pollution and lung cancer risk. *Epidemiology*, *19*(5), 702-710. doi:10.1097/EDE.0b013e318181b3ca
- [21] Beevers, Kitwiroon, Williams, Kelly, Ross Anderson, & Carslaw. (2013). Air pollution dispersion models for human exposure predictions in London. J Expo Sci Environ Epidemiol, 23(6), 647-653. doi:10.1038/jes.2013.6
- [22] Bereitschaft, & Debbage. (2013). Urban form, air pollution, and CO₂ emissions in large US metropolitan areas. *The Professional Geographer*, 65(4), 612-635. doi:10.1080/00330124.2013.799991

- [23] Bhugwant, & Brémaud. (2001). Simultaneous measurements of black carbon, PM₁₀, Ozone and NO_x variability at a locally polluted island in the Southern Tropics. *Journal of Atmospheric Chemistry*, 39(3), 261-280. doi:10.1023/a:1010692201459
- [24] Bowman, & Ben-Akiva. (2001). Activity-based disaggregate travel demand model system with activity schedules. *Transportation Research Part A: Policy and Practice, 35*(1), 1-28. doi:<u>http://dx.doi.org/10.1016/S0965-8564(99)00043-9</u>
- [25] Bradley, & Bowman. (2006). A summary of design features of activity-based microsimulation models for US MPOs. Paper presented at the Conference on Innovations in Travel Demand Modeling, Austin, TX. Retrieved from http://jbowman.net/papers/2006.Bradley_Bowman.Features_of_AB_models_US_MPOs.pdf
- [26] Bradley, Bowman, & Griesenbeck. (2010). SACSIM: An applied activity-based model system with fine-level spatial and temporal resolution. *Journal of Choice Modelling*, *3*(1), 5-31. doi:<u>http://dx.doi.org/10.1016/S1755-5345(13)70027-7</u>
- [27] Brauer, Lencar, Tamburic, Koehoorn, Demers, & Karr. (2008). A cohort study of traffic-related air pollution impacts on birth outcomes. *Environmental Health Perspectives*, *116*(5), 680-686. doi:10.1289/ehp.10952
- [28] Brugge, Durant, & Rioux. (2007). Near-highway pollutants in motor vehicle exhaust: A review of epidemiologic evidence of cardiac and pulmonary health risks. *Environmental Health*, 6:23.
- [29] Brundtland Commission. (1987). *Our Common Future*. Oslo, Norway. Retrieved from: <u>http://www.exteriores.gob.es/Portal/es/PoliticaExteriorCooperacion/Desarrollosostenible/Docume</u> <u>nts/Informe%20Brundtland%20(En%20ingl%C3%A9s).pdf</u>
- Burak, Dog`an, & Gaziog`lu. (2004). Impact of urbanization and tourism on coastal environment. Ocean & Coastal Management, 47(9–10), 515-527. doi:<u>http://dx.doi.org/10.1016/j.ocecoaman.2004.07.007</u>
- [31] Burchell, Listokin, & Galley. (2000). Smart growth: More than a ghost of urban policy past, less than a bold new horizon. *Housing Policy Debate*, *11*(4), 821-879. doi:10.1080/10511482.2000.9521390
- [32] Burke, Zufall, & Ozkaynak. (2001). A population exposure model for particulate matter: Case study results for PM_{2.5} in Philadelphia, PA. *Journal of Exposure Analysis and Environmental Epidemiology*, *11*, 470-489.
- [33] Burton, Jenks, & Williams. (2003). *The compact city: A sustainable urban form?* Routledge.
- [34] Buzzelli, & Jerrett. (2007). Geographies of susceptibility and exposure in the city: Environmental inequity of traffic-related air pollution in Toronto. *Canadian Journal of Regional Science*, *30*(2), 195-210.

- [35] Caiazzo, Ashok, Waitz, Yim, & Barrett. (2013). Air pollution and early deaths in the United States. Part I: Quantifying the impact of major sectors in 2005. *Atmospheric Environment*, 79, 198-208. doi:<u>http://dx.doi.org/10.1016/j.atmosenv.2013.05.081</u>
- [36] Camagni, Gibelli, & Rigamonti. (2002). Urban mobility and urban form: The social and environmental costs of different patterns of urban expansion. *Ecological Economics*, 40, 199-216.
- [37] Castiglione, Bradley, & Gliebe. (2015). *Activity-based travel demand models: A primer*. Washington, DC. Retrieved from: <u>http://www.trb.org/Publications/Blurbs/170963.aspx</u>
- [38] Cervero, & Gorham. (1995). Commuting in transit versus automobile neighborhoods. *Journal of American Planning Association*, *61*(2), 210-225.
- [39] Cervero, & Kockelman. (1997). Travel demand and the 3Ds: Density, diversity, and design. *Transportation Research Part D: Transport and Environment*, 2(3), 199-219. doi:<u>http://dx.doi.org/10.1016/S1361-9209(97)00009-6</u>
- [40] Chakraborty. (2009). Automobiles, air toxics, and adverse health risks: environmental inequities in Tampa Bay, Florida. *Annals of the Association of American Geographers*, *99*(4), 674-697.
- [41] Charypar, & Nagel. (2005). Generating complete all-day activity plans with genetic algorithms. *Transportation*, *32*(4), 369-397. doi:10.1007/s11116-004-8287-y
- [42] Chattopadhyay, & Taylor. (2012). Do smart growth strategies have a role in curbing vehicle miles traveled? A further assessment using household level survey data. *The B.E. Journal of Economic Analysis & Policy*, *12*(1). doi:10.1515/1935-1682.3224
- [43] Chen, Hoey, Nugent, Cook, & Yu. (2012). Sensor-based activity recognition. *IEEE Transactions on Systems, Man, and Cybernetics, Part C (Applications and Reviews), 42*(6), 790-808. doi:10.1109/TSMCC.2012.2198883
- [44] Chen, & Whalley. (2012). Green infrastructure: The effects of urban rail transit on air quality. *American Economic Journal: Economic Policy*, 4(1), 58-97.
- [45] Chu. (2003). Empirical analysis of commute stop-making behavior. *Transportation Research Record: Journal of the Transportation Research Board, 1831*, 106-113. doi:doi:10.3141/1831-12
- [46] Clark, Millet, & Marshall. (2011). Air quality and urban form in US urban areas: Evidence from regulatory monitors. *Environmental Science & Technology*, 45(16), 7028-7035. doi:10.1021/es2006786
- [47] Clark, Millet, & Marshall. (2014). National patterns in environmental injustice and inequality: Outdoor NO₂ air pollution in the United States. *PLOS ONE*, 9(4), e94431. doi:10.1371/journal.pone.0094431

- [48] Cohen. (2006). Urbanization in developing countries: Current trends, future projections, and key challenges for sustainability. *Technology in Society*, 28(1–2), 63-80. doi:<u>http://dx.doi.org/10.1016/j.techsoc.2005.10.005</u>
- [49] Coogan, White, Jerrett, Brook, Su, Seto, . . . Rosenberg. (2012). Air pollution and incidence of hypertension and diabetes in African American women living in Los Angeles. *Circulation*, *125*(6), 767-772.
- [50] Cortese, & Spengler. (1976). Ability of fixed monitoring stations to represent personal carbon monoxide exposure. *Journal of the Air Pollution Control Association*, 26(12), 1144-1150. doi:10.1080/00022470.1976.10470372
- [51] Crane. (2000). The influence of urban form on travel: An interpretative review. *Journal of Planning Literature*, *15*(1), 3-23.
- [52] Davis, Diegel, & Boundy. (2016). *Transportation energy data book*. ORNL-6991. Oak Ridge, TN. Retrieved from: <u>http://cta.ornl.gov/data/tedb34/Edition34_Full_Doc.pdf</u>
- [53] de Hartog, Boogaard, Nijland, & Hoek. (2010). Do the health benefits of cycling outweigh the risks? *Environmental Health Perspectives*, *118*(8), 1109-1116. doi:10.1289/ehp.0901747
- [54] de Nazelle, Seto, Donaire-Gonzalez, Mendez, Matamala, Nieuwenhuijsen, & Jerrett. (2013). Improving estimates of air pollution exposure through ubiquitous sensing technologies. *Environmental Pollution*, *176*, 92-99. doi:<u>http://dx.doi.org/10.1016/j.envpol.2012.12.032</u>
- [55] De Ridder, Lefebre, Adriaensen, Arnold, Beckroege, Bronner, . . . Weber. (2008a). Simulating the impact of urban sprawl on air quality and population exposure in the German Ruhr area. Part I: Reproducing the base state. *Atmospheric Environment*, 42(30), 7059-7069. doi:http://dx.doi.org/10.1016/j.atmosenv.2008.06.045
- [56] De Ridder, Lefebre, Adriaensen, Arnold, Beckroege, Bronner, . . . Weber. (2008b). Simulating the impact of urban sprawl on air quality and population exposure in the German Ruhr area. Part II: Development and evaluation of an urban growth scenario. *Atmospheric Environment*, *42*(30), 7070-7077. doi:http://dx.doi.org/10.1016/j.atmosenv.2008.06.044
- [57] Deguen, & Zmirou-Navier. (2010). Social inequalities resulting from health risks related to ambient air quality: A European review. *European Journal of Public Health*, 20(1), 27-35.
- [58] Delfino, Staimer, Tjoa, Gillen, Kleinman, Sioutas, & Cooper. (2008). Personal and ambient air pollution exposures and lung function decrements in children with asthma. *Environmental Health Perspectives*, *116*(4), 550-558. doi:10.1289/ehp.10911
- [59] Department of Health and Human Services. (2009). Annual update of the HHS poverty guidelines. *Federal Register*, 74(14), 4199-4201.

- [60] Deschenes, Greenstone, & Shapiro. (2012). Defensive investments and the demand for air quality: Evidence from the NO_x budget program and ozone reductions. *National Bureau of Economic Research Working Paper Series, No. 18267.* doi:10.3386/w18267
- [61] Dhondt, Beckx, Degraeuwe, Lefebvre, Kochan, Bellemans, . . . Putman. (2012). Health impact assessment of air pollution using a dynamic exposure profile: Implications for exposure and health impact estimates. *Environmental Impact Assessment Review*, *36*, 42-51. doi:<u>http://dx.doi.org/10.1016/j.eiar.2012.03.004</u>
- [62] Dhondt, Kochan, Beckx, Lefebvre, Pirdavani, Degraeuwe, . . . Putman. (2013). Integrated health impact assessment of travel behaviour: Model exploration and application to a fuel price increase. *Environment International*, *51*, 45-58.
- [63] Di, Wang, Zanobetti, Wang, Koutrakis, Choirat, . . . Schwartz. (2017). Air pollution and mortality in the medicare population. *New England Journal of Medicine*, 376(26), 2513-2522. doi:10.1056/NEJMoa1702747
- [64] Dieleman, Dijst, & Spit. (1999). Planning the compact city: The randstad Holland experience. *European Planning Studies*, 7(5), 605-621. doi:10.1080/09654319908720541
- [65] Dockery, Pope, Xu, Spengler, Ware, Fay, ... Speizer (1993). An association between air pollution and mortality in six US cities. *New England Journal of Medicine*, *329*(24), 1753-1759. doi:doi:10.1056/NEJM199312093292401
- [66] Dockery, & Spengler. (1981). Personal exposure to respirable particulates and sulfates. *Journal of the Air Pollution Control Association*, *31*(2), 153-159. doi:10.1080/00022470.1981.10465205
- [67] Dong, Block, & Mandel. (2004). Activities contributing to total energy expenditure in the United States: Results from the NHAPS study. *International Journal of Behavioral Nutrition and Physical Activity*, *1*(1), 4.
- [68] Dons, Beckx, Arentze, Wets, & Panis. (2011a). Using an activity-based framework to determine effects of a policy measure on population exposure to nitrogen dioxide. *Transportation Research Record: Journal of the Transportation Research Board, 2233,* 72-79. doi:doi:10.3141/2233-09
- [69] Dons, Int Panis, Van Poppel, Theunis, & Wets. (2012). Personal exposure to black carbon in transport microenvironments. *Atmospheric Environment*, 55, 392-398. doi:<u>http://dx.doi.org/10.1016/j.atmosenv.2012.03.020</u>
- [70] Dons, Int Panis, Van Poppel, Theunis, Willems, Torfs, & Wets. (2011b). Impact of time-activity patterns on personal exposure to black carbon. *Atmospheric Environment*, *45*(21), 3594-3602.
- [71] Dons, Van Poppel, Kochan, Wets, & Int Panis. (2014). Implementation and validation of a modeling framework to assess personal exposure to black carbon. *Environment International*, 62(2014), 64-71.

- [72] Downs. (2005). Smart growth: Why we discuss it more than we do it. *Journal of the American Planning Association*, *71*(4), 367-378. doi:10.1080/01944360508976707
- [73] Echenique, Hargreaves, Mitchell, & Namdeo. (2012). Growing cities sustainably. *Journal of the American Planning Association*, 78(2), 121-137. doi:10.1080/01944363.2012.666731
- [74] El-Nasser, & Overberg. (2001, 22 February). A comprehensive look at sprawl in America. USA *Today*, pp. 1A-9A.
- [75] Evans, & Stuart. (2011). A passive sampling study of small-scale variations in ambient acetaldehyde and formaldehyde concentrations. *Air, Soil and Water Research, 2011*(4), 71-79.
- [76] Ewing, & Cervero. (2001). Travel and the built environment: A synthesis. *Transportation Research Record: Journal of the Transportation Research Board*, 1780, 87-114. doi:10.3141/1780-10
- [77] Ewing, & Cervero. (2010). Travel and the built environment. *Journal of the American Planning Association*, *76*(3), 265-294. doi:10.1080/01944361003766766
- [78] Ewing, Pendall, & Chen. (2002). *Measuring sprawl and its impact*. Washington, DC. Retrieved from: <u>http://www.smartgrowthamerica.org/documents/MeasuringSprawl.PDF</u>
- [79] Farber, Maoh, & Kanaroglou. (2009). The impacts of urban growth policies on transportation system usage and performance: A simulation approach. *Canadian Journal of Transportation*, 3(1), 1-25.
- [80] Faustini, Rapp, & Forastiere. (2014). Nitrogen dioxide and mortality: Review and meta-analysis of long-term studies. *Eur Respir J*, 44(3), 744-753. doi:10.1183/09031936.00114713
- [81] Federal Highway Administration. (2009). 2009 national household travel survey. Retrieved from <u>http://nhts.ornl.gov</u>. Accessed on November 10, 2014.
- [82] Florida Department of Transportation. (2017). Tampa Bay Next The future of transportation. Retrieved from <u>http://www.tampabaynext.com/</u>. Accessed on October 31, 2017.
- [83] Forastiere, Stafoggia, Tasco, Picciotto, Agabiti, Cesaroni, & Perucci. (2007). Socioeconomic status, particulate air pollution, and daily mortality: Differential exposure or differential susceptibility. *American Journal of Industrial Medicine*, *50*(3), 208-216.
- [84] Frank, & Engelke. (2001). The built environment and human activity patterns: Exploring the impacts of urban form on public health. *Journal of Planning Literature*, *16*(2), 202-218.
- [85] Frank, & Engelke. (2005). Multiple impacts of the built environment on public health: Walkable places and the exposure to air pollution. *International Regional Science Review*, 28(2), 193-216.

- [86] Frank, Kavage, & Litman. (2006a). Promoting public health through smart growth: Building healthier communities through transportation and land use policies and practices. Vancouver, British Columbia. Retrieved from: http://www.vtpi.org/sgbc_health.pdf
- [87] Frank, Sallis, Conway, Chapman, Saelens, & Bachman. (2006b). Many pathways from land use to health: Associations between neighborhood walkability and active transportation, body mass index, and air quality. *Journal of the American Planning Association*, 72(1), 75-87. doi:10.1080/01944360608976725
- [88] Fridh, & Stuart. (2014). Spatial variation in ambient benzene concentration over a city park. *Journal of Environmental Health*, 76(6), 86-91.
- [89] Frumkin. (2002). Urban sprawl and public health. *Public Health Reports*, 117(3), 201-217.
- [90] Fung, Khan, Krewski, & Chen. (2006). Association between air pollution and multiple respiratory hospitalizations among the elderly in Vancouver, Canada. *Inhalation Toxicology*, *18*(13), 1005-1011.
- [91] Galea, & Vlahov. (2005). Urban health: Evidence, challenges, and directions. *Annu Rev Public Health*, *26*, 341-365. doi:10.1146/annurev.publhealth.26.021304.144708
- [92] Gan, Koehoorn, Davies, Demers, Tamburic, & Brauer. (2011). Long-term exposure to trafficrelated air pollution and the risk of coronary heart disease hospitalization and mortality. *Environmental Health Perspectives*, *119*(4), 501-507. doi:10.1289/ehp.1002511
- [93] Gariazzo, Hänninen, Amicarelli, Pelliccioni, Silibello, Sozzi, & Jantunen. (2011). Integrated model for the estimation of annual, seasonal, and episode PM₁₀ exposures of children in Rome, Italy. *Air Quality, Atmosphere & Health, 4*(3-4), 169-178. doi:10.1007/s11869-010-0128-x
- [94] Gauderman, Mcconnell, Gilliland, London, Thomas, Avol, . . . Peters. (2000). Association between air pollution and lung function growth in Southern California children. *Am J Respir Crit Care Med.*, *162*, 1383-1390.
- [95] Gertler. (2005). Diesel vs. gasoline emissions: Does PM from diesel or gasoline vehicles dominate in the US? *Atmospheric Environment*, 39(13), 2349-2355. doi:<u>http://dx.doi.org/10.1016/j.atmosenv.2004.05.065</u>
- [96] Giles-Corti, Vernez-Moudon, Reis, Turrell, Dannenberg, Badland, . . . Owen. (2016). City planning and population health: A global challenge. *The Lancet*, 388(10062), 2912-2924. doi:<u>http://dx.doi.org/10.1016/S0140-6736(16)30066-6</u>
- [97] Gim. (2012). A meta-analysis of the relationship between density and travel behavior. *Transportation*, *39*(3), 491-519. doi:10.1007/s11116-011-9373-6
- [98] Giuliano. (1995). The weakening transportation-land use connection. *Access Magazine*, 1(6), 3-11.

- [99] Gliebe, Bradley, Ferdous, Outwater, Lin, & Chen. (2014). *Transferability of activity-based model parameters*. 0309273811. Washington, DC. Retrieved from: <u>http://nap.edu/22384</u>
- [100] Good, Molter, Ackerson, Bachand, Carpenter, Clark, . . . Volckens. (2016). The Fort Collins commuter study: Impact of route type and transport mode on personal exposure to multiple air pollutants. *J Expos Sci Environ Epidemiol*, 26(4), 397-404. doi:10.1038/jes.2015.68
- [101] Google (Cartographer). (2017). Hillsborough County, Florida. Retrieved from https://www.google.com/maps/place/Hillsborough+County,+FL/@27.8716814,-82.7190379,10z/data=!3m1!4b1!4m5!3m4!1s0x88dd3512570a8e95:0xdb3a212adf1af30c!8m2!3 d27.9903597!4d-82.3017728
- [102] Gordon, & Richardson. (1997). Are compact cities a desirable planning goal? *Journal of the American Planning Association*, 63(1), 95-106. doi:10.1080/01944369708975727
- [103] Grabow, Spak, Holloway, Stone Jr, Mednick, & Patz. (2012). Air quality and exercise-related health benefits from reduced car travel in the midwestern United States. *Environmental health perspectives*, *120*(1), 68-76.
- [104] Green, Smorodinsky, Kim, McLaughlin, & Ostro. (2004). Proximity of California public schools to busy roads. *Environmental Health Perspectives*, *112*(1), 61-66.
- [105] Grimmond, & Oke. (1999). Aerodynamic properties of urban areas derived from analysis of surface form. *Journal of Applied Meteorology*, *38*(9), 1262-1292.
- [106] Gurram, Stuart, & Pinjari. (2015). Impacts of travel activity and urbanicity on exposures to ambient oxides of nitrogen and on exposure disparities. *Air Quality, Atmosphere & Health*, 8(1), 97-114. doi:10.1007/s11869-014-0275-6
- [107] Hajat, Hsia, & O'Neill. (2015). Socioeconomic disparities and air pollution exposure: A global review. *Current Environmental Health Reports*, 2(4), 440-450. doi:10.1007/s40572-015-0069-5
- [108] Handy. (1996). Methodologies for exploring the link between urban form and travel behavior. *Transportation Research Part D: Transport and Environment, 1*(2), 151-165.
- [109] Handy. (2005). Smart growth and the transportation-land use connection: What does the research tell us? *International Regional Science Review*, 28(2), 146-167.
- [110] Hankey, & Marshall. (2010). Impacts of urban form on future US passenger-vehicle greenhouse gas emissions. *Energy Policy*, 38(9), 4880-4887. doi:<u>http://dx.doi.org/10.1016/j.enpol.2009.07.005</u>
- [111] Hankey, Marshall, & Brauer. (2012). Health impacts of the built environment: Within-urban variability in physical inactivity, air pollution, and ischemic heart disease mortality. *Environmental Health Perspectives*, *120*(2), 247-253.

- [112] Hao, Hatzopoulou, & Miller. (2010). Integrating an activity-based travel demand model with dynamic traffic assignment and emission models: Implementation in the Greater Toronto, Canada, Area. *Transportation Research Record*, 2176, 1-13.
- [113] Harford. (2006). Congestion, pollution, and benefit-to-cost ratios of US public transit systems. *Transportation Research Part D*, *11*(1), 45-58.
- [114] Hatzopoulou, Hao, & Miller. (2011). Simulating the impacts of household travel on greenhouse gas emissions, urban air quality, and population exposure. *Transportation*, 38(6), 871-887. doi:10.1007/s11116-011-9362-9
- [115] Hatzopoulou, & Miller. (2010). Linking an activity-based travel demand model with traffic emission and dispersion models: Transport's contribution to air pollution in Toronto. *Transportation Research Part D: Transport and Environment, 15*(6), 315-325.
- [116] Hayes. (2005). *Statistical methods for communication science*. Mahwah, NJ: Lawrence Erlbaum Associates.
- [117] Health Canada. (2010). Canadian human activity pattern survey (CHAPS). Retrieved from http://www.hc-sc.gc.ca/ewh-semt/air/chaps2-epahc2-eng.php. Accessed on July 10, 2011.
- [118] Healthy People 2020, & US Department of Health and Human Services. (2010). Foundation health measures: Disparities. Retrieved from <u>http://www.healthypeople.gov/2020/about/disparitiesAbout.aspx</u>. Accessed on December 1, 2013.
- [119] HEI Panel on the Health Effects of Traffic-Related Air Pollution. (2010). Traffic-related air pollution: A critical review of the literature on emissions, exposure, and health effects. Boston, MA. Retrieved from: <u>http://pubs.healtheffects.org/getfile.php?u=553</u>
- [120] Hillsborough Metropolitan Planning Organization. (2014). Socioeconomic data forecasting and scenario planning for 2040. Tampa, FL. Retrieved from: <u>http://www.planhillsborough.org/wpcontent/uploads/2014/07/SE-Data-Tech-Memo-2014-7-9-Final.pdf</u>
- [121] Hixson, Mahmud, Hu, Bai, Niemeier, Handy, . . . Kleeman. (2009). Influence of regional development policies and clean technology adoption on future air pollution exposure. *Atmospheric Environment*, 44(4), 552-562.
- [122] Hoek, Brunekreef, Goldbohm, Fischer, & van den Brandt. (2002). Association between mortality and indicators of traffic-related air pollution in the Netherlands: A cohort study. *The Lancet*, *360*(9341), 1203-1209. doi:<u>http://dx.doi.org/10.1016/S0140-6736(02)11280-3</u>
- [123] Horni, Nagel, & Axhausen. (2016). *The multi-agent transport simulation MATSim*. London, UK: Ubiquity Press.

- [124] Houston, Wu, Ong, & Winer. (2004). Structural disparities of urban traffic in Southern California: Implications for vehicle-related air pollution exposure in minority and high-poverty neighborhoods. *Journal of Urban Affairs*, 26(5), 565-592. doi:10.1111/j.0735-2166.2004.00215.x
- [125] Hu. (2011). Globalization of diabetes: The role of diet, lifestyle, and genes. *Diabetes Care, 34*(6), 1249-1257. doi:10.2337/dc11-0442
- [126] Huang, & Batterman. (2000). Residence location as a measure of environmental exposure: A review of air pollution epidemiology studies. *J Expo Anal Environ Epidemiol, 10*(1), 66-85.
- [127] Huang, Lu, & Sellers. (2007). A global comparative analysis of urban form: Applying spatial metrics and remote sensing. *Landscape and Urban Planning*, 82(4), 184-197. doi:<u>http://dx.doi.org/10.1016/j.landurbplan.2007.02.010</u>
- [128] Hystad, Villeneuve, Goldberg, Crouse, & Johnson. (2015). Exposure to traffic-related air pollution and the risk of developing breast cancer among women in eight Canadian provinces: A case-control study. *Environment International*, 74, 240-248. doi:https://doi.org/10.1016/j.envint.2014.09.004
- [129] International Agency for Research on Cancer. (2016). *Outdoor air pollution*. Lyon, France. Retrieved from: <u>http://monographs.iarc.fr/ENG/Monographs/vol109/mono109.pdf</u>
- [130] Jabareen. (2006). Sustainable urban forms: Their typologies, models and, concepts. *Journal of Planning Education and Research*, 26(1), 38-52.
- [131] Jackson. (2003). The relationship of urban design to human health and condition. *Landscape and Urban Planning*, *64*(4), 191-200.
- [132] Jerrett, Arain, Kanaroglou, Beckerman, Potoglou, Sahsuvaroglu, . . . Giovis. (2005). A review and evaluation of intraurban air pollution exposure models. *Journal of Exposure Analysis and Environmental Epidemiology*, *15*, 185-204.
- Johansson, Burman, & Forsberg. (2009). The effects of congestions tax on air quality and health. *Atmospheric Environment*, 43(31), 4843-4854. doi:<u>http://dx.doi.org/10.1016/j.atmosenv.2008.09.015</u>
- [134] Jou, & Mahmassani. (1997). Comparative analysis of day-to-day trip-chaining behavior of urban commuters in two cities. *Transportation Research Record: Journal of the Transportation Research Board, 1607*(1), 163-170. doi:10.3141/1607-22
- [135] Kalnay, & Cai. (2003). Impact of urbanization and land-use change on climate. *Nature*, 423(6939), 528-531.
 doi:<u>http://www.nature.com/nature/journal/v423/n6939/suppinfo/nature01675_S1.html</u>

- [136] Kenworthy, & Laube. (1999). Patterns of automobile dependence in cities: An international overview of key physical and economic dimensions with some implications for urban policy. *Transportation Research Part A: Policy and Practice, 33*(7-8), 691-723.
- [137] Kim, Park, & Sang. (2008). *Spatial and temporal analysis of urban traffic volume*. Paper presented at the 2008 ESRI International User Conference, San Diego, CA Retrieved from http://proceedings.esri.com/library/userconf/proc08/papers/pape_1613.pdf
- [138] Kim, Sass-Kortsak, Purdham, Dales, & Brook. (2006). Associations between personal exposures and fixed-site ambient measurements of fine particulate matter, nitrogen dioxide, and carbon monoxide in Toronto, Canada. *Journal of Exposure Science and Environmental Epidemiology*, *16*(2), 172-183.
- [139] Kim, Smorodinsky, Lipsett, Singer, Hodgson, & Ostro. (2004). Traffic-related air pollution near busy roads: The East Bay children's respiratory health study. Am J Respir Crit Care Med, 170(5), 520-526. doi:10.1164/rccm.200403-281OC
- [140] Kimura, Aoki, Kitahara, & Aiyoshizawa. (2001). Ultra-clean combustion technology combining a low-temperature and premixed combustion concept for meeting future emission standards. Detroit, MI. Retrieved from: <u>https://doi.org/10.4271/2001-01-0200</u>
- [141] Kitamura, Fujii, & Pas. (1997). Time-use data, analysis and modeling: Toward the next generation of transportation planning methodologies. *Transport Policy*, 4(4), 225-235. doi:<u>http://dx.doi.org/10.1016/S0967-070X(97)00018-8</u>
- [142] Klepeis, Nelson, Ott, Robinson, Tsang, Switzer, . . . Engelmann. (2001). The national human activity pattern survey (NHAPS): A resource for assessing exposure to environmental pollutants. *Journal of Exposure Analysis and Environmental Epidemiology*, *11*(3), 231-252.
- [143] Koppelman, & Bhat. (2006). A self instructing course in mode choice modeling: Multinomial and nested logit models. Retrieved from <u>http://www.caee.utexas.edu/prof/bhat/COURSES/LM_Draft_060131Final-060630.pdf</u>. Accessed on November 14, 2017.
- [144] Kornartit, Sokhi, Burton, & Khaiwal. (2010). Activity pattern and personal exposure to nitrogen dioxide in indoor and outdoor microenvironments. *Environment International*, *36*(1), 36-45.
- [145] Koupal, Cumberworth, Michaels, Beardsley, & Brzezinski. (2003). Design and implementation of MOVES: EPA's new generation mobile source emission model. Ann Arbor, MI. Retrieved from: <u>http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.118.3111&rep=rep1&type=pdf</u>
- [146] Kousa, Monn, Rotko, Alm, Oglesby, & Jantunen. (2001). Personal exposures to NO₂ in the EXPOLIS-study: Relation to residential indoor, outdoor and workplace concentrations in Basel, Helsinki, and Prague. *Atmospheric Environment*, 35(20), 3405-3412.

- [147] Krzyzanowski. (1997). Methods for assessing the extent of exposure and effects of air pollution. *Occup Environ Med*, 54(3), 145-151.
- [148] Lai, Kendall, Ferrier, Lindup, Alm, Hanninen, . . . Nieuwenhuijsen. (2004). Personal exposures and microenvironment concentrations of PM_{2.5}, VOC, NO₂, and CO in Oxford, UK. Atmospheric Environment, 38(37), 6399-6410.
- [149] Leech, Nelson, Burnett, Aaron, & Raizenne. (2002). It's about time: A comparison of Canadian and American time-activity patterns. *Journal of Exposure Analysis and Environmental Epidemiology*, *12*(6), 427-432.
- [150] Linder, Marko, & Sexton. (2008). Cumulative cancer risk from air pollution in Houston: Disparities in risk burden and social disadvantage. *Environ Sci Technol*, 42(12), 4312-4322.
- [151] Lode, Schönberger, & Toussaint. (2016). Clean air for all by 2030? Air quality in the 2030 agenda and in international law. *Review of European, Comparative & International Environmental Law*, 25(1), 27-38. doi:10.1111/reel.12151
- [152] Lopez, & Hynes. (2003). Sprawl in the 1990s: Measurement, distribution, and trends. Urban Aff Rev Thousand Oaks Calif, 38(3), 325-355. doi:10.1177/1078087402238805
- [153] Lyons, Kenworthy, Moy, & dos Santos. (2003). An international urban air pollution model for the transportation sector. *Transportation Research Part D: Transport and Environment*, 8(3), 159-167. doi:<u>http://dx.doi.org/10.1016/S1361-9209(02)00047-0</u>
- [154] Makido, Dhakal, & Yamagata. (2012). Relationship between urban form and CO₂ emissions: Evidence from fifty Japanese cities. *Urban Climate*, 2, 55-67. doi:http://dx.doi.org/10.1016/j.uclim.2012.10.006
- [155] Malmqvist, Jakobsson, Tinnerberg, Rignell-Hydbom, & Rylander. (2013). Gestational diabetes and preeclampsia in association with air pollution at levels below current air quality guidelines. *Environmental Health Perspectives*, *121*(4), 488-493. doi:10.1289/ehp.1205736
- [156] Marshall. (2008). Environmental inequality: Air pollution exposures in California's South Coast air basin. *Atmospheric Environment*, 42(21), 5499-5503.
- [157] Marshall, Brauer, & Frank. (2009). Healthy neighborhoods: Walkability and air pollution. *Environmental Health Perspectives*, *117*(11), 1752-1759. doi:10.1289/ehp.0900595
- [158] Marshall, Granvold, Hoats, McKone, Deakin, & Nazaroff. (2006). Inhalation intake of ambient air pollution in California's South Coast air basin. *Atmospheric Environment*, 40(23), 4381-4392.
- [159] McGarigal, Cushman, Neel, & Ene. (2002). FRAGSTATS: Spatial pattern analysis program for categorical maps. Retrieved from <u>http://www.umass.edu/landeco/research/fragstats/fragstats.html</u>. Accessed on May 20, 2017.

- [160] Meng, Wilhelm, Rull, English, & Ritz. (2007). Traffic and outdoor air pollution levels near residences and poorly controlled asthma in adults. *Annals of Allergy, Asthma & Immunology*, 98(5), 455-463.
- [161] Miao, Chen, Buzzelli, & Aronson. (2015). Environmental equity research: Review with focus on outdoor air pollution research methods and analytic tools. Archives of Environmental & Occupational Health, 70(1), 47-55. doi:10.1080/19338244.2014.904266
- [162] Miller, & Roorda. (2003). Prototype model of household activity-travel scheduling. *Transportation Research Record: Journal of the Transportation Research Board*, 1831, 114-121. doi:10.3141/1831-13
- [163] Mitchell, & Dorling. (2003). An environmental justice analysis of British air quality. *Environment and Planning A*, *35*(5), 909-929. doi:doi:10.1068/a35240
- [164] Monn. (2001). Exposure assessment of air pollutants: A review on spatial heterogeneity and indoor/outdoor/personal exposure to suspended particulate matter, nitrogen dioxide and ozone. *Atmospheric Environment*, 35(1), 1-32. doi:<u>http://dx.doi.org/10.1016/S1352-2310(00)00330-7</u>
- [165] Moore, Gould, & Keary. (2003). Global urbanization and impact on health. International Journal of Hygiene and Environmental Health, 206(4), 269-278. doi:<u>https://doi.org/10.1078/1438-4639-00223</u>
- [166] Morgenstern, Zutavern, Cyrys, Brockow, Koletzko, Krämer, . . . Heinrich. (2008). Atopic diseases, allergic sensitization, and exposure to traffic-related air pollution in children. 177(12), 1331-1337. doi:10.1164/rccm.200701-036OC
- [167] Muller, & Mendelsohn. (2007). Measuring the damages of air pollution in the United States. *Journal of Environmental Economics and Management*, 54(1), 1-14. doi:<u>http://dx.doi.org/10.1016/j.jeem.2006.12.002</u>
- [168] Nagel, Kickhofer, Horni, & Charypar. (2016). A closer look at scoring. In *The multi-agent transport simulation: MATSim.* Horni, Nagel, & Axhausen (Eds.). London: Ubiquity Press.
- [169] National Center for Environmental Assessment, Office of Research and Development, & US Environmental Protection Agency. (2011). *Exposure factors handbook: 2011 edition*. (EPA/600/R-09/052F). Washington, DC Retrieved from <u>http://www.epa.gov/ncea/efh/pdfs/efhcomplete.pdf</u>.
- [170] National Exposure Research Laboratory, National Center for Environmental Assessment, Office of Research and Development, & US Environmental Protection Agency. (2000). Strategic plan for the analysis of the national human exposure assessment survey (NHEXAS) pilot study data. Washington, DC. Retrieved from: <u>https://cfpub.epa.gov/si/si_public_record_report.cfm?dirEntryId=82947</u>

- [171] National Research Council, Committee for the Study on the Relationships Among Development Patterns Vehicle Miles Traveled and Energy Consumption, & Transportation Research Board. (2009). Driving and the built environment: The effects of compact development on motorized travel, energy use, and CO₂ emissions. Washington, DC. Retrieved from: <u>http://onlinepubs.trb.org/Onlinepubs/sr/sr298.pdf</u>
- [172] NAVTEQ. (2010). NAVSTREETS street data. Retrieved from <u>http://www.navteq.com/</u>. Accessed on July 10, 2011.
- [173] Neuberger, Rabczenko, & Moshammer. (2007). Extended effects of air pollution on cardiopulmonary mortality in Vienna. *Atmospheric Environment*, *41*(38), 8549-8556.
- [174] Neuman. (2005). The compact city fallacy. *Journal of Planning Education and Research*, 25(1), 11-26. doi:doi:10.1177/0739456X04270466
- [175] O'Neill, Jerrett, Kawachi, Levy, Cohen, Gouveia, . . . Schwartz. (2003). Health, wealth, and air pollution: Advancing theory and methods. *Environ Health Perspect*, *111*(16), 1861-1870.
- [176] Office of Sustainable Communities, Office of Environmental Justice, & US Environmental Protection Agency. (2013). Creating equitable, healthy, and sustainable communities: strategies for advancing smart growth, environmental justice, and equitable development. (EPA 231-K-10-005). Retrieved from <u>http://www.epa.gov/dced/pdf/equitable-dev/equitable-development-report-508-011713b.pdf</u>.
- [177] Oglesby, Kunzli, Roosli, Braun-Fahrlander, Mathys, Stern, . . . Kousa. (2000). Validity of ambient levels of fine particles as surrogate for personal exposure to outdoor air pollution -Results of the European EXPOLIS-EAS Study (Swiss Center Basel). *Journal of the Air & Waste Management Association*, 50(7), 1251-1261.
- [178] Ott. (1982). Concepts of human exposure to air pollution. *Environment International*, 7(3), 179-196.
- [179] Ott, Wallace, Mage, Akland, Lewis, Sauls, . . . Morehouse. (1986). The environmental protection agency's research program on total human exposure. *Environment International*, 12(1-4), 475-494.
- [180] Özkaynak. (1986). Bias due to misclassification of personal exposures in epidemiologic studies of indoor and outdoor air pollution. *Environment International*, 12(1-4), 389-393.
- [181] Pearce, Kingham, & Zawar-Reza. (2006). Every breath you take? Environmental justice and air pollution in Christchurch, New Zealand. *Environment and Planning A*, 38(5), 919-938. doi:doi:10.1068/a37446
- [182] Pekkanen, & Pearce. (2001). Environmental epidemiology: Challenges and opportunities. *Environmental Health Perspectives*, 109(1), 1-5.

- [183] Perlin, Wong, & Sexton. (2001). Residential proximity to industrial sources of air pollution: Interrelationships among race, poverty, and age. *Journal of the Air & Waste Management Association*, 51(3), 406-421.
- [184] Pinjari, & Bhat. (2011). Activity-based travel demand analysis. In A Handbook of Transport Economics. de Palma, Lindsey, Quinet, & Vickerman (Eds.).(Vol. 10, pp. 213-248). Northampton, MA: Edward Elgar Pub.
- [185] Pinjari, Eluru, Copperman, Sener, Guo, Srinivasan, & Bhat. (2006). *Activity-based travel-demand* analysis for metropolitan areas in Texas: CEMDAP models, framework, software architecture and application results. 0-4080-8. <u>http://www.caee.utexas.edu/prof/bhat/REPORTS/4080_8.pdf</u>
- [186] Pinjari, Eluru, Srinivasan, Guo, Copperman, Sener, & Bhat. (2008). CEMDAP: Modeling and microsimulation frameworks, software development, and verification. Paper presented at the Transportation Research Board 87th Annual Meeting, Washington, DC. Retrieved from http://www.caee.utexas.edu/prof/bhat/ABSTRACTS/CEMDAP_paper_LM_23July07.pdf
- [187] Poor. (2008). Nitrogen emission/deposition ratios for air pollution sources that contribute to the nitrogen loading of Tampa Bay: Mid-term progress report. Retrieved from <u>http://www.health.usf.edu/NR/rdonlyres/2A5D4D52-2DBB-4CB6-B05F-5E44715A76B0/0/MidProgReprtSept2008.pdf</u>. Accessed on October 10, 2012.
- [188] Pope III, Burnett, Thun, Calle, Krewski, Ito, & Thurston. (2002). Lung cancer, cardiopulmonary mortality, and long-term exposure to fine particulate air pollution. *The Journal of American Medical Association*, 287(9), 1132-1141.
- [189] Popkin. (1999). Urbanization, lifestyle changes and the nutrition transition. *World Development*, 27(11), 1905-1916. doi:<u>https://doi.org/10.1016/S0305-750X(99)00094-7</u>
- [190] Pulido. (2000). Rethinking environmental racism: White privilege and urban development in Southern California. *Annals of the Association of American Geographers*, 90(1), 12-40.
- [191] Raaschou-Nielsen, Skov, Lohse, Thomsen, & Olsen. (1997). Front-door concentrations and personal exposures of Danish children to nitrogen dioxide. *Environmental Health Perspectives*, *105*(9), 964-970.
- [192] Rajamani, Bhat, Handy, Knaap, & Song. (2003). Assessing impact of urban form measures on nonwork trip mode choice after controlling for demographic and level-of-service effects. *Transportation Research Record: Journal of the Transportation Research Board*, 1831, 158-165. doi:10.3141/1831-18
- [193] Ramaswami, Milford, & Small. (2005). *Integrated environmental modeling: Pollutant transport, fate, and risk in the environment*. Hoboken, NJ: John Wiley & Sons, Inc.
- [194] Raney, & Nagel. (2006). An improved framework for large-scale multi-agent simulations of travel behaviour. In *Towards Better Performing Transport Networks*. Jourquin, Rietveld, & Westin (Eds.).(pp. 305-347). New York, NY: Routledge.
- [195] Rijnders, Janssen, van Vliet, & Brunekreef. (2001). Personal and outdoor nitrogen dioxide concentrations in relation to degree of urbanization and traffic density. *Environmental Health Perspectives*, *109*(suppl 3), 411-417.
- [196] Romieu, Ramírez-Aguilar, Moreno-Macias, Barraza-Villarreal, Miller, Hernández-Cadena, . . . Hernandez-Avila. (2004). Infant mortality and air pollution: Modifying effect by social class. *Journal of Occupational & Environmental Medicine*, 46(12), 1210-1216.
- [197] Rotko, Kousa, Alm, & Jantunen. (2001). Exposures to nitrogen dioxide in EXPOLIS-Helsinki: Microenvironment, behavioral and sociodemographic factors. *Journal of Exposure Analysis and Environmental Epidemiology*, 11(3), 216-223.
- [198] Samet, & White. (2004). Urban air pollution, health, and equity. *J Epidemiol Community Health*, 58, 3-5.
- [199] Santos, McGuckin, Nakamoto, Gray, & Liss. (2011). Summary of travel trends: 2009 national household travel survey. Washington, DC. Retrieved from: <u>http://nhts.ornl.gov/2009/pub/stt.pdf</u>
- [200] Sarnat, Klein, Sarnat, Flanders, Waller, Mulholland, . . . Tolbert. (2010). An examination of exposure measurement error from air pollutant spatial variability in time-series studies. *Journal of exposure science & environmental epidemiology*, 20(2), 135-146. doi:10.1038/jes.2009.10
- [201] Schrank, Eisele, & Lomax. (2012). *TTI's 2012 urban mobility report*. College Station, TX. Retrieved from: <u>https://static.tti.tamu.edu/tti.tamu.edu/documents/ums/archive/mobility-report-2012.pdf</u>
- [202] Schultz, Hallberg, Bellander, Bergström, Bottai, Chiesa, . . . Melén. (2015). Early-life exposure to traffic-related air pollution and lung function in adolescence. *American Journal of Respiratory and Critical Care Medicine, 193*(2), 171-177. doi:10.1164/rccm.201505-0928OC
- [203] Schwartz, Dockery, Neas, Wypij, Ware, Spengler, . . . Ferris Jr. (1994). Acute effects of summer air pollution on respiratory symptom reporting in children. *American Journal of Respiratory and Critical Care Medicine*, 150(5), 1234-1242.
- [204] Schweitzer, & Zhou. (2010). Neighborhood air quality, respiratory health, and vulnerable populations in compact and sprawled regions. *Journal of the American Planning Association*, 76(3), 363-371. doi:10.1080/01944363.2010.486623
- [205] Seinfeld, & Pandis. (1998). *Atmospheric chemistry and physics*. Hoboken, NJ: John Wiley & Sons, Inc.

- [206] Seto, Sánchez-Rodríguez, & Fragkias. (2010). The new geography of contemporary urbanization and the environment. Annual Review of Environment and Resources, 35(1), 167-194. doi:<u>http://10.1146/annurev-environ-100809-125336</u>
- [207] Setton, Marshall, Brauer, Lundquist, Hystad, Keller, & Cloutier-Fisher. (2011). The impact of daily mobility on exposure to traffic-related air pollution and health effect estimates. *Journal of Exposure Science and Environmental Epidemiology*, 21(1), 42-48.
- [208] Shekarrizfard, Faghih-Imani, Tétreault, Yasmin, Reynaud, Morency, . . . Hatzopoulou. (2017). Regional assessment of exposure to traffic-related air pollution: Impacts of individual mobility and transit investment scenarios. *Sustainable Cities and Society*, 29, 68-76. doi:<u>http://dx.doi.org/10.1016/j.scs.2016.12.002</u>
- [209] Sheppard, Burnett, Szpiro, Kim, Jerrett, Pope III, & Brunekreef. (2012). Confounding and exposure measurement error in air pollution epidemiology. *Air Quality, Atmosphere & Health,* 5(2), 203-216.
- [210] Smart Growth America. (2014). *Measuring sprawl 2014*. Washington, DC. Retrieved from: https://www.smartgrowthamerica.org/app/legacy/documents/measuring-sprawl-2014.pdf
- [211] Snyder, Venkatram, Heist, Perry, Petersen, & Isakov. (2013). RLINE: A line source dispersion model for near-surface releases. *Atmospheric Environment*, 77, 748-756. doi:<u>http://dx.doi.org/10.1016/j.atmosenv.2013.05.074</u>
- [212] Soubbotina. (2004). *Beyond economic growth: An introduction to sustainable development* Retrieved from <u>http://www.worldbank.org/depweb/english/beyond/beyondco/beg_all.pdf</u>
- [213] South Coast Air Quality Management District (AQMD). (2005). Guidance document for addressing air quality issues in general plans and local planning: a reference for local governments within the South Coast air quality management district. Retrieved from http://www.aqmd.gov/prdas/aqguide/doc/aq_guidance.pdf. Accessed on December 1, 2013.
- [214] Stevenson, Thompson, de Sa, Ewing, Mohan, McClure, . . . Woodcock. (2016). Land use, transport, and population health: Estimating the health benefits of compact cities. *Lancet*, 388(10062), 2925-2935. doi:10.1016/s0140-6736(16)30067-8
- [215] Stone. (2008). Urban sprawl and air quality in large US cities. *Journal of Environmental Management*, *86*(4), 688-698.
- [216] Stone, Mednick, Holloway, & Spak. (2007). Is compact growth good for air quality? *Journal of the American Planning Association*, 73(4), 404-418.
- [217] Stone, Mednick, Holloway, & Spak. (2009). Mobile source CO₂ mitigation through smart growth development and vehicle fleet hybridization. *Environmental Science & Technology*, *43*(6), 1704-1710.

- [218] Stuart, Mudhasakul, & Sriwatanapongse. (2009). The social distribution of neighborhood-scale air pollution and monitoring protection. *Journal of the Air & Waste Management Association*, 59(5), 591-602.
- [219] Stuart, & Zeager. (2011). An inequality study of ambient nitrogen dioxide and traffic levels near elementary schools in the Tampa area. *Journal of Environmental Management*, 92(8), 1923-1930.
- [220] Sunyer, Esnaola, Alvarez-Pedrerol, Forns, Rivas, López-Vicente, . . . Querol. (2015). Association between traffic-related air pollution in schools and cognitive development in primary school children: A prospective cohort study. *PLOS Medicine*, 12(3), e1001792. doi:10.1371/journal.pmed.1001792
- [221] Sutton. (2003). A scale-adjusted measure of "urban sprawl" using nighttime satellite imagery. *Remote Sensing of Environment*, 86(3), 353-369. doi:<u>http://dx.doi.org/10.1016/S0034-4257(03)00078-6</u>
- [222] Tampa Bay Area Regional transportation Authority. (2015). *Regional transportation master plan:* 2040 regional long range transportation plan. Tampa, FL. Retrieved from: http://tbarta.com/files/TM_2040RegionalTransportationPlan.pdf
- [223] Thomas, Stram, & Dwyer. (1993). Exposure measurement error: Influence on exposure-disease relationships and methods of correction. *Annual Review of Public Health*, 14(1), 69-93.
- [224] UN General Assembly. (2015). Transforming our world: The 2030 agenda for sustainable development. A/RES/70/1. New York, US. Retrieved from: <u>http://www.un.org/ga/search/view_doc.asp?symbol=A/RES/70/1&Lang=E</u>
- [225] UN General Assembly. (2016). New York declaration for refugees and migrants. A/RES/71/1. New York, US. Retrieved from: <u>http://www.un.org/en/ga/search/view_doc.asp?symbol=A/RES/71/1</u>
- [226] United Nations, Department of Economic and Social Affairs, & Population Division. (2015). World urbanization prospects: The 2014 revision. ST/ESA/SER.A/366. New York, USA. Retrieved from: <u>https://esa.un.org/unpd/wup/Publications/Files/WUP2014-Report.pdf</u>
- [227] United Nations Environment Programme. (2003). Urban air pollution. Retrieved from <u>http://www.unep.org/urban_environment/issues/urban_air.asp</u>. Accessed on July 1, 2011.
- [228] US Census Bureau. (2010a). Hillsborough County, Florida, selected population demographics. Retrieved from <u>http://factfinder.census.gov/servlet/DatasetMainPageServlet? program=ACS& submenuId=& lang=en& ds_name=ACS_2009_5YR_G00_&ts=</u>. Accessed on March 22, 2013.
- [229] US Census Bureau. (2010b). Urban and rural. Retrieved from <u>https://factfinder.census.gov/faces/tableservices/jsf/pages/productview.xhtml?pid=DEC_00_SF1_P002&prodType=table</u>. Accessed on December 20, 2016.

- [230] US Environmental Protection Agency. (1994). Automobile emissions: An overview. Retrieved from <u>http://esa21.kennesaw.edu/activities/smog-cars/doe-veh-pollutants.pdf</u>
- [231] US Environmental Protection Agency. (2008). Integrated science assessment for oxides of nitrogen: Health criteria. October 12, 2013. Research Triangle, NC Retrieved from: http://www.epa.gov/ncea/isa/nitroxide.htm
- [232] US Environmental Protection Agency. (2013). Our built and natural environments: A technical review of the interactions between land use, transportation, and environmental quality. EPA 231K13001. Washington, DC. Retrieved from: <u>https://www.epa.gov/sites/production/files/2014-03/documents/our-built-and-natural-environments.pdf</u>
- [233] US Environmental Protection Agency. (2016). Inventory of US greenhouse gas emissions and sinks: 1990-2014. EPA 430-R-16-002. Washington, DC. Retrieved from: <u>https://www.epa.gov/sites/production/files/2016-04/documents/us-ghg-inventory-2016-maintext.pdf</u>
- [234] US Environmental Protection Agency, Assessment and Standards Division, & Office of Transportation and Air Quality. (2015). *Population and activity of on-road vehicles in MOVES2014*. EPA-420-D-15-001. Washington, DC. Retrieved from: https://cfpub.epa.gov/si/si_public_record_report.cfm?dirEntryId=309336
- [235] US Environmental Protection Agency, & Office of Transportation and Air Quality. (2015). MOVES2014a user guide. EPA-420-B-15-095. Washington, DC. Retrieved from: <u>http://www.epa.gov/otaq/models/moves/documents/420b12001.pdf</u>
- [236] US Environmental Protection Agency, Transportation and Regional Programs Division, & Office of Transportation and Air Quality (2010). *Transportation conformity guidance for quantitative hot-spot analyses in PM_{2.5} and PM₁₀ nonattainment and maintenance areas. EPA-420-B-10-040 Washington, DC. Retrieved from: https://www3.epa.gov/ttn/naaqs/aqmguide/collection/cp2/20101201_otaq_epa-420_b-10-040_transport_conform_hot-spot_analysis_appx.pdf*
- [237] US EPA Office of Minority Health. (2003). Building healthy environments to eliminate health disparities symposium. Washington, DC. Retrieved from: <u>http://www.epa.gov/environmentaljustice/resources/reports/annual-project-reports/healthdisparities-symposium-2003.pdf</u>
- [238] Vallamsundar, Lin, Konduri, Zhou, & Pendyala. (2016). A comprehensive modeling framework for transportation-induced population exposure assessment. *Transportation Research Part D: Transport and Environment*, 46, 94-113. doi:<u>http://dx.doi.org/10.1016/j.trd.2016.03.009</u>
- [239] Vallero. (2008). Fundamentals of air pollution. Burlington, MA: Academic Press.
- [240] Vlahov, & Galea. (2002). Urbanization, urbanicity, and health. *Journal of Urban Health*, 79(1), S1-S12. doi:10.1093/jurban/79.suppl_1.S1

- [241] Volk, Lurmann, Penfold, Hertz-Picciotto, & McConnell. (2013). Traffic-related air pollution, particulate matter, and autism. *JAMA Psychiatry*, *70*(1), 71-77. doi:10.1001/jamapsychiatry.2013.266
- [242] von Klot, Gryparis, Tonne, Yanosky, Coull, Goldberg, . . . Schwartz. (2009). Elemental carbon exposure at residence and survival after acute myocardial infarction. *Epidemiology*, 20(4), 547-554.
- [243] Waraich, Charypar, Balmer, & Axhausen. (2015). Performance improvements for large-scale traffic simulation in MATSim. In *Computational Approaches for Urban Environments*. Helbich, Jokar Arsanjani, & Leitner (Eds.).(pp. 211-233). Cham: Springer International Publishing.
- [244] Wiehe, Carroll, Liu, Haberkorn, Hoch, Wilson, & Fortenberry. (2008). Using GPS-enabled cell phones to track the travel patterns of adolescents. *International Journal of Health Geographics*, 7(1), 22. doi:10.1186/1476-072x-7-22
- [245] Winkelmann, Levermann, Ridgwell, & Caldeira. (2015). Combustion of available fossil fuel resources sufficient to eliminate the Antarctic Ice Sheet. Sci Adv, 1(8), e1500589. doi:10.1126/sciadv.1500589
- [246] Woodcock, Edwards, Tonne, Armstrong, Ashiru, Banister, . . . Roberts. (2009). Public health benefits of strategies to reduce greenhouse-gas emissions: Urban land transport. *The Lancet*, *374*(9705), 1930-1943. doi:http://dx.doi.org/10.1016/S0140-6736(09)61714-1
- [247] World Cancer Research Fund, & American Institute for Cancer Research. (2007). Policy and action for cancer prevention. Food, nutrition, and physical activity: A global perspective. Washington, DC. Retrieved from: http://www.aicr.org/assets/docs/pdf/reports/Second_Expert_Report.pdf
- [248] World Health Organization. (2004). Urban air pollution. In *Comparative Quantification of Health Risks*. Ezzati, Lopez, Rodgers, & Murray (Eds.).(Vol. 1, pp. 1353-1434). Geneva, Switzerland.
- [249] World Health Organization. (2010). Urbanization and health. *Bulletin of the World Health Organization*, 88(4), 245-246. doi:10.2471/BLT.10.010410
- [250] World Health Organization. (2014). Burden of disease from ambient air pollution for 2012. Geneva, Switzerland. Retrieved from: <u>http://www.who.int/phe/health_topics/outdoorair/databases/FINAL_HAP_AAP_BoD_24March2_014.pdf?ua=1</u>
- [251] Ye, Konduri, Pendyala, Sana, & Waddell. (2009). *A methodology to match distributions of both household and person attributes in the generation of synthetic populations*. Paper presented at the Transportation Research Board, Washington D.C.

- [252] Young, Sandler, DeRoo, Vedal, Kaufman, & London. (2014). Ambient air pollution exposure and incident adult asthma in a nationwide cohort of US women. *American Journal of Respiratory and Critical Care Medicine*, 190(8), 914-921. doi:10.1164/rccm.201403-0525OC
- [253] Yu, & Stuart. (2013). Spatiotemporal distributions of ambient oxides of nitrogen, with implications for exposure inequality and urban design. *Journal of the Air & Waste Management Association*, 63(8), 943-955. doi:10.1080/10962247.2013.800168
- [254] Yu, & Stuart. (2016). Exposure and inequality for select urban air pollutants in the Tampa Bay area. *Science of the Total Environment*, 551–552, 474-483. doi:https://doi.org/10.1016/j.scitotenv.2016.01.157
- [255] Yu, & Stuart. (2017). Impacts of compact growth and electric vehicles on future air quality and urban exposures may be mixed. *Science of the Total Environment*, 576, 148-158. doi:<u>https://doi.org/10.1016/j.scitotenv.2016.10.079</u>
- [256] Yusuf, Reddy, Ôunpuu, & Anand. (2001). Global burden of cardiovascular diseases: Part I: General considerations, the epidemiologic transition, risk factors, and impact of urbanization. *Circulation*, 104(22), 2746-2753. doi:10.1161/hc4601.099487
- [257] Zeger, Thomas, Dominici, Samet, Schwartz, Dockery, & Cohen. (2000). Exposure measurement error in time-series studies of air pollution: Concepts and consequences. *Environmental Health Perspectives*, *108*(5), 419-426.
- [258] Zhang, & Batterman. (2013). Air pollution and health risks due to vehicle traffic. *The Science of the total environment*, 450-451, 307-316. doi:10.1016/j.scitotenv.2013.01.074

APPENDIX A: COPYRIGHT PERMISSIONS

Below is the permission for the material published in Chapter 3.

Rightslink@ by Copyrigh X	ł	31	0	9	11
E 🗧 🖉 🔯 https://s100.copyright.com/AppDispatchServlet?publisherName=Springer&orderBeanReset=true&orderSource=SpringerLink&author=Sashkanth+Gurrani%/2C+Amy+Lynette+Stu	. 1	۵	6	0	:
Centra RightsLink Home Centre Help					
Springer Title: Impacts of travel activity and urbanicity on exposures to on exposure dispansion Author: Sashikanth Gurram, Any Lynote Stuart, Abdul Rawoof Pinjari Publication: Art (usiler, Atmosphere 8 Health Publisher: Springer Date: Jan 1, 2014 Compite 8 2044, The Author(2)					
Permissions Request					
This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestructed use, distribution, and reproduction in any medium, provided the original work is properly other.					
Springer and BioMed Central offer a reprint service for those who require professionally produced copies of articlas published under Creative Commons Attribution (CC BY) licenses. To obtain a quotation, please email reprints@scringeropen.com with the article details, quantity(ies) and delivery destination. Minimum order 25 copies.					
Copyright © 2017 Copyright Clearance Center, Inc. All Rights Reserved. <u>Privacy statements. Terms and</u> Conditions. Comments? We would like to hear from you. E-mail us at <u>pustomercaseBoopyright.com</u>					